SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934.

For the quarterly period ended August 25, 2000

or

☐ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934.

For the transition period

to

Commission File Number: 001-14965

The Goldman Sachs Group, Inc.

(Exact name of registrant as specified in its charter)

Delaware

(State or Other Jurisdiction of Incorporation or Organization)

13-4019460 (I.R.S. Employer Identification No.)

85 Broad Street, New York, NY (Address of principal executive offices)

10004 (Zip Code)

(212) 902-1000

(Registrant's Telephone Number, Including Area Code)

Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. \square Yes \square No

APPLICABLE ONLY TO CORPORATE ISSUERS

As of September 22, 2000, there were 454,235,989 shares of the registrant's common stock outstanding.

The Goldman Sachs Group, Inc. FORM 10-Q

		Page No.
PART I:	FINANCIAL INFORMATION	
Item 1:	Financial Statements (Unaudited)	
	Condensed Consolidated Statements of Earnings for the periods ended August 25, 2000 and August 27, 1999	2
	Condensed Consolidated Statements of Financial Condition as of August 25, 2000 and November 26, 1999	3
	Condensed Consolidated Statements of Changes in Stockholders' Equity and Partners' Capital for the periods ended August 25, 2000 and November 26, 1999	4
	Condensed Consolidated Statements of Cash Flows for the periods ended August 25, 2000 and August 27, 1999	5
	Condensed Consolidated Statements of Comprehensive Income for the periods ended August 25, 2000 and August 27, 1999	6
	Notes to Condensed Consolidated Financial Statements	7
	Review Report of Independent Accountants	14
Item 2:	Management's Discussion and Analysis of Financial Condition and Results of Operations	15
Item 3:	Quantitative and Qualitative Disclosures About Market Risk	26
PART II:	OTHER INFORMATION	
Item 1:	Legal Proceedings	26
Item 2:	Changes in Securities and Use of Proceeds	27
Item 5:	Other Information	27
Item 6:	Exhibits and Reports on Form 8-K	28
Signature	.s	29

PART I: FINANCIAL INFORMATION

Item 1: Financial Statements

THE GOLDMAN SACHS GROUP, INC. and SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF EARNINGS (UNAUDITED)

	Three Months Ended August			1	Nine Months E	August		
		2000		1999	2000			1999
		(in mi	llions	s, except share	and	per share amo	unts)	
Revenues								
Global capital markets								
Investment banking Trading and principal	\$	1,316	\$	1,150	\$	4,131	\$	3,054
investments		2,112		1,423		5,543		4,540
services		872		629		2,758		1,788
Interest income		4,551		3,238		12,579		9,269
Total revenues		8,851		6,440		25,011		18,651
Interest expense		4,324		3,032		11,836		8,779
Revenues, net of interest								
expense		4,527		3,408		13,175		9,872
Operating expenses Compensation and benefits, excluding employee initial public								
offering awards Nonrecurring employee initial public		2,263		1,704		6,587		4,932
offering awards (1)				_				2,257
public offering awards		102		115		314		154
Brokerage, clearing and exchange fees		136		108		419		328
Market development		126		92		343		247
Communications and technology		111		75		304		224
Depreciation and amortization		119		71		322		229
Occupancy		116		76		312		221
Professional services and other Charitable contribution		181 —		85		464		297 200
				2,326		0.065		
Total operating expenses Pre-tax earnings		3,154 1,373		1,082		9,065 4,110		9,089 783
Provision/(benefit) for taxes		549		444		1,644		(1,202)
Net earnings	\$	824	\$	638	\$	2,466	\$	1,985
Earnings per share								
Basic	\$	1.71 1.62	\$	1.34 1.32	\$	5.10 4.85	\$	4.18 4.11
Basic		31,252,647 98,894,645		74,694,245 83,892,677		33,403,066 08,181,472		4,698,130 3,146,111

⁽¹⁾ Includes expense of \$666 million related to the initial irrevocable contribution of shares of common stock to a defined contribution plan.

THE GOLDMAN SACHS GROUP, INC. and SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF FINANCIAL CONDITION (UNAUDITED)

(0107051125)	<i>A</i>	As of
	August 2000	November 1999
	(in millions	s, except share nare amounts)
Assets		,
Cash and cash equivalents	\$ 3,036	\$ 3,055
regulations	14,370	9,135
Receivables from brokers, dealers and clearing organizations	5,167 23,995	4,490 30,140
Receivables from customers and counterparties Securities borrowed	23,995 87,631	78.418
Securities purchased under agreements to resell	41,016	37,106
Right to receive securities	1,717	1,604
Financial instruments owned, at fair value	4.070	4 405
Commercial paper, certificates of deposit and time deposits	1,070 24,056	1,435 22,193
Corporate debt	12,360	9,821
Equities and convertible debentures	20,902	16,381
State, municipal and provincial obligations	822	756
Derivative contracts	32,848	30,661
Physical commodities	444 5,570	562 4,734
Other assets		
	\$275,004	<u>\$250,491</u>
Liabilities and Stockholders' Equity	Ф 07 047	Ф 07 7EC
Short-term borrowings, including commercial paper	\$ 37,917 3,420	\$ 37,756 2,129
Payables to customers and counterparties	57,879	57,405
Securities loaned	6,535	9,169
Securities sold under agreements to repurchase	42,697	40,183
Obligation to return securities	4,098	1,595
Financial instruments sold, but not yet purchased, at fair value U.S. government, federal agency and sovereign obligations	23,062	19,170
Corporate debt	4,521	2,642
Equities and convertible debentures	13,019	14,002
Derivative contracts	32,786	28,488
Physical commodities Other liabilities and accrued expenses	690 7,159	586 6,269
Long-term borrowings	28,528	20,952
3	262,311	240,346
Commitments and contingencies		0,0 .0
Stockholders' Equity		
Preferred stock, par value \$0.01 per share; 150,000,000 shares authorized, no shares issued and outstanding		
Common stock, par value \$0.01 per share; 4,000,000,000 shares authorized,	_	_
455,226,134 and 441,421,899 shares issued, 454,535,989 and 441,421,899		
shares outstanding as of August 2000 and November 1999, respectively	5	4
Restricted stock units; 67,361,270 and 76,048,404 units issued and	4.048	4,339
outstanding as of August 2000 and November 1999, respectively Nonvoting common stock, par value \$0.01 per share; 200,000,000 shares	4,040	4,339
authorized, 7,440,362 shares issued and outstanding as of		
November 1999		
Additional paid-in capital	7,728	7,359
Retained earnings	2,747 (1,681)	444 (2,038)
Accumulated other comprehensive (loss)/income	(89)	37
Treasury stock, at cost, par value \$0.01 per share; 690,145 shares as of	, ,	
August 2000	(65)	
Total stockholders' equity	12,693	10,145
	\$275,004	\$250,491

THE GOLDMAN SACHS GROUP, INC. and SUBSIDIARIES

CONDENSED CONSOLIDATED STATEMENTS OF CHANGES IN STOCKHOLDERS' EQUITY AND PARTNERS' CAPITAL (UNAUDITED)

(01110=11==)	Perio	d Ended
	August 2000	November 1999
	(in milli	ons, except re amounts)
Partners' capital Balance, beginning of period Transfer of beginning partners' capital allocated for income taxes and potential	\$ —	\$ 6,310
withdrawals Net earnings Capital contributions Return on capital and certain distributions to partners	_ _	74 2,264 (1) 48 (306)
Distributions of remaining partners' capital Exchange of partnership interests for shares of common stock Transfer to accumulated other comprehensive income.		(4,520)(2) (3,901) 31
Balance, end of period		
Common stock, par value \$0.01 per share Balance, beginning of period Issued	4 1	4
Balance, end of period	5	4
Restricted stock units Balance, beginning of period	4,339	_
Granted	363	4,381
Delivered Forfeited	(501) (153)	(42)
Balance, end of period	4,048	4,339
Nonvoting common stock, par value \$0.01 per share Balance, beginning of period	_	_
Balance, end of period.		
Additional paid-in capital Balance, beginning of period Exchange of partnership interests for shares of common stock Issuance of common stock Issuance of common stock contributed to a defined contribution plan	7,359 369	3,901 2,891 674
Dividends paid		(107)(3)
Balance, end of period	7,728	7,359
Retained earnings Balance, beginning of period Net earnings Dividends paid	444 2,466 (163)	444 (4)
Balance, end of period	2,747	444
Unearned compensation Balance, beginning of period Restricted stock units granted Restricted stock units forfeited Amortization of restricted stock units	(2,038) (363) 126 594	(2,334) 23 273
Balance, end of period	(1,681)	(2,038)
Accumulated other comprehensive (loss)/income Balance, beginning of period	<u>37</u>	(31)
Currency translation adjustment	(126)	
Balance, end of period	(89)	37
Treasury stock, at cost, par value \$0.01 per share Balance, beginning of period	<u> </u>	=
Balance, end of period	(65) \$12,693	<u>—</u> \$10,145

Represents net earnings of the partnership from November 28, 1998 through May 6, 1999.
 Represents the retired limited partners' exchanges of partnership interests for cash and junior subordinated debentures, the redemption of senior limited partnership interests for cash and other distributions of partners' capital in accordance with the partnership agreement.
 Represents two quarterly dividends of \$0.12 per common share each.
 Represents net earnings of the corporation from May 7, 1999 through November 26, 1999.

THE GOLDMAN SACHS GROUP, INC. and SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

	Nine M Ended A	
	2000	1999
	(in mill	ions)
Cash flows from operating activities		
Net earnings	\$ 2,466	\$ 1,985
Noncash items included in net earnings		
Depreciation and amortization	322	229
Stock-based compensation	576	2,405
Changes in operating assets and liabilities		
Cash and securities segregated in compliance with U.S. federal and other	(5.005)	(4.050)
regulations	(5,235)	(1,058)
Net receivables from brokers, dealers and clearing organizations	614	116
Net payables to customers and counterparties	6,619	(3,812)
Securities borrowed, net	(11,847)	(9,025)
Financial instruments owned, at fair value	(9,187)	(5,983)
Financial instruments sold, but not yet purchased, at fair value	10,173	10,012
Other, net	474	(176)
Net cash used for operating activities	(5,025)	(5,307)
Property, leasehold improvements and equipment	(990)	(368)
Financial instruments owned, at fair value	(118)	119
Net cash used for investing activities	(1,108)	(249)
Cash flows from financing activities	(1,100)	(240)
Short-term borrowings, net	(5.092)	720
Issuance of long-term borrowings	13,157	9.098
Repayment of long-term borrowings	(327)	(572)
Securities sold under agreements to repurchase, net	(1,396)	(2,842)
Common stock repurchased	(65)	(=,0 :=)
Dividends paid	(163)	(53)
Capital contributions	_	48
Returns on capital and certain distributions to partners	_	(306)
Proceeds from issuance of common stock	_	2,639
Partners' capital distributions, net		(4,112)
Net cash provided by financing activities	6.114	4,620
Net decrease in cash and cash equivalents	(19)	(936)
Cash and cash equivalents, beginning of period	3,055	2,836
	\$ 3.036	\$ 1,900
Cash and cash equivalents, end of period	ψ 3,030 ==================================	φ 1,900

SUPPLEMENTAL DISCLOSURES:

Cash payments for interest approximated the related expense for each of the fiscal periods presented. Payments of income taxes were \$1.59 billion and \$236 million for the nine months ended August 25, 2000 and August 27, 1999, respectively.

Other, net for the nine months ended August 27, 1999 includes an increase in deferred tax assets of \$1.78 billion associated with the firm's conversion to corporate form and related transactions.

Noncash activities:

In connection with the firm's conversion to corporate form, junior subordinated debentures of \$371 million were issued to retired limited partners in exchange for their partnership interests.

THE GOLDMAN SACHS GROUP, INC. and SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (UNAUDITED)

	Three Months Ended August		Nine M Ended		
	2000 1999		2000	1999	
		(in	millions)		
Net earnings	\$824	\$638	\$2,466	\$1,985	
Currency translation adjustment, net of tax	(16)	47	(126)	12	
Comprehensive income	\$808	\$685	\$2,340	\$1,997	

Note 1. Description of Business

The Goldman Sachs Group, Inc. ("Group Inc."), a Delaware corporation, together with its consolidated subsidiaries (collectively, the "firm"), is a global investment banking and securities firm that provides a wide range of financial services worldwide to a substantial and diversified client base. On May 7, 1999, the firm converted from a partnership to a corporation and completed its initial public offering.

The firm's activities are divided into two segments:

- Global Capital Markets. This segment comprises Investment Banking, which includes Financial Advisory and Underwriting, and Trading and Principal Investments, which includes Fixed Income, Currency and Commodities ("FICC"), Equities and Principal Investments (Principal Investments primarily represents net revenues from the firm's merchant banking investments); and
- Asset Management and Securities Services. This segment comprises Asset Management, Securities Services and Commissions.

Note 2. Significant Accounting Policies

Basis of Presentation

The condensed consolidated financial statements include the accounts of Group Inc. and its U.S. and international subsidiaries including Goldman, Sachs & Co. ("GS&Co.") and J. Aron & Company in New York, Goldman Sachs International ("GSI") in London and Goldman Sachs (Japan) Ltd. ("GSJL") in Tokyo. These condensed consolidated financial statements are unaudited and should be read in conjunction with the audited consolidated financial statements included in the Annual Report on Form 10-K of Group Inc. for the fiscal year ended November 26, 1999. The condensed consolidated financial information as of and for the period ended November 26, 1999 has been derived from audited consolidated financial statements not included herein. Certain reclassifications have been made to prior-year amounts to conform to the current-year presentation. All material intercompany transactions and balances have been eliminated.

These condensed consolidated financial statements have been prepared in accordance with generally accepted accounting principles that require management to make estimates and assumptions regarding trading inventory valuations, the outcome of pending litigation and other matters that affect the consolidated financial statements and related disclosures. These estimates and assumptions are based on judgment and available information and, consequently, actual results could be materially different from these estimates.

These unaudited condensed consolidated financial statements reflect all adjustments, consisting only of normal recurring adjustments that are, in the opinion of management, necessary for a fair statement of the results in the interim periods presented. Interim period operating results may not be indicative of the operating results for a full year.

Unless otherwise stated herein, all references to August 2000 and August 1999 refer to the firm's fiscal period ended, or the date, as the context requires, August 25, 2000 and August 27, 1999, respectively. All references to November 1999 refer to the firm's fiscal year ended, or the date, as the context requires, November 26, 1999.

Accounting Developments

In September 2000, the Financial Accounting Standards Board ("FASB") issued Statement of Financial Accounting Standards ("SFAS") No. 140, "Accounting for Transfers and Servicing of Financial Assets and Extinguishments of Liabilities — a replacement of FASB Statement No. 125", which revises the standards for accounting for securitizations and other transfers of financial assets and collateral. In addition, specific implementation guidelines have been established to further distinguish transfers of financial assets that are sales from transfers that are secured borrowings. SFAS No. 140 is effective for transfers occurring after March 31, 2001 and for disclosures relating to securitization transactions and collateral for fiscal years ending after December 15, 2000. The firm intends to adopt the provisions of SFAS No. 140 in 2001 and is currently assessing its effect.

In June 2000, the FASB issued SFAS No. 138, "Accounting for Certain Derivative Instruments and Certain Hedging Activities", which is an amendment of SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities." The Statement is effective concurrently with SFAS No. 137, "Accounting for Derivative Instruments and Hedging Activities — Deferral of the Effective Date of FASB Statement No. 133 — an amendment of FASB Statement No. 133", which deferred to fiscal years beginning after June 15, 2000 the effective date of the accounting and reporting requirements of SFAS No. 133. These statements establish accounting and reporting standards for derivative instruments, including certain derivative instruments embedded in other contracts (collectively, referred to as "derivatives"), and for hedging activities. These statements require that an entity recognize all derivatives as either assets or liabilities in the statement of financial condition and measure those instruments at fair value. The accounting for changes in the fair value of a derivative instrument depends on its intended use and the resulting designation. The firm intends to adopt the provisions of SFAS No. 133 deferred by SFAS No. 137 and amended by SFAS No. 138 in fiscal 2001 and is currently assessing its effect.

In March 1998, the Accounting Standards Executive Committee of the American Institute of Certified Public Accountants issued Statement of Position ("SOP") No. 98-1, "Accounting for the Costs of Computer Software Developed or Obtained for Internal Use". SOP No. 98-1 requires capitalization of certain internal use software costs. SOP No. 98-1 was adopted by the firm in the first quarter of fiscal 2000 and was not material to the firm's financial condition or its results of operations for the quarter and nine months ended August 2000.

Note 3. Financial Instruments

Gains and losses on financial instruments and commission income and related expenses are recorded on a trade date basis in the condensed consolidated statements of earnings. The condensed consolidated statements of financial condition generally reflect purchases and sales of financial instruments, including agency transactions, on a trade date basis.

Substantially all financial instruments used in the firm's trading and nontrading activities are carried at fair value or amounts that approximate fair value, and unrealized gains and losses are recognized in earnings. Fair value is based generally on listed market prices or broker or dealer price quotations. To the extent that prices are not readily available, or if liquidating the firm's position is reasonably expected to affect market prices, fair value is based on either internal valuation models or management's estimate of amounts that could be realized under current market conditions, assuming an orderly liquidation over a reasonable period of time. Certain over-the-counter derivative instruments are valued using pricing models that consider, among

other factors, current and contractual market prices, time value, and yield curve and/or volatility factors of the underlying positions.

Derivative Activities

Most of the firm's derivative transactions are entered into for trading purposes. The firm uses derivatives in its trading activities to facilitate customer transactions, to take proprietary positions and as a means of risk management. The firm also enters into nontrading derivative contracts to manage the interest rate and currency exposure on its long-term borrowings.

Derivative contracts are financial instruments, such as futures, forwards, swaps or option contracts, that derive their value from underlying assets, indices, reference rates or a combination of these factors. Derivatives may involve future commitments to purchase or sell financial instruments or commodities, or to exchange currency or interest payment streams. The amounts exchanged are based on the specific terms of the contract with reference to specified rates, securities, commodities or indices.

Derivative contracts exclude certain cash instruments, such as mortgage-backed securities, interest-only and principal-only obligations, and indexed debt instruments, that derive their values or contractually required cash flows from the price of some other security or index. Derivatives also exclude option features that are embedded in cash instruments, such as the conversion features and call provisions embedded in bonds. The firm has elected to include commodity-related contracts in its derivative disclosure, although not required to do so, as these contracts may be settled in cash or are readily convertible into cash.

The firm utilizes replacement cost as a measure of derivative credit risk. Replacement cost, as reported in "Financial instruments owned, at fair value" on the condensed consolidated statements of financial condition, represents amounts receivable from various counterparties, net of any unrealized losses, where management believes a legal right of setoff exists under an enforceable netting agreement. Replacement cost for purchased option contracts is the market value of the contract. The firm controls its credit risk through an established credit approval process, by monitoring counterparty limits, obtaining collateral where appropriate and, in some cases, entering into enforceable netting agreements.

The fair value of derivative financial instruments used for trading purposes, computed in accordance with the firm's netting policy, is set forth below:

	As of Au	gust 2000	As of Nove	mber 1999
	Assets	Liabilities	Assets	Liabilities
Forward settlement contracts	\$ 4,680	\$ 4,419	\$ 4,555	\$ 4,625
Swap agreements	15,507	14,982	12,052	11,587
Option contracts	12,659	13,335	14,018	12,274
Total	\$ 32,846	\$ 32,736	\$ 30,625	\$ 28,486

Derivatives used for nontrading purposes generally include interest rate futures contracts and interest rate and currency swap agreements, which are primarily utilized to convert a substantial portion of the firm's fixed rate debt into U.S. dollar-based floating rate obligations. Gains and losses on these derivatives are generally deferred and recognized as adjustments to interest expense over the life of the derivative contract. Gains and losses resulting from the early termination of derivatives used for nontrading purposes are generally deferred and recognized

over the remaining life of the underlying debt. If the underlying debt is terminated prior to its stated maturity, gains and losses on these transactions, including the associated hedges, are recognized in earnings immediately.

The fair value and carrying value of derivatives used for nontrading purposes are set forth below:

	As of August 2000		As of Nov	vember 1999
	Assets	Liabilities	Assets	Liabilities
		(in mi	llions)	
Fair value	\$13	\$595	\$ 3	\$159
Carrying value	2	50	36	2

Note 4. Short-Term Borrowings

The firm obtains secured short-term financing principally through the use of repurchase agreements and securities lending agreements, collateralized mainly by U.S. government, federal agency, investment-grade foreign sovereign obligations and equity securities. The firm obtains unsecured short-term borrowings through issuance of commercial paper, promissory notes and bank loans. The carrying value of these short-term obligations approximates fair value due to their short-term nature.

Short-term borrowings are set forth below:

	A	s of
	August 2000	November 1999
	(in n	nillions)
Commercial paper	\$15,201	\$ 9,403
Promissory notes	13,330	11,061
Bank loans and other(1)	9,386	17,292
Total	\$37,917	\$37,756

⁽¹⁾ As of August 2000 and November 1999, short-term borrowings included \$5.95 billion and \$10.82 billion, respectively, of long-term borrowings maturing within one year.

The firm maintains unencumbered securities with a market value in excess of all uncollateralized short-term borrowings.

Note 5. Equity

In March 2000, the Board of Directors of Group Inc. approved a common stock repurchase program authorizing the repurchase of up to 15 million shares of the firm's common stock. For the nine months ended August 2000, the firm repurchased 690,145 shares of its common stock.

On August 21, 2000, Sumitomo Bank Capital Markets, Inc. exchanged all 7,440,362 shares of its nonvoting common stock, par value \$0.01 per share, of Group Inc. for an equal number of shares of common stock.

Note 6. Earnings Per Share

Earnings per share ("EPS") is computed in accordance with SFAS No. 128, "Earnings Per Share". Basic EPS is calculated by dividing net earnings by the weighted average number of

common shares outstanding. Diluted EPS includes the determinants of basic EPS and, in addition, gives effect to dilutive potential common shares.

The computations of basic and diluted EPS are set forth below:

	Three Months	Ended August	Nine Months Ended August				
	2000	1999	2000 1999				
	(in millions, except share and per share amounts)						
Numerator for basic and diluted EPS — earnings available to common stockholders	\$ 824	<u>\$ 638</u>	\$ 2,466	<u>\$ 1,985</u>			
Denominator for basic EPS — weighted average number of common shares(1)	481,252,647	474,694,245	483,403,066	474,698,130			
Effect of dilutive securities Restricted stock units Stock options	17,101,019 10,540,979	4,987,721 4,210,711	14,775,882 10,002,524	4,508,530 3,939,451			
Dilutive potential common shares	27,641,998	9,198,432	24,778,406	8,447,981			
Denominator for diluted EPS — weighted average number of common shares and dilutive potential common shares	508,894,645	483,892,677	508,181,472	483,146,111			
Basic EPS	\$ 1.71 1.62	\$ 1.34 1.32	\$ 5.10 4.85	\$ 4.18 4.11			

⁽¹⁾ Includes common stock and nonvoting common stock as well as restricted stock units awarded to employees for which no future service is required as a condition to the delivery of the underlying shares of common stock.

Note 7. Commitments and Contingencies

The firm is involved in a number of judicial, regulatory and arbitration proceedings concerning matters arising in connection with the conduct of its businesses. Management believes, based on currently available information, that the results of such proceedings, in the aggregate, will not have a material adverse effect on the firm's financial condition, but might be material to the firm's operating results for any particular period, depending, in part, upon the operating results for such period.

Note 8. Regulated Subsidiaries

GS&Co. is a registered U.S. broker-dealer subsidiary, which is subject to the Securities and Exchange Commission's "Uniform Net Capital Rule," and has elected to compute its net capital in accordance with the "Alternative Net Capital Requirement" of that rule. As of August 2000, GS&Co. had regulatory net capital, as defined, of \$4.31 billion, which exceeded the amount required by \$3.63 billion.

GSI, a registered U.K. broker-dealer and subsidiary of Group Inc., is subject to the capital requirements of the Securities and Futures Authority Limited, and GSJL, a Tokyo-based broker-dealer, is subject to the capital requirements of the Financial Services Agency. As of August 2000, GSI and GSJL were in compliance with their local capital adequacy requirements.

Certain other subsidiaries of the firm are also subject to capital adequacy requirements promulgated by authorities of the countries in which they operate. As of August 2000, these subsidiaries were in compliance with their local capital adequacy requirements.

Note 9. Business Segments

In reporting to management, the firm's operating results are categorized into two principal segments: Global Capital Markets; and Asset Management and Securities Services. For a further discussion of the firm's segments, see the firm's Annual Report on Form 10-K for the fiscal year ended November 1999.

Management believes that the following information provides a reasonable representation of each segment's contribution to consolidated pre-tax earnings and total assets:

			Months August	Nine N Ended	lonths August
		2000	1999	2000	1999
			(in mill	lions)	
Global Capital	Net revenues	\$ 3,436	\$ 2,597	\$ 9,753	\$ 7,576
Markets	Operating expenses	2,211	1,623	6,389	4,701
	Pre-tax earnings	\$ 1,225	\$ 974	\$ 3,364	\$ 2,875
	Segment assets	\$143,039	\$116,680	\$143,039	\$116,680
Asset Management	Net revenues	\$ 1,091	\$ 811	\$ 3,422	\$ 2,296
and Securities	Operating expenses	841	588	2,362	1,777
Services	Pre-tax earnings	\$ 250	\$ 223	\$ 1,060	\$ 519
	Segment assets	\$130,974	\$118,283	\$130,974	\$118,283
Total	Net revenues	\$ 4,527	\$ 3,408	\$ 13,175	\$ 9,872
	Operating expenses (1)	3,154	2,326	9,065	9,089(3)
	Pre-tax earnings	\$ 1,373	\$ 1,082	\$ 4,110	\$ 783
	Total assets(2)	\$275,004	\$236,273	\$275,004	\$236,273

⁽¹⁾ Includes the ongoing amortization of employee initial public offering awards that has not been allocated to the firm's segments.

⁽²⁾ Includes deferred tax assets relating to the firm's conversion to corporate form and certain other assets that are not allocable to a particular segment.

⁽³⁾ Includes nonrecurring employee initial public offering awards of \$2.26 billion and a charitable contribution to The Goldman Sachs Foundation of \$200 million made at the time of the firm's initial public offering that have not been allocated to the firm's segments.

Note 10. Subsequent Events

The Board of Directors of Group Inc. declared a dividend of \$0.12 per share to be paid on November 20, 2000 to common stockholders of record on October 23, 2000.

On September 11, 2000, the firm announced an agreement to combine with Spear, Leeds & Kellogg, L.P. (''SLK''), a leader in securities clearing and execution, floor-based market making and off-floor market making. As part of this agreement, the consideration will include approximately 34 million shares of Group Inc. common stock and approximately \$2 billion in cash. In addition, the firm is establishing a \$900 million retention pool in Group Inc. common stock for all SLK employees, which will have varying vesting and delivery provisions. The transaction is expected to close before year-end, and is subject to customary regulatory and other approvals.

Review Report of Independent Accountants

To the Directors and Shareholders, The Goldman Sachs Group, Inc.

We have reviewed the accompanying condensed consolidated statement of financial condition of The Goldman Sachs Group, Inc. and Subsidiaries (the "Company") as of August 25, 2000, the related condensed consolidated statements of earnings for the three and nine months ended August 25, 2000 and August 27, 1999, the condensed consolidated statement of changes in stockholders' equity and partners' capital for the nine months ended August 25, 2000, the condensed consolidated statements of cash flows for the nine months ended August 25, 2000 and August 27, 1999, and the condensed consolidated statements of comprehensive income for the three and nine months ended August 25, 2000 and August 27, 1999. These condensed financial statements are the responsibility of the Company's management.

We conducted our review in accordance with standards established by the American Institute of Certified Public Accountants. A review of interim financial information consists principally of applying analytical procedures to financial data and making inquiries of persons responsible for financial and accounting matters. It is substantially less in scope than an audit conducted in accordance with auditing standards generally accepted in the United States of America, the objective of which is the expression of an opinion regarding the financial statements taken as a whole. Accordingly, we do not express such an opinion.

Based on our review, we are not aware of any material modifications that should be made to the accompanying condensed consolidated interim financial statements for them to be in conformity with accounting principles generally accepted in the United States of America.

We previously audited in accordance with auditing standards generally accepted in the United States of America, the consolidated statement of financial condition of The Goldman Sachs Group, Inc. and Subsidiaries as of November 26, 1999, and the related consolidated statements of earnings, changes in stockholders' equity and partners' capital, cash flows and comprehensive income for the year ended November 26, 1999 (not presented herein); and in our report dated January 21, 2000, we expressed an unqualified opinion on those consolidated financial statements. In our opinion, the information set forth in the accompanying condensed consolidated statement of financial condition as of November 26, 1999, and the condensed consolidated statement of changes in stockholders' equity and partners' capital for the year ended November 26, 1999, is fairly stated in all material respects in relation to the consolidated financial statements from which it has been derived.

/s/ PricewaterhouseCoopers LLP

New York, New York October 9, 2000.

Item 2: Management's Discussion and Analysis of Financial Condition and Results of Operations

Introduction

Goldman Sachs is a global investment banking and securities firm that provides a wide range of services worldwide to a substantial and diversified client base. On May 7, 1999, we converted from a partnership to a corporation and completed our initial public offering.

Our activities are divided into two segments:

- Global Capital Markets. This segment comprises Investment Banking, which includes Financial Advisory and Underwriting, and Trading and Principal Investments, which includes Fixed Income, Currency and Commodities ("FICC"), Equities and Principal Investments (Principal Investments primarily represents net revenues from our merchant banking investments); and
- Asset Management and Securities Services. This segment comprises Asset Management, Securities Services and Commissions.

Unless specifically stated otherwise, all references to August 2000 and August 1999 refer to our fiscal period ended, or the date, as the context requires, August 25, 2000 and August 27, 1999, respectively. All references to November 1999 and November 1998, unless specifically stated otherwise, refer to our fiscal year ended, or the date, as the context requires, November 26, 1999 and November 27, 1998, respectively.

When we use the terms "Goldman Sachs", "we" and "our", we mean, prior to our conversion to corporate form, The Goldman Sachs Group, L.P., a Delaware limited partnership, and its consolidated subsidiaries and, after our conversion to corporate form, The Goldman Sachs Group, Inc. ("Group Inc."), a Delaware corporation, and its consolidated subsidiaries.

Business Environment

During the quarter, economic growth in the world economy continued to be strong, although the pace of growth slowed primarily as a result of earlier increases in interest rates, stabilization in asset prices and rising oil prices.

Growth in the U.S. economy was driven by strong exports and corporate investment. Despite these factors, the rate of growth declined during the quarter as some elements of consumer spending slowed and residential investment fell. U.S. equity markets rebounded from the sharp declines at the end of the prior quarter, particularly the Nasdaq which gained 23% during the fiscal quarter. The fixed income markets generally benefited from narrowing credit spreads and steady short-term interest rates.

The European economy continued to grow at a healthy rate, fueled by strong growth in foreign demand, consumer spending and corporate investment. During the quarter, the European Central Bank continued to raise short-term rates in response to inflation and the weakness of the euro.

Economic growth in Japan slowed from the exceptional rates recorded earlier in the year. Despite this slowdown, deflationary pressures receded and growth in investment rose significantly amidst a rebound in corporate profitability. The Bank of Japan's zero-rate policy was terminated and interest rates were raised for the first time in a decade. Growth fell back to more normal rates in several other Asian economies as exports slowed modestly, and corporate investment failed to rebound to earlier levels.

Results of Operations

The composition of our net revenues has varied over time as financial markets and the scope of our operations have changed. The composition of net revenues can also vary over the shorter term due to fluctuations in U.S. and global economic and market conditions. As a result, period-to-period comparisons may not be meaningful. In addition, Goldman Sachs' conversion to corporate form has affected, and will continue to affect, our operating results in several significant ways:

- 1. Former Partner Compensation. As a corporation, payments for services rendered by managing directors who, prior to our conversion to corporate form, were profit participating limited partners are included in compensation and benefits expense. Prior to our conversion to corporate form, these payments were accounted for as distributions of partners' capital rather than as compensation and benefits expense.
- 2. Ongoing Stock-Based Compensation. As part of compensation, restricted stock units and other forms of stock-based compensation can be awarded to employees. Of the total restricted stock units that were granted at the end of November 1999, approximately 50% require future service as a condition to the delivery of the underlying shares of common stock. In accordance with Accounting Principles Board Opinion No. 25, these restricted stock units with future service requirements will generally be recorded as compensation expense over the four-year service period following the date of grant as follows: 52%, 28%, 14% and 6% in years one, two, three and four, respectively.
- **3. Amortization of Employee Initial Public Offering Awards.** We have recorded, and will continue to record over the five-year vesting period following the date of grant, noncash expense related to the amortization of certain restricted stock units awarded to employees in connection with our initial public offering. These restricted stock units had a value of \$1.76 billion at date of grant, approximately 26% of which was amortized as a noncash expense, after giving effect to forfeitures, in the 12 months following the date of grant. The remaining 74% of the value of these restricted stock units is being amortized over the next four years as follows: 26%, 26%, 15% and 7% in years two, three, four and five, respectively.
- **4. Income Taxes.** As a corporation, our operating results have become, and will continue to be, subject to U.S. federal, state and local corporate income taxes, and, therefore, to a higher tax rate than we incurred as a partnership. Our effective tax rate for the quarter and the ninemonth period ended August 2000 was 40%.

Overview

The following table sets forth a summary of our financial results:

Financial Overview (in millions, except per share amounts)

	Three M	Ionths End	ed August	Nine Months Ended August			
	Actual		Actual Pro Forma		Actual		
	2000	1999	1999	2000	1999(1)	1999	
Net revenues	\$4,527	\$3,408	\$3,408	\$13,175	\$9,872	\$9,865	
Pre-tax earnings	1,373	1,082	1,082	4,110	783	3,041	
Net earnings	824	638	638	2,466	1,985	1,794	
Diluted earnings per share	1.62	1.32	1.31	4.85	4.11	3.73	

⁽¹⁾ Includes 23 weeks as a partnership.

16

Pro forma net earnings reflect the results of Goldman Sachs as if our conversion to corporate form and related transactions had taken place at the beginning of 1999.

Pro forma net earnings do not give effect to the following items due to their nonrecurring nature:

- the employee initial public offering award of restricted stock units, for which future service is not required as a condition to the delivery of the underlying shares of common stock;
- the initial irrevocable contribution of shares of common stock to the defined contribution plan;
- · the recognition of certain net tax assets; and
- a contribution to The Goldman Sachs Foundation, a charitable foundation.

Pro forma net earnings give effect to the following items:

- interest expense on junior subordinated debentures issued to retired limited partners in exchange for their partnership interests;
- the amortization of the restricted stock units awarded to employees in connection with our initial public offering, for which future service is required as a condition to the delivery of the underlying shares of common stock; and
- the provision for income taxes in corporate form.

For the purpose of calculating pro forma diluted average common shares outstanding for the quarter and nine months ended August 1999 we used the initial public offering price of \$53 per share from the beginning of fiscal 1999 until May 4, 1999, the day trading in our common stock commenced.

The following table sets forth the net revenues, operating expenses and pre-tax earnings of our segments:

Results by Segment

(in millions) Three Months **Nine Months Ended August Ended August** 2000 1999 2000 1999 \$2,597 Global Capital Net revenues \$3,436 9,753 \$7,576 Markets Operating expenses 2,211 1,623 6,389 4,701 974 3.364 \$1,225 \$2.875 Asset Management Net revenues \$1,091 811 \$ 3,422 \$2,296 And Securities Operating expenses 841 588 2,362 1,777 Services \$ 250 Pre-tax earnings 1,060 519 Total Net revenues \$4.527 \$3,408 \$13,175 \$9.872 Operating expenses (1) 3,154 2,326 9,065 9,089(2) \$1,373 \$1,082 4,110 783

⁽¹⁾ Includes the ongoing amortization of employee initial public offering awards that has not been allocated to our segments.

⁽²⁾ Includes nonrecurring employee initial public offering awards of \$2.26 billion and a charitable contribution to The Goldman Sachs Foundation of \$200 million made at the time of our initial public offering that have not been allocated to our segments.

Global Capital Markets

The components of the Global Capital Markets segment are set forth below:

Investment Banking. Goldman Sachs provides a broad range of investment banking services to a diverse group of corporations, financial institutions, governments and individuals. Our investment banking activities are divided into two categories:

- **Financial Advisory.** Financial Advisory includes advisory assignments with respect to mergers and acquisitions, divestitures, corporate defense activities, restructurings and spin-offs; and
- **Underwriting.** Underwriting includes public offerings and private placements of equity and debt securities.

Trading and Principal Investments. Our Trading and Principal Investments business facilitates transactions with a diverse group of corporations, financial institutions, governments and individuals and takes proprietary positions through market making in and trading of fixed income and equity products, currencies, commodities, and swaps and other derivatives. Trading and Principal Investments is divided into three categories:

- FICC. We make markets in and trade fixed income products, currencies and commodities, structure and enter into a wide variety of derivative transactions, and engage in proprietary trading and arbitrage activities;
- Equities. We make markets in and trade equities and equity-related products, structure and enter into equity derivative transactions, and engage in proprietary trading and equity arbitrage; and
- **Principal Investments.** Principal Investments primarily represents net revenues from our merchant banking investments.

Net revenues from Principal Investments do not include management fees and the increased share of the income and gains from our merchant banking funds to which Goldman Sachs is entitled when the return on investments exceeds certain threshold returns to fund investors. These management fees and increased shares of income and gains are included in the net revenues of Asset Management and Securities Services.

Substantially all of our inventory is marked-to-market daily and, therefore, its value and our net revenues are subject to fluctuations based on market movements. In addition, net revenues derived from our principal investments in privately held concerns and in real estate may fluctuate significantly depending on the revaluation or sale of these investments in any given period.

The following table sets forth the net revenues of our Global Capital Markets segment:

Global Capital Markets Net Revenues

(in millions)

	Three Months Ended August		Nine Months Ended August	
	2000	1999	2000	1999
Financial Advisory	\$ 673	\$ 616	\$1,968	\$1,648
Underwriting	648	534	2,183	1,406
Investment Banking	1,321	1,150	4,151	3,054
FICC	872	661	2,522	2,448
Equities	763	458	2,707	1,531
Principal Investments	480	328	373	543
Trading and Principal Investments	2,115	1,447	5,602	4,522
Total	\$3,436	\$2,597	\$9,753	\$7,576

Three Months Ended August 2000 versus Three Months Ended August 1999

Net revenues in Global Capital Markets increased 32% to \$3.44 billion, reflecting strong revenue growth in Investment Banking and Trading and Principal Investments. Operating expenses increased 36%, principally due to higher levels of compensation commensurate with growth in net revenues, and increased costs associated with global expansion, higher employment levels and increased business activity. Pre-tax earnings were \$1.23 billion compared to \$974 million in 1999.

Net revenues in Investment Banking increased 15% to \$1.32 billion. Revenue growth was strong in all major regions. Financial Advisory net revenues increased 9% as we capitalized on increased mergers and acquisitions activity in the communications, media and entertainment and high technology sectors. Net revenues in Underwriting increased 21% as we benefited from increased new issue activity in global equity markets. Net revenue growth was largely driven by strong performances in the communications, media and entertainment and high technology sectors. Our investment banking transaction backlog as of August 2000 remained strong.

Net revenues in Trading and Principal Investments increased 46% to \$2.12 billion. FICC net revenues increased 32%, primarily due to increased customer flow in fixed income derivatives and improved performances in the Japanese and European government bond businesses, partially offset by lower net revenues from decreased customer activity in our commodities and high-yield businesses. Net revenues in Equities rose 67%, primarily resulting from strength in equity derivatives and higher transaction volumes in our U.S. and European shares businesses. Principal Investments net revenues increased 46% to \$480 million. These net revenues included significant gains, balanced between realized and unrealized, on certain of our merchant banking investments in the high technology and telecommunications sectors.

Nine Months Ended August 2000 versus Nine Months Ended August 1999

Net revenues in Global Capital Markets increased 29% to \$9.75 billion, reflecting strong performances in both Investment Banking and Trading and Principal Investments. Operating expenses increased 36%, principally due to higher levels of compensation commensurate with growth in net revenues, and increased costs associated with global expansion and higher levels of business activity. Pre-tax earnings were \$3.36 billion compared to \$2.88 billion in 1999.

Investment Banking net revenues increased 36% to \$4.15 billion. Net revenues in Financial Advisory and Underwriting increased 19% and 55%, respectively, reflecting continued strength in our mergers and acquisitions and equity new issues businesses, particularly in the communications, media and entertainment and high technology sectors. Net revenue growth was strong in all major regions.

Net revenues in Trading and Principal Investments increased 24% to \$5.60 billion. FICC net revenues grew 3%, as increased customer flow in our fixed income derivatives business was partially offset by lower net revenues in our credit-sensitive (which includes high-yield debt, bank loans and investment-grade corporate debt), commodities and U.S. government bond businesses. Net revenues in Equities increased 77%, primarily due to strength in equity derivatives and higher transaction volumes in our U.S. and European shares businesses. Principal Investments decreased 31% primarily due to lower unrealized gains in technology and telecommunications stocks, partially offset by higher net revenues from the disposition of investments.

Asset Management and Securities Services

The components of the Asset Management and Securities Services segment are set forth below:

- Asset Management. Asset Management generates management fees by providing investment advisory services to a diverse client base of institutions and individuals;
- Securities Services. Securities Services includes prime brokerage, financing services and securities lending, and our matched book businesses, all of which generate revenues primarily in the form of fees or interest rate spreads; and
- **Commissions.** Commissions include agency transactions for clients on major stock and futures exchanges and revenues from the increased share of the income and gains derived from our merchant banking funds.

The following table sets forth the net revenues of our Asset Management and Securities Services segment:

Asset Management and Securities Services Net Revenues (in millions)

· ·	Three Months Ended August			Nine Months Ended August	
	2000	1999	2000	1999	
Asset Management	\$ 327	\$221	\$ 987	\$ 637	
Securities Services	234	195	724	576	
Commissions	530	395	1,711	1,083	
Total	\$1,091	<u>\$811</u>	\$3,422	\$2,296	

Our assets under supervision consist of assets under management and other client assets. Assets under management typically generate fees based on a percentage of their value and include our mutual funds, separate accounts managed for institutional and individual investors, our merchant banking funds and other alternative investment funds. Other client assets consist of assets in brokerage accounts of primarily high-net-worth individuals, on which we earn commissions. Substantially all assets under supervision are valued as of calendar month-end.

The following table sets forth our assets under supervision:

Assets Under Supervision

(in millions)

	As of August 31,		As of November 30,		
	2000	1999	1999	1998	
Assets under management	\$307,851	\$220,522	\$258,045	\$194,821	
Other client assets	273,090	192,034	227,424	142,018	
Total	<u>\$580,941</u>	<u>\$412,556</u>	<u>\$485,469</u>	<u>\$336,839</u>	

Three Months Ended August 2000 versus Three Months Ended August 1999

Net revenues in Asset Management and Securities Services increased 35% to \$1.09 billion as all major components of the business exhibited strong growth. Operating expenses increased 43%, principally due to higher levels of compensation commensurate with growth in net revenues, and increased costs associated with global expansion, higher employment levels and increased business activity. Pre-tax earnings were \$250 million compared to \$223 million in 1999.

Net revenues in Asset Management increased 48%, primarily reflecting a 37% increase in average assets under management as well as favorable changes in the composition of assets managed. Strong net inflows and market appreciation led to the growth in assets under management. Securities Services net revenues were 20% higher, primarily due to increased customer balances in securities lending and margin lending, partially offset by reduced spreads in the fixed income matched book. Commissions increased 34%, primarily due to higher transaction volumes in global equity markets. Revenues from the increased share of income and gains from our merchant banking funds also contributed to the increase in Commissions.

Nine Months Ended August 2000 versus Nine Months Ended August 1999

Net revenues in Asset Management and Securities Services increased 49% to \$3.42 billion, due to increased contributions from all major components of the business. Operating expenses increased 33% due to higher levels of compensation commensurate with growth in net revenues, and increased costs associated with global expansion and higher levels of business activity. Pretax earnings were \$1.06 billion compared to \$519 million in 1999.

Net revenues in Asset Management increased 55%, primarily due to a 35% increase in average assets under management, as well as favorable changes in the composition of assets managed. Securities Services net revenues increased 26%, reflecting the impact of higher customer balances in our securities lending and margin lending businesses. This increase was partially offset by reduced spreads in our fixed income matched book. Commissions increased 58% due to increased worldwide transaction volumes and higher net revenues from the increased share of income and gains from our merchant banking funds.

Operating Expenses

The following table sets forth our operating expenses and number of employees:

Operating Expenses and Employees

(\$ in millions)

	Three Months Ended August			Nine Months Ended August	
	2000	00 1999 2000		1999	
Compensation and benefits, excluding employee initial public offering awards	\$2,263	\$1,704	\$6,587	\$4,932	
Nonrecurring employee initial public offering awards	_	_	_	2,257	
Amortization of employee initial public offering awards	102	115	314	154	
Brokerage, clearing and exchange fees	136	108	419	328	
Market development	126	92	343	247	
Communications and technology	111	75	304	224	
Depreciation and amortization	119	71	322	229	
Occupancy	116	76	312	221	
Professional services and other	181	85	464	297	
Charitable contribution				200	
Total operating expenses	\$3,154	\$2,326	\$9,065	\$9,089	
Employees at period end (1)	18,666	14,454			

⁽¹⁾ Excludes employees of Goldman Sachs' property management subsidiaries. Substantially all of the costs of these employees are reimbursed to Goldman Sachs by the real estate investment funds to which these companies provide property management services.

Three Months Ended August 2000 versus Three Months Ended August 1999

Operating expenses increased 36% to \$3.15 billion, primarily reflecting increased compensation and benefits commensurate with higher net revenues levels.

Compensation and benefits expense increased 33% to \$2.26 billion. The ratio of compensation and benefits to net revenues was 50% for each of the quarters ended August 2000 and August 1999. Employment levels increased 29%, reflecting growth in our businesses. Expenses associated with our temporary staff and consultants were \$183 million, an increase of 71%, reflecting increased global expansion and consulting costs associated with technology initiatives.

Brokerage, clearing and exchange fees increased 26% primarily due to higher levels of trading volumes in equity derivatives and U.S. equities. Market development expenses increased 37%, principally due to higher levels of travel and entertainment costs associated with growth in employment levels and business activity. Communications and technology expenses increased 48%, reflecting higher telecommunications and market data costs associated with higher employment levels. Additional spending on technology initiatives also led to the increase in communications and technology expenses. Depreciation and amortization expenses increased 68% primarily due to additional technology-related equipment expenditures, leasehold improvements and telecommunications equipment needed for our continued global expansion. Occupancy expenses increased 53%, reflecting continued office expansion needed to accommodate growth in employment levels. Professional services and other expenses increased significantly due to increased business activity.

Nine Months Ended August 2000 versus Nine Months Ended August 1999

Operating expenses of \$9.07 billion remained at prior year levels. In 1999, operating expenses included nonrecurring charges associated with Goldman Sachs' conversion to corporate form and related transactions. These nonrecurring charges included \$2.26 billion for employee initial public offering awards and \$200 million for a contribution to The Goldman Sachs Foundation. Excluding the effect of these nonrecurring charges, operating expenses increased 37%.

Compensation and benefits expense increased 34% to \$6.59 billion. The ratio of compensation and benefits to net revenues was 50% for each of the nine-month periods ended August 2000 and August 1999. Expenses associated with our temporary staff and consultants were \$447 million, an increase of 53% compared with 1999, reflecting increased global expansion and consulting costs associated with technology initiatives.

Brokerage, clearing and exchange fees increased 28% primarily due to higher transaction volumes in equity derivatives and U.S. and European equities. Market development expenses increased 39%, principally due to higher levels of travel and entertainment costs associated with growth in employment levels and business activity. Communications and technology expenses increased 36%, reflecting higher telecommunications and market data costs associated with higher employment levels, and additional spending on technology initiatives. Depreciation and amortization expenses increased 41% primarily due to leasehold improvements and technology-related and telecommunications equipment in support of our increased worldwide activities. Occupancy expenses increased 41%, reflecting global office expansion needed to accommodate increased employment levels. Professional services and other expenses increased 56% due to higher levels of business activity.

Provision for Taxes

The provision for taxes for the quarter and nine months ended August 2000 was \$549 million and \$1.64 billion, respectively. Goldman Sachs' effective tax rate for the quarter and year to date was 40%.

Liquidity

Management believes that one of the most important issues for a company in the financial services sector is access to liquidity. Accordingly, Goldman Sachs has established a comprehensive structure to oversee its liquidity and funding policies, which are described below.

Diversification of Funding Sources and Liquidity Planning. Goldman Sachs seeks to maintain broad and diversified funding sources globally. These diversified funding sources include insurance companies, mutual funds, banks, bank trust departments and other asset managers. Management believes that Goldman Sachs' relationships with its lenders are critical to its liquidity.

We access liquidity in a variety of markets in the United States, Europe and Asia. We make extensive use of the repurchase agreement markets and have raised debt publicly as well as in the private placement and commercial paper markets, and through Eurobonds, money broker loans, commodity-based financings, letters of credit and promissory notes. We seek to structure our liabilities to avoid significant amounts of debt coming due on any one day or during any single week or year.

Asset Liquidity. Goldman Sachs maintains a highly liquid balance sheet. Many of our assets are readily funded in the repurchase agreement markets, which generally have proven to be a consistent source of funding even in periods of market stress. A substantial portion of our

inventory turns over rapidly and is marked-to-market daily. We maintain long-term borrowings and stockholders' equity substantially in excess of our less liquid assets.

Excess Liquidity. In addition to maintaining a highly liquid balance sheet and a significant amount of longer-term liabilities to assure liquidity even during adverse conditions, we seek to maintain a liquidity cushion that consists principally of unencumbered U.S. government and agency obligations to ensure the availability of immediate liquidity.

Dynamic Liquidity Management. Goldman Sachs seeks to manage the composition of its asset base and the maturity profile of its funding to ensure that it can liquidate its assets prior to its liabilities coming due, even in times of liquidity stress. We have traditionally been able to fund our liquidity needs through security-based and collateralized funding, such as repurchase transactions and securities lending, as well as short-term and long-term borrowings and equity capital. To further evaluate the adequacy of our liquidity management policies and guidelines, we perform weekly "stress funding" simulations of disruptions to our access to unsecured credit.

Liquidity Ratio Maintenance. It is Goldman Sachs' policy to further manage its liquidity by maintaining a "liquidity ratio" of at least 100%. This ratio measures the relationship between the loan value of our unencumbered assets and our short-term unsecured liabilities. The maintenance of this liquidity ratio is intended to ensure that we could fund our positions on a fully secured basis in the event that we were unable to replace our unsecured debt maturing within one year. Under this policy, we seek to maintain unencumbered assets in an amount that, if pledged or sold, would provide the funds necessary to replace unsecured obligations that are scheduled to mature (or where holders have the option to redeem) within the coming year.

Intercompany Funding. Most of the liquidity of Goldman Sachs is raised by the parent company, Group Inc. The parent company then lends the necessary funds to its subsidiaries and affiliates. We carefully manage our intercompany exposure by generally requiring intercompany loans to have maturities equal to or shorter than the maturities of the aggregate borrowings of the parent company. This policy ensures that the subsidiaries' obligations to the parent company will generally mature in advance of the parent company's third-party long-term borrowings. In addition, many of the advances made to our subsidiaries and affiliates are secured by marketable securities or other liquid collateral. We generally fund our equity investments in subsidiaries with equity capital.

The Balance Sheet

Goldman Sachs maintains a highly liquid balance sheet that fluctuates significantly between financial statement dates. The following table sets forth our total assets, adjusted assets, leverage ratios and book value per share:

	As of	
	August November 2000 1999	
	(\$ in billion per share	ons, except e amounts)
Total assets	\$ 275	\$ 250
Adjusted assets(1)	199	188
Leverage ratio(2)	21.7x	24.7x
Adjusted leverage ratio(3)	15.6x	18.5x
Book value per share(4)	\$26.43	\$20.94

- Adjusted assets represent total assets less securities purchased under agreements to resell, certain securities borrowed transactions and the increase in total assets related to certain provisions of Statement of Financial Accounting Standards No. 125.
- (2) Leverage ratio equals total assets divided by stockholders' equity.
- (3) Adjusted leverage ratio equals adjusted assets divided by stockholders' equity.
- (4) Book value per share was based on common shares outstanding, including restricted stock units granted to employees with no future service requirements, of 480,263,530 as of August 2000 and 484.566.184 as of November 1999.

As of August 2000, we held approximately \$2.43 billion in high-yield debt and emerging market securities and \$2.55 billion in bank loans. These assets may be relatively illiquid during times of market stress. We seek to diversify our holdings of these assets by industry and by geographic location.

As of August 2000, the aggregate carrying value of our principal investments held directly or through our merchant banking funds was \$3.54 billion, which consisted of corporate principal investments with an aggregate carrying value of \$2.63 billion and real estate investments with an aggregate carrying value of \$913 million.

Credit Ratings

Goldman Sachs relies upon the debt capital markets to fund a significant portion of its day-to-day operations. The cost and availability of debt financing is influenced by our credit ratings. Credit ratings are also important to us when competing in certain markets and when seeking to engage in longer-term transactions, including over-the-counter derivatives. A reduction in our credit ratings could increase our borrowing costs and limit our access to the capital markets. This, in turn, could reduce our earnings and adversely affect our liquidity.

The following table sets forth our credit ratings as of August 2000:

	Short-Term Debt	Long-Term Debt
CBRS	A-1 (High)	AA
Fitch	F1+	AA-
Moody's	P-1	A1
Standard & Poor's	A-1+	A+
Thomson Financial BankWatch	TBW-1	AA

Long-Term Debt

As of August 2000, our consolidated long-term borrowings were \$28.53 billion. Substantially all of these borrowings were unsecured and consisted principally of senior borrowings with maturities extending to 2024. The weighted average maturity of our long-term borrowings as of August 2000 was approximately five years. Substantially all of our long-term borrowings are swapped into U.S. dollar obligations with short-term floating rates of interest in order to minimize our exposure to interest rate and foreign exchange movements.

Item 3: Quantitative and Qualitative Disclosures About Market Risk

For a description of our risk management policies and procedures, value-at-risk (VaR) model, including such model's assumptions and limitations, and nontrading risk sensitivity analysis, see Part II, Item 7A "Quantitative and Qualitative Disclosures About Market Risk" in our Annual Report on Form 10-K for the fiscal year ended November 26, 1999 and the information incorporated by reference therein.

PART II: OTHER INFORMATION

Item 1: Legal Proceedings

The following supplements and amends our discussion set forth under Part I, Item 3 "Legal Proceedings" in our Annual Report on Form 10-K for the fiscal year ended November 26, 1999, as updated by our Quarterly Reports on Form 10-Q for the quarters ended February 25, 2000 and May 26, 2000.

Antitrust Matters

The Goldman Sachs Group, Inc. is a named defendant in a purported class action filed on August 17, 2000 in the U.S. District Court for the Southern District of Florida by an alleged issuer in a 1996 initial public offering. The action asserts substantively similar allegations to the New York action which alleges a conspiracy to fix at 7% the discount that underwriting syndicates receive from issuers of shares in certain offerings. On October 2, 2000, defendants, most of which are named in the New York action, moved to transfer the action to the New York federal court.

In the lawsuit alleging a conspiracy to preclude the multiple listing of equity options on the exchanges, certain defendants including Hull Trading Co. L.L.C. have entered into a settlement, subject to court approval, pursuant to which Hull will be required to pay an aggregate of \$2.475 million.

Rockefeller Center Properties, Inc. Litigation

On July 18, 2000, the federal district court granted plaintiffs' motion for leave to file an amended complaint. Defendants have renewed their motion to dismiss with respect to the amended complaint.

Reichhold Chemicals Litigation

The lawsuit brought by Reichhold Chemicals, Inc. and Reichhold Norway ASA was dismissed on August 29, 2000 pursuant to a settlement.

HUD Litigation

The civil action under the *qui tam* provisions of the federal False Claims Act was voluntarily dismissed without prejudice on August 28, 2000.

Laidlaw Bondholders Litigation

Goldman, Sachs & Co. has been named as a defendant in purported class actions filed on September 24, 2000 in the U.S. District Court for the Southern District of New York and on September 22, 2000 in the U.S. District Court for the District of South Carolina arising from certain offerings of debentures by Laidlaw, Inc. during 1997-1999. The defendants include Laidlaw, certain of its officers and directors, the lead underwriters for the offerings (including Goldman, Sachs & Co., which was lead manager in the offerings), and Laidlaw's outside auditors. The offerings included a total of \$1.125 billion principal amount of debentures, of which Goldman, Sachs & Co. underwrote \$286.25 million.

The lawsuits, brought by certain institutional holders of the debentures, allege that the prospectuses issued in connection with the offerings were false and misleading in violation of the disclosure requirements of the federal securities laws. The plaintiffs are seeking, among other things, unspecified damages.

World Online Litigation

On September 11, 2000, several Dutch World Online shareholders as well as a Dutch entity purporting to represent the interests of certain World Online shareholders commenced a proceeding in Amsterdam District Court against "ABN AMRO Bank N.V., also acting under the name of ABN AMRO Rothschild", alleging misrepresentations and omissions relating to the initial public offering of World Online in March 2000. The lawsuit seeks, among other things, the return of the purchase price of the shares purchased by the plaintiffs or unspecified damages. Neither Goldman, Sachs & Co. nor Goldman Sachs International has been named in the proceeding, but the firm and ABN AMRO Rothschild served as joint global co-ordinators of the offering.

Item 2: Changes in Securities and Use of Proceeds

On August 21, 2000, Sumitomo Bank Capital Markets, Inc. exchanged all 7,440,362 shares of its nonvoting common stock, par value \$.01 per share, of Group Inc. for an equal number of shares of common stock. The issuance of common stock upon exchange was not registered under the Securities Act of 1933 in reliance on Section 3(a)(9) thereof as involving exclusively the exchange of one security of Group Inc. for another security of Group Inc. where no commission or other remuneration was paid in connection with the exchange.

Item 5: Other Information

Cautionary Statement Pursuant to The Private Securities Litigation Reform Act of 1995

We have included in this Form 10-Q filing, and from time to time our management may make, statements which may constitute "forward-looking statements" within the meaning of the safe harbor provisions of The Private Securities Litigation Reform Act of 1995. These forward-looking statements are not historical facts but instead represent only our belief regarding future events, many of which, by their nature, are inherently uncertain and outside of our control. It is possible that our actual results may differ, possibly materially, from the anticipated results indicated in these forward-looking statements. Important factors that could cause actual results to differ from those in our specific forward-looking statements include, but are not limited to, the following:

- a decline in general economic conditions or the global financial markets;
- losses due to unidentified or unanticipated risks;
- · competitive pressure, including for our employees;
- a lack of liquidity, i.e., ready access to funds, for use in our business;
- · losses caused by financial or other problems experienced by third parties; and
- volatility or a downturn in the technology and communications sectors.

Additional information regarding these and other important factors that could cause actual results to differ from those in our forward-looking statements is contained in our prospectus, dated August 1, 2000 (as filed with the SEC on August 2), under the caption "Risk Factors".

Forward-looking statements regarding the expected date of completion of the transaction with Spear, Leeds & Kellogg, L.P. are subject to the risk that the closing conditions will not be satisfied, including the risk that the necessary regulatory and other approvals will not be obtained.

Statements about our investment banking transaction backlog also may constitute forward-looking statements. Such statements are subject to the risk that the terms of these transactions may be modified or that they may not be completed at all; therefore, the net revenues that we expect to earn from these transactions may differ, possibly materially, from those currently expected. Important factors that could result in a modification of the terms of a transaction or a transaction not being completed include, in the case of underwriting transactions, a decline in general economic conditions, volatility in the securities markets generally or an adverse development with respect to the issuer of the securities and, in the case of financial advisory transactions, a decline in the securities markets, an adverse development with respect to a party to the transaction or a failure to obtain a required regulatory approval. Other important factors that could adversely affect our investment banking transactions are contained in our prospectus, dated August 1, 2000 (as filed with the SEC on August 2), under the caption "Risk Factors".

Rule 144 Sales Program

On September 22, 2000, our Board of Directors approved a program to permit the former profit participating limited partners of The Goldman Sachs Group, L.P. and the former owners of our Hull subsidiary to sell, in a coordinated manner, a portion of their shares of common stock in accordance with the volume and manner of sale limitations of Rule 144 under the Securities Act of 1933. Sales under the program commenced on September 25, 2000.

Item 6: Exhibits and Reports on Form 8-K

- (a) Exhibits:
- 2.1 Agreement and Plan of Merger, dated as of September 10, 2000, by and between The Goldman Sachs Group, Inc. and SLK LLC.
- 10.1 Form of Indemnification Agreement, dated as of July 5, 2000.
- 10.2 Form of Amendment No. 1, dated as of July 10, 2000, to the Pledge Agreement, dated as of May 7, 1999 (incorporated by reference to Exhibit F to Amendment No. 4 to Schedule 13D, filed July 11, 2000 (No. 005-56295)).
- 10.3 Amendment No. 1, dated as of September 5, 2000, to the Tax Indemnification Agreement, dated as of May 7, 1999.
- 11.1 Statement re computation of per share earnings.
- 12.1 Statement re computation of ratios of earnings to fixed charges.
- 15.1 Letter re Unaudited Interim Financial Information.
- 27.1 Financial Data Schedule.
 - (b) Reports on Form 8-K:

On July 18, 2000, Group Inc. filed a Current Report on Form 8-K reporting the sale of \$1,250,000,000 principal amount of its 7.625% Notes due 2005.

On July 31, 2000, Group Inc. filed a Current Report on Form 8-K reporting the waiver of transfer restrictions to permit its former partners to pledge Group Inc. common stock to obtain loan commitments to invest in certain merchant banking funds sponsored by Group Inc.

On September 11, 2000, Group Inc. filed a Current Report attaching its press release announcing the agreement to combine with Spear, Leeds & Kellogg, L.P.

On September 19, 2000, Group Inc. filed a Current Report on Form 8-K reporting the firm's net earnings for its fiscal third quarter ended August 25, 2000.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

THE GOLDMAN SACHS GROUP, INC.

By: /s/ David A. Viniar

Name: David A. Viniar Title: Chief Financial Officer

By: /s/ SARAH G. SMITH

Name: Sarah G. Smith

Title: Principal Accounting Officer

Date: October 10, 2000

AGREEMENT AND PLAN OF MERGER dated as of September 10, 2000

by and between

The Goldman Sachs Group, Inc.

and

SLK LLC

TABLE OF CONTENTS

SECTI	ON
RECIT	ALS
	ARTICLE I
	Certain Definitions; Interpretation
1.01	Certain Definitions
	ARTICLE II
	The Merger
2.01 2.02 2.03 2.04	The Merger. .9 Closing. 10 Tax Treatment of the Merger. 10 Organization of NewCo. 10
	ARTICLE III
	Consideration; Exchange; Employee Incentives
3.01 3.02 3.03 3.04 3.05 3.06 3.07	Merger Consideration. 10 Exchange Procedures. 11 Withdrawn Member and Managing Director Subordinated Debt Election 12 Election Procedures. 13 Restricted Stock Units Awards. 13 Incremental Liquidity Gain 14 Adjustments to Prevent Dilution. 14
	ARTICLE IV
	Actions Pending the Effective Time
	Forbearances of the Company
	-i-

ARTICLE V

Representations and Warranties

5.01 5.02	Disclosure Schedules	
5.02 5.03 5.04 5.05 5.06	Representations and Warranties with Respect to the Company	9 4 4
	- <u>-</u>	
	ARTICLE VI	
	Covenants	
6.01	Reasonable Best Efforts	7
6.02	Access; Information	
6.03	No Rights Triggered	
6.05	Regulatory Compliance	
6.06	Performance Ranking	
6.07 6.08	Notification of Certain Matters	
6.09	Fee Agreements.	
6.10	Private Placement4	
6.11 6.12	Employee Benefits	
	ARTICLE VII	
	TAX MATTERS	
7.01	Tax Representations	1
7.02	Covenants4	
7.03	Termination of Tax Sharing Agreements	
7.04	Seller Tax Indemnification	

SECTIO	SECTION PAGE			
7.06 7.07	Survival of Obligations			
	ARTICLE VIII			
	Conditions to Consummation of the Merger			
8.01 8.02 8.03	Conditions to Each Party's Obligation to Effect the Merger			
	ARTICLE IX			
	Indemnification			
9.01 9.02 9.03	Indemnification			
	ARTICLE X			
	Termination			
	Termination			
	ARTICLE XI			
	Miscellaneous			
11.02 11.03 11.04 11.05 11.06	Entire Understanding; No Third-Party Beneficiaries			

Annex	1	Members
Annex	2	Partners
Annex	3	Form of Member Agreement
Annex	4	Form of Custody Agreement
Annex	5	Intentionally Omitted
Annex	6	Retention RSUs
Annex	7	SLK RSUs
Annex	8	Acquiror Subordinated Note

Note: Certain of these annexes have been omitted from this Exhibit 2.1 pursuant to Item 601(b)(2) of Regulation S-K. The Registrant hereby agrees to furnish supplementally a copy of any omitted annex to the SEC upon request.

AGREEMENT AND PLAN OF MERGER, dated as of September 10, 2000, by and between The Goldman Sachs Group, Inc. ("Acquiror") and SLK LLC (the "Company").

RECITALS

- A. The Company. The Company is a New York limited liability company having its principal place of business in New York, New York. Annex 1 lists the names and profit and loss allocations of all the members of the Company (each, a "Member", and collectively, the "Members").
- B. The Partnership. Spear, Leeds & Kellogg, L.P. (the "Partnership") is a New York limited partnership having its principal place of business in New York, New York. The Company is the sole general partner of the Partnership. Annex 2 lists the names and profit and loss allocations of all partners of the Partnership (each, a "Partner" and collectively, the "Partners").
- C . Acquiror Acquiror is a Delaware corporation, having its principal place of business in New York, New York.
- D. The Merger. Subject to the terms and conditions contained in this Agreement, the Company and Acquiror intend to effect the merger (the "Merger") of a direct or indirect wholly owned subsidiary of Acquiror that will be organized as a New York limited liability company ("NewCo") with and into the Company, with the Company being the limited liability company surviving such merger.
- E. Supplemental Agreements. Each of the Members identified as such in Annex 1 will, prior to Closing, enter into an instrument substantially in the appropriate form of Annex 3, in which each Member will have adopted and consented to this Agreement and the transactions contemplated hereby (each, a "Member Agreement") and to the Supplemental Members Agreement which is attached as an Annex to the Member Agreement (the "Supplemental Members Agreement"). (The Member Agreements and the Supplemental Members Agreement referred to in this Recital E, together with the Custody Agreements and the counterparts to the Acquiror Shareholders Agreement to be executed by the Members pursuant to Section 2(e) of the Member Agreement, are referred to collectively as the "Supplemental Agreements".)
- F. SLK Investing Co. The Members directly or indirectly holding all of the partnership interests in SLK Investing Co. will agree simultaneously herewith to cause SLK Investing Co. to transfer all of the Class A limited partnership interest it owns in the Partnership to Acquiror or its designee at the Effective Time.

NOW, THEREFORE, in consideration of the premises, and of the mutual covenants, representations, warranties and agreements contained herein and in the Supplemental Agreements, the parties agree as follows:

ARTICLE I

CERTAIN DEFINITIONS: INTERPRETATION

1.01 Certain Definitions. The following terms are used in this Agreement and the Supplemental Agreements with the meanings set forth below:

"Acquiror" has the meaning assigned in the preamble to this $\mbox{\sc Agreement.}$

"Acquiror Common Stock" means the Common Stock, par value \$0.01 per share, of Acquiror.

"Acquiror Financial Statements" shall have the meaning assigned in Section 5.05(d).

"Acquiror Party" means each of Acquiror, its respective directors, officers, employees, agents and controlling persons, and each of the heirs, executors, successors and assigns of any of the foregoing and, without duplication, the Surviving LLC (effective at the Effective Time).

"Acquiror Shareholders Agreement" means the Shareholders' Agreement, dated May 7, 1999, among Acquiror and the other parties named therein.

"Acquiror Subordinated Notes" has the meaning assigned in Section 3.03.

"Affiliate" means, with respect to any specified person, any other person directly or indirectly controlling or controlled by or under direct or indirect common control with such specified person. For the purposes of this definition, (a) "control" when used with respect to any specified person means the power to direct the management and policies of such person, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise; and (b) the terms "controlling" and "controlled" have meanings correlative to the foregoing.

"Agreement" means this agreement, as amended or modified from time to time in accordance with Section 11.02.

"business day" means any day other than a Saturday, a Sunday or a day on which banks in New York City are authorized or obligated by law or executive order to close.

"Cash Consideration" has the meaning assigned in Section 3.01(a).

"Cash Pool Amount" means \$2.1 billion reduced by the sum of (A) any amount of the Withdrawn Member Amount that is exchanged for cash or Acquiror Subordinated Notes pursuant to Section 3.03 and (B) any amount paid by Acquiror pursuant to Section 8.03(g) to acquire all of the outstanding Class A limited partnership interests of the Partnership and

(C) any charitable contribution, but in no event in excess of \$100,000,000, the Members may cause the Company or the Partnership to make after the date hereof.

"CBOE" means the Chicago Board Options Exchange, Incorporated.

"CFTC" means the United States Commodity Futures Trading Commission.

"Chosen Courts" has the meaning set forth in Section 11.07(a).

"Clearing Agreement" has the meaning assigned in Section $5.03\,(h)$.

"Closing" and "Closing Date" have the meanings assigned in Section 2.02.

"Closing Stock Price" has the meaning assigned in Section 3.06.

"Code" means the Internal Revenue Code of 1986, as amended, and the rules and regulation thereunder.

"Company" has the meaning assigned in the preamble to this $\mbox{\sc Agreement.}$

"Company Membership Interests" means the membership interests in the Company. $\,$

"Compensation Plans" has the meaning assigned in Section 5.03(o).

"Confidentiality Agreement" means the letter agreement, dated March 6, 2000, between the Partnership and the Acquiror.

"Consideration Percentage" means, with respect to any Member, a percentage on a schedule previously agreed to by the Company and Acquiror. The Company may deliver to Acquiror a revised schedule changing the Consideration Percentage of any Member or Members if necessary to reflect any changes in capital at least five (5) days prior to the Closing.

"Constitutive Documents" means, with respect to any person, such person's articles or certificate of incorporation and by-laws, limited liability company agreement or operating agreement, partnership agreement or other constitutive documents.

"Contract" means, with respect to any person, any agreement, indenture, undertaking, debt instrument, contract, lease, understanding, arrangement, or commitment to which such person or any of its Subsidiaries is a party or by which any of them may be bound or to which any of their properties may be subject.

"Custodian" has the meaning assigned in the form of Custody Agreement.

"Custody Agreements" means, collectively, a Custody Agreement to be entered into by each of the Members and each other person who will receive Acquiror Common Stock in the Merger or by election pursuant to Section 3.03, in substantially the form of Annex 4.

"Disclosure Schedule" has the meaning assigned in Section 5.01.

- "Effective Time" has the meaning assigned in Section 2.01(e).
- "Election Deadline" has the meaning assigned in Section 3.04.
- "Employees" has the meaning set forth in Section 5.03(o).
- "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" has, with respect to any person, the meaning assigned in Section 5.03(o).

"ERISA Plans" has the meaning assigned in Section 5.03(k).

"Excess Shares" shall mean , initially, that number of shares of Acquiror Common Stock equal to the quotient derived by dividing (I)the product of (A) the amount by which the Closing Stock Price exceeds \$128.025, times (B) the Stock Pool Amount, times (C) 0.27 by (II) the Closing Stock Price, provided, that, if the sale price of shares of Acquiror Common Stock sold pursuant to Section 3.05 is different from the Closing Stock Price, such number shall be adjusted so as to enable the Members to receive net proceeds (without regard to income taxes payable in respect of such sales) not less than the product described in clause (I) above (it being understood that if, at the time of any sales of shares of Acquiror Common Stock during the same tax year as the Closing, the sale price(s) of the shares is less than the Closing Stock Price, then the number of Excess Shares shall be adjusted appropriately to reflect the reduction in the capital gains tax payable as a result of the Merger after taking into account any capital loss on any sale of shares of Acquiror Common Stock (whether usable or not) at a price below the Closing Stock Price, in each case assuming a 27% effective tax rate.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder.

"Financial Statements" has the meaning assigned in Section $5.03\left(g\right)$.

- "Focus Report" has the meaning assigned in Section 5.03(g).
- "Foreign Plans" has the meaning assigned in Section 5.03(o).
- "Governmental Authority" means any court, administrative agency or commission or other federal, state, local or foreign governmental authority or instrumentality.

- "Indemnified Party" has the meaning assigned in Section $9.02\,(a)$.
- "Indemnifying Party" has the meaning assigned in Section $9.02\,(a)$.
- "Intellectual Property" has the meaning assigned in Section $5.03\,(h)$.
- "Investment Advisers Act" means the Investment Advisers Act of 1940, as amended, and the rules and regulations thereunder.
- "Investment Company Act" means the Investment Company Act of 1940, as amended, and the rules and regulations thereunder.
 - "IRS" means the United States Internal Revenue Service.
- "Lien" means a charge, mortgage, pledge, security interest, restriction (other than a restriction on transfer arising under Securities Laws), claim, lien, or encumbrance of any nature whatsoever, except for liens incurred in the ordinary course of business in respect of Taxes for which adequate reserves have been established.
- "Limited Partnership Interest" means an ownership interest in the Partnership held by the limited partners of the Partnership.
 - "Litigation" has the meaning assigned in Section 5.03(j).
 - "Losses" has the meaning assigned in Section 9.01(a).
- "Managing Director Subordinated Note" means the subordinated and secured demand notes substantially all of which are held by current/retired Managing Directors (as such term is used in the Offering Memorandum), or related entities, and are included within the caption "Junior Subordinated Debt" set forth in the Capitalization section of the Offering Memorandum, having a principal amount not in excess of \$210 million as of the date hereof.
- "Material Adverse Effect" means, with respect to Acquiror, the Company or the Surviving LLC, respectively, an effect or change that, individually or in the aggregate with other such effects or changes, is both material and adverse with respect to the respective financial condition, results of operations, assets, business or management of Acquiror and its Subsidiaries, the Company and its Subsidiaries, or the Surviving LLC and its Subsidiaries, as applicable, and in each case taken as a whole.
- "Material Contract" has the meaning assigned in Section 5.03(h).
- "Member" and "Members" have the meaning assigned in the Recitals.
 - "Member Agreement" has the meaning assigned in the Recitals.

- "Merger" has the meaning assigned in the Recitals.
- "Merger Certificate" has the meaning assigned in Section 2.01(e).
- "Merger Consideration" has the meaning assigned in Section 3.01(a).
 - "NewCo" has the meaning assigned in the Recitals.
 - "NYSE" means the New York Stock Exchange, Inc.
- "Offering Memorandum" has the meaning assigned in Section 5.03(y).
- "Operating Agreement" means the Amended and Restated Operating Agreement of SLK LLC, dated as of August 1, 2000, as amended from time to time.
 - "Order" has the meaning assigned in Section 8.01(b).
- "Owner" means a Member or Partner, as applicable, and "Owners" means the Members and Partners collectively.
- "Partner" and "Partners" have the meaning assigned in the Recitals.
 - "Partnership" has the meaning assigned in the Recitals.
- "Partnership Agreement" means the Spear, Leeds & Kellogg Amended and Restated Articles of Partnership, dated as of January 1, 2000.
 - "Partnership Interest" means an ownership interest in the Partnership.
 - "Pension Plan" has, with respect to any person, the meaning assigned in Section 5.03(o).
 - "person" means any individual, bank, corporation, limited liability company, partnership, joint venture, association, joint-stock company, business trust, governmental entity, or unincorporated organization.
 - "Plans" has the meaning assigned in Section 5.03(o).
 - "Pre-Closing Period" has the meaning assigned in Section $7.02\,(b)$.
 - "Pre-Closing Taxes" has the meaning assigned in Section $7.02\,(\mathrm{b})$.
 - "Previously Disclosed" has the meaning assigned in Section 5.01.

"Regulatory Orders" has the meaning assigned in Section 5.03(j).

"Reports" has the meaning assigned in Section 5.03(f).

"Retention RSUs" has the meaning assigned in Section 3.05.

"Rights" means, with respect to any person, any Contracts, securities or obligations convertible into or exercisable or exchangeable for, or giving any person any right to subscribe for or acquire, or any options, calls or commitments relating to, or any equity interest in or equity appreciation right or other instrument the value of which is determined in whole or in part by reference to the market price or value of, shares of capital stock or other equity interest in such person.

 $\mbox{"SEC"}$ means the United States Securities and Exchange Commission.

"SEC Documents" has the meaning assigned in Section 5.05(d).

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations thereunder.

"Securities Laws" means, collectively, the Securities Act, the Exchange Act, the Investment Advisers Act, the Investment Company Act, the Commodity Exchange Act, as amended, and any other federal, state, local or foreign securities, derivatives or commodity laws.

"Self-Regulatory Organization" means the National Association of Securities Dealers, Inc., the American Stock Exchange, the National Futures Association, the CBOE, the Chicago Board of Trade, the NYSE, any Self-Regulatory Organization Previously Disclosed with respect to Section 5.03(f) and any other similar federal, state or foreign self-regulatory body or organization having jurisdiction over the Company or Acquiror (as the case may be), any of their respective Subsidiaries or Affiliates or any of their respective business operations.

"Share Premium" is the sum of (i) the difference between the principal amount of the Managing Director Subordinated Notes exchanged for Acquiror Common Stock and the aggregate value of any Acquiror Common Stock received in exchange therefor and (ii) the difference between the Withdrawn Member Amount exchanged for Acquiror Common Stock and the aggregate value of any Acquiror Common Stock received in exchange therefor, in each case, using \$128.025 as the value of each share of Acquiror Common Stock.

"SLK Representative" means any person designated by the Company in writing by notice to Acquiror prior to the Effective Time.

"SLK RSUs" has the meaning assigned in Section 3.05.

"Stock Consideration" has the meaning assigned in Section 3.01(a).

"Stock Pool Amount" means 34,368,287 shares of Acquiror Common Stock reduced by the number of shares of Acquiror Common Stock elected to be issued in respect of any Managing Director Subordinated Notes or the Withdrawn Member Amount; provided, however, that the first 390,549 shares of Acquiror Common Stock that represent any Share Premium paid on Managing Director Subordinated Notes or in respect of the Withdrawn Member Amount shall not reduce the Stock Pool Amount.

"Subsidiary" has the meaning assigned in Rule 1-02 of Regulation S-X of the SEC.

"Supplemental Members Agreement" has the meaning assigned in the Recitals.

"Supplemental Agreements" has the meaning assigned in the Recitals.

"Surviving LLC" has the meaning assigned in Section 2.01(a).

"Tax Returns" means any return (including, without limitation, information returns, annual reports or other returns in respect of any Compensation Plan), declaration, report, claim for refund, information return or statement relating to any Tax, together with all schedules or attachments thereto, and including any amendments of any of the foregoing.

"Tax" or "Taxes" means all taxes, charges, fees, levies or other assessments, however denominated and whether imposed by a taxing authority within or without the United States, including, without limitation, all net income, gross income, gross receipts, sales, use, ad valorem, goods and services, capital, transfer, franchise, profits, license, withholding, payroll, employment, employer health, excise, estimated, severance, stamp, occupation, property or other taxes, custom duties, fees, assessments or charges of any kind whatsoever, together with any interest and any penalties, additions to tax or additional amounts imposed by any taxing authority.

"Third Party Claim" has the meaning assigned in Section 9.02(a).

"Third-Party Intellectual Property Rights" has the meaning assigned in Section 5.03(s).

"Withdrawn Member" shall have the meaning ascribed to such term in the Operating Agreement.

"Withdrawn Member Amount" means the sum of the Shares (as such term is defined in the Operating Agreement) of all Withdrawn Members determined in accordance with the Operating Agreement and all predecessor agreements, which shall not exceed \$165,000,000.

"Withdrawn Member Interest" has the meaning assigned in Section 3.03.

is made in this Agreement to "Recitals", "Sections", "Annexes" or "Schedules", such reference shall be to a Recital of, or Annex or Schedule to, this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and are not part of this Agreement. Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed followed by the words "without limitation". References herein to "transactions contemplated by this Agreement" shall be deemed to include a reference to each transaction contemplated by or provided for in this Agreement and any documents or agreements, including the Supplemental Agreements, entered into in connection herewith. No rule of construction against the draftsperson shall be applied in connection with the interpretation or enforcement of this Agreement. Whenever this Agreement shall require a party to take an action, such requirement shall be deemed to include an undertaking by such party to cause its Subsidiaries, and to use its reasonable best efforts to cause its other Affiliates, to take all necessary and appropriate action in connection therewith.

ARTICLE II

THE MERGER

- 2.01 The Merger. At the Effective Time, the Merger shall occur and in furtherance thereof:
 - (a) Structure and Effects of the Merger. NewCo shall merge with and into the Company, and the separate legal existence of NewCo shall thereupon cease. The Company shall be the surviving limited liability company in the Merger (sometimes hereinafter referred to as the "Surviving LLC") and shall continue to be governed by the laws of the State of New York, and the separate legal existence of the Company with all its rights, privileges, immunities, powers and franchises shall continue unaffected by the Merger. The Merger shall have the effects specified in Section 1004 of the New York Limited Liability Company Law.
 - (b) Certificate of Limited Liability Company. The Certificate of Limited Liability Company of the Surviving LLC shall be the Certificate of Limited Liability Company of the Company as in effect immediately prior to the Effective Time, until duly amended in accordance with the terms thereof and applicable law.
 - (c) Operating Agreement. The operating agreement of the Surviving LLC shall be the operating agreement of the Company as in effect immediately prior to the Effective Time, until duly amended in accordance with the terms thereof and applicable law.
 - (d) Manager. The initial manager of the Surviving LLC shall be the manager of NewCo immediately prior to the Effective Time.

- (e) Effective Time. The Merger shall become effective upon the filing, by the offices of the Secretary of State of the State of New York, of a certificate of merger (the "Merger Certificate") in accordance with applicable law and the issuance of a certificate of merger by the Secretary of State of New York, or at such later date and time as may be set forth in such Merger Certificate (the "Effective Time"). The parties shall use their reasonable best efforts to cause the Merger to become effective as soon as is reasonably practicable, but in any event on a date that is not later than three business days after the last of the conditions set forth in Article VIII shall have been satisfied or waived in accordance with the terms of this Agreement (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of those conditions).
- 2.02 Closing. The closing of the Merger (the "Closing") shall take place at 9:00 a.m. at the offices of Sullivan & Cromwell, 125 Broad Street, New York, New York, or at such other place as the parties shall agree, on the date on which the Effective Time shall occur (the "Closing Date").
- 2.03 Tax Treatment of the Merger. The parties intend that for U.S. federal income tax purposes the Merger be treated as a taxable purchase by Acquiror of the Company Membership Interests owned by the Members including Withdrawn Member Interests (and subordinated debt owned by former Managing Directors of the Company) in exchange for cash and Acquiror Common Stock. For these purposes, and all other U.S. federal income tax purposes, the parties agree that the Acquiror Common Stock subject to the Transfer Restrictions (as defined in the Member Agreements) and received by the Members in the Merger shall be valued for all U.S. federal income tax purposes at 100% of the mean of the high and low of the trading price of the Acquiror Common Stock on the Closing Date.
- 2.04 Organization of NewCo. Before the Effective Time, Acquiror will (a) organize NewCo as a New York limited liability company solely for the purpose of this transaction that will have no material assets and liabilities and will be treated, for federal income tax purposes, as a disregarded entity or partnership and (b) cause NewCo to become a party to this Agreement, to be evidenced by the execution by it of a supplement to this Agreement and delivery thereof to Acquiror and the Company.

ARTICLE III

CONSIDERATION; EXCHANGE; EMPLOYEE INCENTIVES

- 3.01 Merger Consideration. At the Effective Time, automatically and without any action on the part of any party, Member or other person:
 - (a) Outstanding Company Membership Interests. Subject to the terms and conditions set forth in this Agreement, all Company Membership Interests that are outstanding and owned by a Member immediately prior to the Effective Time shall become

and be converted into the right to receive (i) an amount in cash equal to the Consideration Percentage of such Member multiplied by the Cash Pool Amount (the "Cash Consideration"), and (ii) a number of shares of Acquiror Common Stock equal to the Consideration Percentage of such Member multiplied by the Stock Pool Amount (the "Stock Consideration" and, collectively with the Cash Consideration, the "Merger Consideration"), subject to adjustment as provided in paragraph (c) of this Section 3.01.

- (b) Outstanding NewCo Membership Interests. The membership interests of NewCo outstanding immediately prior to the Effective Time shall be converted into an identical number of Membership Interests in the Surviving LLC.
- (c) Fractional Shares. Notwithstanding any other provision in this Agreement, no fractional shares of Acquiror Common Stock and no certificates or scrip therefor, or other evidence of ownership thereof, will be issued in the Merger or otherwise hereunder; instead, Acquiror shall pay to each holder of Company Membership Interests who otherwise would be entitled to a fractional share of Acquiror Common Stock an amount in cash (without interest) determined by multiplying such fraction by \$128.025.
- (d) Rights as Members. At the Effective Time, holders of Company Membership Interests shall have no rights as members of the Company or otherwise with respect to their ownership therein other than the right to receive their allocated share of the Merger Consideration provided for in this Article III.
- (e) Merger Consideration Payment. The Stock Consideration shall be paid to the Members as provided for herein and held in custody pursuant to the Custody Agreements. It is a condition to any Member receiving the Member's allocated share of the Merger Consideration that the Member execute and deliver a Member Agreement and a Custody Agreement.
 - 3.02 Exchange Procedures.
 - (a) Delivery of Merger Consideration. Acquiror will:
- (1) prior to the Effective Time, cause to be delivered to each Member any necessary or appropriate transmittal materials for use in paying the Merger Consideration to such Member on the Closing Date.
- (2) promptly thereafter cause to be delivered to the Custodian on behalf of each Member certificates representing the Stock Consideration into which such Member's Company Membership Interests have been converted and shall on the Closing Date deliver by wire transfer to each Member the cash that such Member is entitled to receive pursuant to the provisions of this Article III in respect of the Cash Consideration, fractional shares and dividends or distributions; provided that certificates representing Acquiror Common Stock will not be issued to or on behalf of, and such wire transfer will not be made to, any Member

until the Member has properly completed and returned any transmittal materials delivered by Acquiror pursuant to Section 3.02(a)(1) and has executed and delivered to Acquiror a Custody Agreement.

- (b) Registration of Acquiror Stock. All Acquiror Common Stock to be delivered on behalf of a Member will be registered in the name of the Custodian or its nominee and all checks to be issued to a Member will be made out in the name of such Member.
- (c) No Interest. No interest will be paid on any consideration provided for in this Article III, including the Merger Consideration, cash in lieu of fractional shares or dividends or distributions on any of the preceding.
- (d) No Rights Until Surrender. No dividends or other distributions on Acquiror Common Stock will be paid in respect of any Acquiror Common Stock deliverable on behalf of a Member as a result of the Merger until the Custodian is properly entitled to receive certificates representing such Acquiror Common Stock in accordance with this Section 3.02, and no Member will be eligible to vote such Acquiror Common Stock until the Custodian becomes so entitled. After the Custodian becomes entitled to receive certificates in accordance with this Section 3.02, the Member will also be entitled to receive dividends or distributions with a record date after the Effective Time that have become payable on the Acquiror Common Stock represented thereby, without interest.
- 3.03 Withdrawn Member and Managing Director Subordinated Debt Election. Each Withdrawn Member with a Share (as such term is defined in the Operating Agreement) (a "Withdrawn Member Interest") and/or any person that holds a Managing Director Subordinated Note may make an election to exchange such Withdrawn Member Interest and/or Managing Director Subordinated Note for:
 - (a) a subordinated note, substantially in the form set forth in Annex 8 (an "Acquiror Subordinated Note"), of like principal amount plus cash in an amount equal to all accrued and unpaid interest as of the date of the exchange;
 - (b) (A) that number of shares of Acquiror Common Stock equal to the quotient of (i) the product of 1.5 multiplied by the sum of the dollar amount of such Withdrawn Member Interest and/or of the principal amount of the Managing Director Subordinated Note, as applicable, being exchanged for Acquiror Common Stock divided by (ii) \$128.025 and (B) cash in an amount equal to all accrued and unpaid interest as of the date of the exchange; or
 - (c) cash in an amount equal to the dollar amount of such Withdrawn Member Interest and/or the principal amount of the Managing Director Subordinated Note, plus cash in an amount equal to all accrued and unpaid interest thereon as of the date of the exchange.

In the event that a person entitled to make the foregoing election fails to do so prior to the Election Deadline, such person shall only be entitled to receive in exchange for such Withdrawn Member Interest and/or Managing Director Subordinated Note, cash in an amount equal to the dollar amount of such Withdrawn Member Interest and/or the principal amount, of the Managing Director Subordinated Note, plus cash in an amount equal to all accrued and unpaid interest thereon as of the date of the exchange.

Election Procedures. Each person entitled to make the 3.04 election referred to in Section 3.03 above shall (i) make the election by written notice delivered to Acquiror not earlier than the twentieth business day after the date hereof and no later than the twenty-fifth business day after the date hereof, subject to the Company's ability to extend for an additional 10 business days (the "Election Deadline") and (ii) at the time of such election, execute and deliver to Acquiror a retired member agreement (to the extent such person does not sign a Member Agreement) which shall contain terms substantially identical to the terms contained in the Member Agreement, except that such retired member agreement shall not contain the terms set forth in Sections 1 (except with respect to termination), 2(a), 2(e), 3(i), 5 through 13 and 22 of the Member Agreement. Any election may be revoked until the Election Deadline by written notice to the Acquiror received prior to such Election Deadline, at which time such election shall be irrevocable. In the event that either a Withdrawn Member or a holder of a Managing Director Subordinated Note fails to make the election referred to in Section 3.03 by the Election Deadline with respect to a Withdrawn Member Interest or a Managing Director Subordinated Note, then the Company shall (i) in the case of a Withdrawn Member Interest, redeem such amount in accordance with the terms of the Operating Agreement, and (ii) in the case of a Managing Director Subordinated Note, have such note redeemed by the Company, in each case, effective as of the Effective Time. In connection therewith, the Company agrees to take, or cause to be taken, all necessary steps to redeem pursuant to the terms thereof, as of the Effective Time, all outstanding Withdrawn Member Interests or Managing Director Subordinated Notes not subject to an effective election (it being understood that Section 3.03 and 3.04 will not result in a duplication of amounts owed to Withdrawn Members or holders of Managing Director Subordinated Notes). Shares of Acquiror Common Stock and Acquiror Subordinated Notes issued pursuant to this Article III shall be issued at or promptly after the Effective Time consistent with the payment of the Merger Consideration pursuant to Section 3.02.

3.05 Restricted Stock Units Awards. Effective as of the Effective Time, Acquiror shall grant to certain employees (which may include Members who may or may not be employees) of the Company or its Subsidiaries who remain employed by the Company or its Subsidiaries through the Effective Time in substantially the same capacity as on the date of this Agreement and become employees of Acquiror or one of its Subsidiaries an aggregate of 5,467,682 Acquiror restricted stock units, as such may be equitably adjusted pursuant to the terms of Section 3.07, having the terms set forth in Annex 6 (the "Retention RSUs"). In addition, effective as of the Effective Time, Acquiror shall grant to employees of the Company or any of its Subsidiaries who remain employed by the Company or such Subsidiary through the Effective Time and become employees of Acquiror or one of its Subsidiaries an aggregate of 1,562,195 Acquiror restricted stock units, as such may be equitably adjusted pursuant to the terms of Section 3.07, having the terms set forth in Annex 7 (the

"SLK RSUs"). Acquiror and the Company shall mutually agree prior to the Effective Time on which employees shall be granted SLK RSUs and the number of SLK RSUs to be granted to each such employee. The Company shall consult with Acquiror regarding which employees should be granted Retention RSUs and the number of Retention RSUs that should be granted to each such employee, and after such consultation the Company shall determine, with Acquiror's written consent (which shall not be unreasonably withheld), which employees shall receive the Retention RSUs and how many Retention RSUs are allocated to each such employee. Moreover, Acquiror shall have no obligation to issue Retention RSUs or SLK RSUs to any employee who has not executed and delivered to Acquiror an agreement in form and substance approved by Acquiror imposing on such employee confidentiality, noncompetition and nonsolicitation obligations comparable to those contained in Annex 3. The issuance and delivery of Retention RSUs and SLK RSUs shall be conditioned upon the recipient's executing and delivering to Acquiror a custody agreement in the form customarily used by Acquiror for recipients of restricted stock units. All Retention RSUs and SLK RSUs shall be granted pursuant to the Acquiror 1999 Stock Incentive Plan. All material communications by the Company in respect of the grant of Retention RSUs and SLK RSUs shall be approved in advance by Acquiror.

3.06 Incremental Liquidity Gain. In the event that the closing per share price of Acquiror Common Stock on the trading day immediately preceding the Closing Date (the "Closing Stock Price") exceeds \$128.025, Acquiror shall as promptly as reasonably practicable following the Closing Date, file a registration statement on an appropriate form under the Securities Act of 1933 to register the resale of the Excess Shares. Acquiror will use reasonable best efforts to cause such registration statement to become effective as soon as reasonably practicable. The Excess Shares shall be sold through Goldman, Sachs & Co. in such manner as Goldman, Sachs & Co. shall determine in its sole discretion. The Members whose Excess Shares are included in the registration statement shall pay all of their and Acquiror's out-of-pocket expenses incurred in connection with the preparation and filing of the registration statement and the resale of the Excess Shares. Acquiror may delay any such filing, effectiveness or offering if Acquiror in good faith believes that such filing, effectiveness or offering would interfere with a proposed corporate transaction (including an acquisition or disposition) or offering of securities by Acquiror or any stockholder or require Acquiror to disclose material, nonpublic information and will use reasonable best efforts to effect such filing, effectiveness or offering once such transaction or offering is completed or abandoned or such information is disclosed.

3.07 Adjustments to Prevent Dilution. In the event that Acquiror changes the number of shares of Acquiror Common Stock, or securities convertible or exchangeable into or exercisable for shares of Acquiror Common Stock, issued and outstanding prior to the Effective Time as a result of a reclassification, stock split (including a reverse split), stock dividend or extraordinary distribution or other similar transaction, the Stock Consideration, the number of Retention RSUs, the number of SLK RSUs, and references to \$128.025 where used as a calculation element or reference price shall be equitably adjusted so as to give effect to the intended purpose of this Agreement.

ARTICLE IV

ACTIONS PENDING THE EFFECTIVE TIME

- 4.01 Forbearances of the Company. Until the Effective Time, except as Previously Disclosed or as expressly contemplated by this Agreement or any Supplemental Agreement, without the prior written consent of Acquiror, the Company will, and the Company will cause its Subsidiaries to, conduct the business of the Company and its Subsidiaries in the ordinary and usual course and, to the extent consistent therewith, shall use reasonable best efforts to preserve intact their business organizations and assets and maintain their rights, franchises and existing relations with clients, customers, suppliers, employees and business associates. In addition, without Acquiror's written consent (which shall not be unreasonably withheld) the Company will not, and will cause its Subsidiaries and its Subsidiaries' employees, members, partners, agents and other representatives not to:
 - (a) Ordinary Course. Engage in any new activities or lines of business or take any action reasonably likely to have an adverse effect upon the ability of the Company to perform any of its obligations under this Agreement.
 - (b) Equity Interests. Issue, sell or otherwise permit to become outstanding, or authorize the creation of, any additional equity interests of the Company or any of its Subsidiaries or any Rights in respect thereof or permit any transfers or dispositions of any equity interests in the Company, or any of its Subsidiaries from and after the date of this Agreement; provided, that (i) the Partnership shall be permitted to issue additional Class C limited partnership interests and (ii) First Options of Chicago Inc. shall be permitted to issue additional shares of Class A Preferred Stock, in each case, in the ordinary and usual course of business consistent with past practice.
 - (c) Distributions, Etc. Make, declare, pay or set aside for payment any dividend on or in respect of, or declare or make any distribution on, any shares of capital stock or equity interests of the Company or any of its Subsidiaries, other than distributions from Subsidiaries to the Company or a wholly owned Subsidiary of the Company, or directly or indirectly adjust, split, combine, redeem, reclassify, purchase or otherwise acquire, any shares of its capital stock or equity interests; provided, that the Company shall be permitted to make distributions to its Members (i) in an amount not to exceed \$116,503,000 (including \$50 million reserved for such purposes) in respect of Taxes accrued as of June 30, 2000, (ii) in an amount not to exceed \$35,134,000 in respect of distributions available as of January 1, 2000, (iii) in an amount not to exceed \$26,689,000 in respect of distributions available as of June 30, 2000, (iv) in amounts necessary to permit Members to pay the aggregate net U.S. federal, state and local income taxes imposed on them with respect to the net earnings of the Company during the period commencing on July 1, 2000 and ending immediately prior to the Effective Time (after taking into account Tax credits and other available Tax assets and any prior distributions to fund such Taxes and not including any Taxes payable as the result of sales or other dispositions requiring the recognition of unrealized gains reflected on the June

- 30, 2000 Financial Statements (including the "short-against the box" positions)), (v) in an amount with respect to each Member and taking into account any amounts previously paid to such Member (other than the distributions described in clauses (i) through (iv) and (vi) through (ix) of this subsection (c)) since June 30, 2000, not to exceed 4% of a Member's average capital account balance (as such amount is calculated on a monthly basis in the ordinary and usual course of the Company's business, but excluding any net income earned after June 30, 2000) for the period from July 1, 2000 and ending immediately prior to the Effective Time, (vi) of customary bi-weekly draws, in amounts per Member consistent with past practice and in the ordinary and usual course of business, (vii) for purposes of making charitable contributions not in excess of \$100,000,000, as contemplated by clause (C) of the definition of Cash Pool Amount, (viii) of two percent (2%) charitable giving draws of not more than \$68,000 in the aggregate for all Members and (ix) as previously disclosed on Schedule 5.03(g); provided, however, that the Tax rates and taxable income used for purposes of these computations are not inconsistent with any Tax Returns filed by the Partnership or the Company, the Financial Statements and any other available financial information, and provided, further, that the manner in which the effective Tax rate is computed is subject to Acquiror's reasonable approval and is based upon the highest applicable U.S. federal, state and local rates applicable to individuals.
- (d) Compensation; Employment Agreements; Etc. Enter into, amend, modify or renew any employment, consulting, severance or similar contracts with any director, officer, employee or Member, or grant any salary, wage or other compensation increase or increase any employee payment or benefit (including incentive or bonus payments), except (1) for normal individual increases in compensation to persons other than Directors or Managing Directors in the ordinary course of business consistent with past practice, (2) for other changes that are required by applicable law, (3) to satisfy contractual obligations existing as of the date hereof that have been previously made available to Acquiror, and (4) for employment arrangements for, or grants of awards (other than any ownership interest in the Company or any of its Subsidiaries, or Rights with respect thereto) to, newly hired employees of the Company (other than officers or directors) or any of its Subsidiaries in the ordinary course of business consistent with past practice.
- (e) Benefit Plans. Enter into, establish, adopt or amend or communicate in writing or orally any intention to take such action, or grant any waiver with respect to, any pension, retirement, equity option, equity purchase, savings, profit sharing, deferred compensation, consulting, bonus, group insurance or other employee benefit, incentive or welfare contract, plan or arrangement, or any trust agreement (or similar arrangement) related thereto, in respect of any current of former director, officer, employee or Owner, except (1) for non-material items in the ordinary and usual course of business or by the transactions to be effected by this Agreement or (2) as may be required (i) by applicable law or (ii) to satisfy contractual obligations existing on the date hereof that have been Previously Disclosed and accrued on the Financial Statements, or take any action to accelerate the funding, vesting or exercisability of options, restricted membership interest awards or other compensation or benefits payable thereunder.

- (f) Dispositions. Initiate, solicit or otherwise encourage, directly or indirectly, any (i) enquiries or the making of any proposal or offer with respect to a merger, consolidation or similar transaction involving the Company or any of its Subsidiaries, or any purchase of membership interests or securities of the Company or any of its Subsidiaries, or otherwise participate in or facilitate any effort or attempt to make or effectuate such an offer or proposal including, without limitation, by engaging in any discussions or negotiations or providing any confidential information or data to, any person making such other offer or proposal; and (ii) except for sales, transfers, mortgages, encumbrances or other dispositions of securities or other investments or assets in the ordinary course of business consistent with past practice, sell, transfer, mortgage, encumber or otherwise dispose of or discontinue (1) any material amount of assets or properties or (2) any material business or operations.
- (g) Acquisitions. Except for the purchase of securities or other investments or assets in the ordinary course of business consistent with past practice, acquire any assets, business, or properties of any other entity.
- (h) Governing Documents. Amend the Constitutive Documents of the Company or any of its Subsidiaries.
- (i) Accounting Methods. Implement or adopt any change in its accounting principles, practices or methods, other than as may be required by generally accepted accounting principles or applicable law or regulation.
- (j) Contracts. Without limiting the applicability of Section 4.01(d) or (e), enter into or terminate any material contract (other than a material contract which expires by its terms) or amend, modify or waive in any respect any of its existing material contracts or its rights thereunder other than in the ordinary course of business consistent with past practice.
- (k) Claims. Settle any claim, action or proceeding, except for any claim, action or proceeding involving solely money damages in an amount, individually and in the aggregate for all such settlements, not more than \$1,000,000 and which is not reasonably likely to establish an adverse precedent or basis for subsequent claims, actions, proceedings or settlements.
- (1) Adverse Actions. Knowingly take any action that is reasonably likely to result in (1) any of its representations or warranties set forth in this Agreement being or becoming untrue at any time at or prior to the Effective Time (after giving effect to Section 5.02) or (2) any of the conditions set forth in Article VIII not being satisfied, except, in either case, as may be required by applicable law or regulation.
- (m) Indebtedness. Incur any indebtedness for borrowed money (other than any indebtedness with a final maturity less than 12 months from the date of its issuance incurred in the ordinary course of business consistent with past practice) or issue any debt obligations that would be considered Managing Director Subordinated Notes.

- (n) Commitments. Agree, commit to or enter into any agreement to take any of the actions referred to in Section 4.01(a) through (m).
- 4.02 Forbearances of Acquiror. Until the Effective Time, except as expressly contemplated by this Agreement, without the prior written consent of the Company, Acquiror will not, and will cause each of its Subsidiaries not to, knowingly take any action reasonably likely to result in (1) any of its representations and warranties set forth in this Agreement being or becoming untrue at any time at or prior to the Effective Time (after giving effect to Section 5.02) or (2) any of the conditions set forth in Article VIII not being satisfied except, in either case, as may be required by applicable law or regulation. In addition, Acquiror shall not declare, make or pay any extraordinary dividend or other extraordinary distribution on Acquiror Common Stock, whether in cash or otherwise (which shall not include any quarterly cash dividends in the ordinary and usual course of business) unless an equitable adjustment is made to the Stock Consideration and the calculation element or reference price of \$128.025.

ARTICLE V

REPRESENTATIONS AND WARRANTIES

- 5.01 Disclosure Schedules. On or prior to the date hereof, the Company has delivered to Acquiror, and Acquiror has delivered to the Company, a schedule (respectively, its "Disclosure Schedule") setting forth, among other things, items the disclosure of which is necessary or appropriate either (a) in response to an express informational requirement contained in or requested by a provision hereof or (b) as an exception to one or more representations or warranties contained in Section 5.03, 5.04, 5.05 and 7.01. Information set forth in a party's Disclosure Schedule, whether in response to an express informational requirement or as an exception to one or more representations, warranties or covenants, as applicable, in each case that is contained (or incorporated by reference) in a correspondingly enumerated portion of such Disclosure Schedule or as disclosed in the Offering Memorandum, is described herein as "Previously Disclosed".
- 5.02 Standard. No representation or warranty contained in Section 5.03, 5.04, 5.05 and 7.01 shall be deemed untrue or incorrect, and no party hereto shall be deemed to have breached a representation or warranty, as a consequence of the existence of any fact, event, or circumstance that should have been disclosed as an exception to one or more representations or warranties, unless such fact, event or circumstance, whether individually or taken together with all other facts, events or circumstances that should have been so disclosed (whether or not as exceptions) with respect to such representation or warranty contained in Section 5.03, 5.04, 5.05 or 7.01, has had or is reasonably likely to have a Material Adverse Effect with respect to the Company (in the case of Section 5.03 and Section 7.01), the Partnership (in the case of Section 5.04 and Section 7.01), Acquiror (in the case of Section 5.05); provided, that this Section 5.02 will not apply to, and will not limit breaches of, the first sentence of Section 5.03(a), Section 5.03(b), the last two sentences of the first paragraph of Section 5.03(g) (3) (B), the first sentence of Section 5.03(g) (1), Section 5.03(g) (3) (B), the first sentence of Section 5.04(a), Section 5.04(b), the

first sentence of Section 5.05(a), Section 5.05(b), the last sentence of 5.05(d)(1), 5.05(d)(2)(B), 5.05(d)(3) and the last sentence of Section 5.05(e); and provided, further, that this Section 5.02 will not apply in determining whether indemnification is available under Article VII or Article IX hereof.

- 5.03 Representations and Warranties with Respect to the Company. Except as Previously Disclosed in a paragraph of its Disclosure Schedule corresponding to the relevant paragraph below, the Company hereby represents and warrants to, and agrees with, Acquiror as follows:
 - (a) Organization, Standing and Authority. The Company has been duly organized and is validly existing as a limited liability company in good standing under the laws of the State of New York. The Company is duly qualified to do business and is in good standing in the States of the United States and foreign jurisdictions where its ownership or leasing of property or the conduct of its business requires it to be so qualified. The Company and each of its Subsidiaries has the requisite power and authority and has in effect all federal, state, local, and foreign governmental authorizations necessary for it to own or lease its properties and assets and to carry on its business as it is now conducted. The Company has delivered to Acquiror a true and complete copy of its articles of organization and operating agreement, which are the only Constitutive Documents of the Company.
 - (b) Membership Interests. Annex 1 is an accurate and complete list of the Members of the Company and is an accurate and complete list of their respective ownership interests in all material respects. As of the date of this Agreement, the Company has no Company Membership Interests reserved for issuance and no obligation to admit any other person as a Member. All the outstanding Company Membership Interests are duly authorized and validly issued and were not issued in violation of any subscriptive or preemptive rights. There are no other Company Membership Interests authorized, issued or outstanding and there are no preemptive rights or any outstanding Rights of the Company or any of its Subsidiaries of any character relating to the issued or unissued securities of the Company (including those relating to the issuance, sale, purchase, redemption, conversion, exchange, redemption, voting or transfer thereof). Since June 30, 2000 and until the date hereof, no dividend or other distribution has been paid on or in respect of the Company Membership Interests other than (i) as would have been permitted under Section 4.01(c)(iv), (v) and (vi) had this Agreement been in effect at such time or (ii) as Previously Disclosed on Schedule 5.03(b).

Immediately after Effective Time, Acquiror shall own all of the Membership Interests of the Company.

(c) Subsidiaries. The Company has Previously Disclosed a list of all its Subsidiaries, including the states or foreign jurisdictions in which such Subsidiaries are organized and, if any of such Subsidiaries is not wholly owned by the Company or one or more of its Subsidiaries, the percentage owned by the Company or each such Subsidiary and the names, addresses and type and percentage ownership by any other person. No equity interests in any

of the Company's Subsidiaries are or may become required to be issued (other than to the Company or a wholly owned Subsidiary of the Company) by reason of any Rights with respect thereto. There are no Contracts or outstanding claims assessed against the Company by which any of the Company's Subsidiaries is or may be bound to sell or otherwise issue any equity interests, and there are no Contracts relating to the rights of the Company to vote or to dispose of such equity interests. All of the equity interests in each of the Company's Subsidiaries have been duly authorized and validly issued and, to the extent applicable, are fully paid and nonassessable and are subject to no subscriptive or preemptive rights or Rights and are owned by the Company or one of its Subsidiaries free and clear of any Liens other than those Liens created by the existence of the Constitutive Documents. Each of the Company's Subsidiaries is in good standing under the laws of the jurisdiction in which it is organized, and is duly qualified to do business and in good standing in each jurisdiction where its ownership or leasing of property or the conduct of its business requires it to be so qualified. The Company is the sole general partner of the Partnership and (including the Float (as such term is defined in the Partnership Agreement)), has the right to 98.942% of the Partnership's profits, free and clear of all Liens other than those Liens created by the existence of the Constitutive Documents, and is responsible for 100% of the Partnership's losses, and no other person has any interest, legal, beneficial or otherwise in such Partnership Interest. At the Closing Date, the Company shall have the right to 100% of the Partnership's profits. Upon the consummation of the Merger, the Surviving LLC will continue to own, beneficially and of record, such Partnership Interest, free and clear of all Liens other than those Liens created by the existence of the Constitutive Documents.

The Company has Previously Disclosed a true and complete list of all equity securities (other than those issued by a Subsidiary) it or a Subsidiary of the Company holds or controls as of the date of this Agreement involving, in the aggregate, ownership or control of 5% or more of any class of the issuer's voting securities or 25% or more of the issuer's equity. The Company has Previously Disclosed a list of all corporations (and any other entities or joint undertakings treated as a partnership for any Tax purpose), limited liability companies, joint ventures or similar entities, in which it owns or controls a 5% or more interest, directly or indirectly, and the nature and amount of each such interest.

(d) LLC Action. The Company has the requisite power and authority, and has taken all limited liability company action necessary, (1) to authorize the execution and delivery of and performance of its obligations under this Agreement and (2) to approve and adopt the Merger and in accordance herewith, to consummate the Merger and the transactions contemplated by this Agreement. Without limiting the foregoing, any action of the Members of the Company required to approve or adopt this Agreement and the transactions contemplated by this Agreement has been duly taken in accordance with the requirements of the New York State Limited Liability Company Law and the Operating Agreement. No further action of the Members or Managers of the Company is required in order to consummate the Merger and, except for the execution and delivery of documents referred to herein that are contemplated to be entered into at or prior to Closing, the transactions contemplated by this Agreement. As of the Closing, no Member will have any

dissenter's or similar rights with respect to the Merger. This Agreement has been duly executed and delivered on behalf of the Company and constitutes the valid and legally binding agreement of the Company, enforceable in accordance with its terms except as the enforceability hereof may be limited by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other similar laws relating to the enforcement of creditors' rights generally and by general principles of equity.

- Disclosed required regulatory approvals, and the filing of the Merger Certificate, the execution, delivery and performance of this Agreement and the consummation by the Company of the transactions contemplated by this Agreement, do not and will not (1) conflict with or constitute a breach or violation of, or a default under, or cause or allow the acceleration or creation of any right, obligation or Lien (with or without the giving of notice, passage of time or both) pursuant to any law, rule or regulation or any judgment, decree, order, or any governmental or non-governmental permit, license, franchise or privilege or any Contract of it or any of its Affiliates or to which it or any of its Affiliates or its or their properties is subject or bound, (2) constitute a breach or violation of, or a default under, the Constitutive Documents of the Company or any of its Subsidiaries, or (3) require any consent or approval under any such law, rule, regulation, judgment, decree, order, governmental or non-governmental permit, license, franchise or privilege or the consent or approval of any other party to any such Contract.
- (f) Reports. The Company and its Subsidiaries have timely filed all reports, registrations, statements and other filings (other than those relating to Taxes or Tax Returns), together with any amendments required to be made with respect thereto, that were required to be filed by the Company or any of its Subsidiaries since December 31, 1998 with (1) the SEC or the CFTC, (2) any other applicable Governmental Authorities (other than taxing authorities) or (3) any Self-Regulatory Organization (all such reports and statements, including the financial statements, exhibits and schedules thereto, being collectively referred to herein as the "Reports"), including without limitation, all reports, registrations, statements and filings required under the Securities Laws. Each of the Reports, when filed, complied (or, if filed after the date hereof, will comply) as to form with all applicable statutes, rules, regulations and orders (whether or not enforced or promulgated by the Governmental Authority with which they were filed) and did not (or will not) contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading.

(g) Financial Statements.

(1) The Company has delivered to Acquiror copies of each of the audited consolidated statements of financial condition of the Partnership (including all related notes and schedules thereto) as of September 30, 1998 and 1999, and the audited consolidated statements of income, changes in partner's capital and cash flows of the Partnership (including any related notes

and schedules thereto) for the fiscal years ended September 30, 1997, 1998 and 1999, and the unaudited consolidated financial statements of the Partnership as of, and for the fiscal quarters ended, June 30, 1999 and 2000 and the unaudited unconsolidated financial statements of each of the Subsidiaries of the Partnership that are registered as broker-dealers under Section 15 of the Exchange Act contained in the Financial and Operational Combined Uniform Single Reports (each, a "Focus Report") for the fiscal quarters ended June 30, 1999 and 2000 (collectively, the "Financial Statements"). Each of the statements of financial condition included in the Financial Statements fairly presents in all material respects the consolidated financial position of the property of the relevant Subsidiary on an unconsolidated the case of the Focus Reports, of the relevant Subsidiary on an unconsolidated basis) as of its date, and each of the statements of income and changes in owners' capital and cash flows or equivalent statements included in the Financial Statements fairly presents in all material respects the consolidated results of operations, changes in owners' capital and changes in cash flows, as the case may be, of the Partnership (or, in the case of the Focus Reports, of the relevant Subsidiary on an unconsolidated basis) for the periods set forth therein (subject, in the case of unaudited statements, to normal year-end adjustments and footnote disclosure), in each case in accordance with generally accepted accounting principles consistently applied during the periods involved (except as may be noted therein and except that unaudited statements may not include notes). As of the Closing Date, the only asset of the Company will be its general partnership interest in the Partnership, and the Company will not, except as Previously Disclosed, engage (and will not, except as Previously Disclosed, have engaged since the date hereof) in any business other than owning such general partnership interest, and the Company shall have no liabilities whether accrued, contingent, absolute, determined, determinable or otherwise, other than those directly arising from the Company's ownership of such general partnership interest and the Company shall not have transferred any liabilities to the Partnership or any other Subsidiary.

(2) There are no liabilities as of the date hereof of the Partnership or any of its Subsidiaries of any kind whatsoever that are required to be disclosed on the balance sheet, whether accrued, contingent, absolute, determined, determinable or otherwise, and there is no existing condition, situation or set of circumstances known to the Company or any of its Subsidiaries which could reasonably be expected to result in such a liability, other than:

- (A) liabilities reflected or reserved against in the Financial Statements; and
- (B) liabilities arising, in the ordinary course of business consistent with past practice, after the date of the most recent statement of financial condition included in the Financial Statements that are not and could not reasonably be expected to be materially adverse to the Company or its Subsidiaries.

(3) Since September 30, 1999, (A) the Company and its Subsidiaries have conducted their respective businesses in the ordinary course consistent with past practice and (B) there has not occurred any change, occurrence or event, and no change, occurrence or event has

become reasonably likely, which has had, or is reasonably likely to have, a Material Adverse Effect with respect to the Company.

(4) The pricing of securities and loans held in the Company's and its Subsidiaries' trading accounts or securities portfolios and reflected in the financial statements contained in the Focus Report for the quarter ended June 30, 2000 is consistent with past practices.

(h) Contracts.

- (1) The Company has Previously Disclosed, and delivered true and complete copies of, each of the following Contracts to which it or any of its Subsidiaries is a party, by which it or any of its Subsidiaries is bound, or to which its or any of its Subsidiaries' properties are subject (other than trading commitments with customers or counterparties to purchase or sell securities in the ordinary course of business and consistent with past practice):
 - (A) the real property leases relating to its (i) 220,000 square feet lease for 120 Broadway, New York, New York and (ii) 90,000 square feet lease for its Jersey City, New Jersey location;
 - (B) any joint venture, shareholder or other similar agreement or arrangement or any options, rights or obligations to acquire from any person any capital stock, voting securities or equity interests or Rights or securities convertible into or exchangeable for capital stock, voting securities or equity interests of such person;
 - (C) any Contract entered into within the last three (3) years relating to the acquisition or disposition of any material business or operations (whether by merger, sale of stock, sale of assets or otherwise);
 - (D) any Contract providing for existing or future borrowing of money or payment of the deferred purchase price of property in excess of \$10,000,000 (in either case, whether incurred, assumed, guaranteed or secured by any asset) (other than borrowings with a maturity of less than one (1) year);
 - (E) any application, license, franchise or other Contract relating to any trademark and trademark rights, tradename and tradename rights, service mark and service mark rights, service name and service name rights, copyright and copyright rights or patent and patent rights (collectively, the "Intellectual Property");

- (F) any Contract providing for exclusive dealing or that limits the freedom of the Company or any of its Affiliates to compete in any line of business or with any person or in any area;
- (G) any Contract, other than this Agreement and the Supplemental Agreements, between the Company or any of its Subsidiaries and (i) an Owner or any "associates" or members of the "immediate family" (as such terms are respectively defined in Rule 12b-2 and Rule 16a-1 of the Exchange Act) of an Owner, (ii) any other Affiliate of the Company or of an Owner, (iii) any person 5% or more of whose outstanding voting securities are directly or indirectly owned, controlled or held with power to vote by the Company or any Affiliate of the Company or of an Owner or (iv) any officer of the Company or any Affiliate of the Company or any Affiliate of the Company or of an Owner or any "associates" or members of the "immediate family" (as such terms are respectively defined in Rule 12b-2 and Rule 16a-1 of the Exchange Act) of any such Owner; or
- (H) other than the Supplemental Agreements, any Member agreement, voting agreement, voting trust agreement or similar Contract involving the interests of the Company or any of its Subsidiaries.

The Company has made available its forms of agreements relating to the clearing of futures or securities transactions, the custody of assets or the extension of credit (a "Clearing Agreement"). Each Clearing Agreement entered into by the Company or any of its Subsidiaries is not modified in any material respect from such form of Clearing Agreement that has been made available.

Each contract required to be disclosed or made available in respect of this Section 5.03(h) is referred to as a "Material Contract".

(2) Each Material Contract is a valid and binding agreement of the Company (and/or its Subsidiaries party thereto (and/or its Affiliates in the case of clause (F) above)) and, with respect to any such of the Company and its Subsidiaries (and/or its Affiliates in the case of clause (F) above), is in full force and effect, and none of the Company, any of its Subsidiaries (and/or its Affiliates in the case of clause (F) above) or, as of the date hereof, to the knowledge of the Company or any of its Subsidiaries (and/or its Affiliates in the case of clause (F) above), any other party thereto is in default under any such Contract and there has not occurred any event that, with the lapse of time or the giving of notice or both, would constitute such a default (to the knowledge of the Company or any of its Subsidiaries with respect to any event relating to a party other than the Company or its Subsidiaries).

(i) Properties; Securities.

- (1) Except as reflected or reserved against in the Financial Statements, each of the Company and its Subsidiaries has good and marketable title to, or in the case of leased property, has valid leasehold interests in all property and assets purported to be owned or leased by it (whether real or personal, tangible or intangible, and including investment securities and other investments) reflected in the Financial Statements or acquired after the date thereof, except for property and assets sold or transferred since such date in the ordinary course of business consistent with past practices. None of such property or assets (except property or assets sold or transferred since such date in the ordinary course of business consistent with past practices) is subject to any Liens (including Tax-related Liens), except:
 - (A) Liens incurred in the ordinary course of business consistent with past practice;
 - (B) Liens securing liabilities which, in each case, are disclosed or reserved against in the Financial Statements;
 - (C) Liens for Taxes not yet due or payable or being contested in good faith (and, in either case, for which adequate accruals or reserves have been established in the Financial Statements);
 - (D) Mechanics or materialmen liens and similar liens arising by operation of law; or
 - (E) Liens which do not materially detract from the value or materially interfere with any current use of such property or assets.
- (2) To the knowledge of the Company and its Subsidiaries, all buildings and all fixtures, equipment, and other property and assets used but not owned by the Company or any of its Subsidiaries are held under valid leases or subleases enforceable in accordance with their respective terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting creditors' rights generally and by general equity principles.
- (3) Each of the Company and its Subsidiaries has a "security entitlement" (as defined in the Uniform Commercial Code) in all securities or investments held or purported to be held by it (except securities sold under repurchase agreements or held in any fiduciary or agency capacity), free and clear of any Lien, except to the extent such securities are pledged in the ordinary course of business consistent with past practices to secure obligations of each of the Company or any of its Subsidiaries. Such securities are valued on the books of the Company and its Subsidiaries in accordance with generally accepted accounting principles.

- (4) All Managing Director Subordinated Notes, by their terms, are redeemable by the Company at a price equal to the principal amount thereof, plus accrued interest, if any, through the date of such redemption without penalty.
- (j) Litigation; Regulatory Action. Except as Previously Disclosed, no litigation, proceeding or investigation ("Litigation") before or by any court, arbitrator, mediator or Governmental Authority is pending against the Company or any of its Subsidiaries, and, to the knowledge of the Company or any of its Subsidiaries, no such Litigation has been threatened, and there is no existing condition, situation or set of circumstances known to the Company or any of its Subsidiaries which could reasonably be expected to result in such litigation; neither the Company nor any of its Subsidiaries, or any of their properties is a party to or is subject to any order, decree, agreement, memorandum of understanding or similar arrangement with, or a commitment letter or similar submission to, any Self-Regulatory Organization or any Governmental Authority (including the SEC, the CFTC and the Federal Trade Commission) charged with the supervision or regulation of broker-dealers, securities underwriting or trading, stock exchanges, commodities exchanges, alternative trading systems or the supervision or regulation of the Company or any of its Subsidiaries (collectively, "Regulatory Orders"). There is no Litigation involving the Company, and to the Company's knowledge there are no claims, disputes or Litigation involving any Member or former Member, related, in each case, to the ownership of the membership interests of the Company.
- (k) Compliance with Laws. Each of the Company and its Subsidiaries, and their respective Owners, managers, directors, officers and employees:
 - (1) in the conduct of its businesses has been and is in compliance with all applicable federal, state, local and foreign statutes, laws, regulations, ordinances, rules, judgments, orders or decrees applicable thereto or to the employees conducting such businesses, and with the applicable rules of all Self-Regulatory Organizations including without limitation, (w) all applicable regulatory net capital requirements, including SEC Rule 15c3-1 and, as applicable, the "early warning" and "expansion-contraction" capital requirements in NYSE Rules 325 and 326, the liquid net assets requirements of NYSE Rule 104.20 and the capital requirements of any futures exchanges; (x) all rules and regulations relating to the maintenance and preservation of books and records; and (y) the provisions of the Foreign Corrupt Practices Act of 1977, as amended, and the rules and regulations promulgated thereunder;
 - (2) has and has had all permits, licenses, authorizations, orders and approvals of, and has made all filings, applications and registrations with, all Governmental Authorities and Self-Regulatory Organizations that are required in order to permit them to own and operate their businesses as presently conducted; all such permits, licenses, authorizations, orders, franchises and approvals are in full force and effect and, to the Company's and its Subsidiaries' knowledge, no suspension or cancellation of any of them is threatened or reasonably likely; and all such filings,

applications and registrations are current and do not need to be amended in any material respect;

- (3) since January 1, 1997, except as disclosed on the Partnership's or its Affiliates' most recent Form BD filed with the SEC, the Partnership has received no written notification or communication (or, to the Company's and its Subsidiaries' knowledge, any other communication) from any Governmental Authority or Self-Regulatory Organization (A) asserting that any of them is not in compliance with any of the statutes, rules, regulations, or ordinances which such Governmental Authority or Self-Regulatory Organization enforces, or has otherwise engaged in any unlawful business practice except such assertions that were resolved to the satisfaction of such Governmental Authority or Self-Regulatory Organization and did not result in a material change to the operations or practices of the Company or such Subsidiary, (B) threatening to revoke any license, franchise, permit, seat on any stock or commodities exchange, or governmental authorization, (C) requiring any of them (including any of the Company's or its Subsidiaries' Owners, managers, directors, officers or controlling persons) to enter into a cease and desist order, agreement, or memorandum of understanding (or requiring the board of directors (or similar body) thereof to adopt any resolution or policy) or (D) restricting or disqualifying the activities of the Company or any of its Subsidiaries (except for restrictions generally imposed by rule, regulation or administrative policy on brokers or dealers generally); and the Company has delivered true and complete copies of each such written notification or communication to Acquiror;
- (4) is not the subject of any pending or aware of any threatened investigation, review, disciplinary proceedings or unresolved issues raised by any Governmental Authority or Self-Regulatory Organization against or with respect to the Company, any of its Subsidiaries or any Owner, manager, officer, director or employee thereof;
- (5) in the conduct of its business with respect to employee benefit plans subject to Title I of ERISA or Section 4975 of the Code ("ERISA Plans"), has not (A) engaged in any conduct which would subject it or them to liability under Sections 405, 406, 409, 502(i) or 502(l) of ERISA or Section 4975 of the Code or (B) engaged in any conduct that could constitute a crime or violation listed in Section 411 of ERISA which could preclude such person from providing services to any ERISA Plan;
- (6) is not, nor is any Affiliate of any of them, subject to a "statutory disqualification" as defined in Section 3(a)(39) of the Exchange Act; is not subject to a disqualification that would be a basis for censure, limitations on the activities, functions or operations of, or suspension or revocation of, the registration of any broker-dealer Subsidiary as broker-dealer under Section 15, or municipal securities dealer, government securities broker or government securities dealer under Section 15, Section 15B or Section 15C of the Exchange Act; is not, nor are any of their

"principals" (as defined in Section 8a(2) of the Commodity Exchange Act), subject to any of the provisions of Section 8a that would permit the CFTC, subject to the terms of such section, to refuse to register or to suspend or revoke the registration of any of them or their respective principals; and, to the Company's knowledge, there is no current investigation, whether formal or informal, or whether preliminary or otherwise, that is reasonably likely to result in, any such censure, limitations, suspension or revocation; and

(7) has not acted as an underwriter of any securities.

(1) Registrations.

- (1) The Company has previously made available each registration of the Company and any of its Affiliates as a broker-dealer, a registered representative or a sales person (or in a similar capacity) with the SEC, the securities or commodities commission or similar authority of any state or foreign jurisdiction or any Self-Regulatory Organization; each such registration is in good standing (to the extent such concept is applicable) and in full force and effect; and no other such registrations are required in order to permit them to own and operate their businesses as presently conducted.
- (2) Other than as previously made available, none of the Company or its Affiliates, nor any of their respective Owners, directors, managers, officers and employees, is required to be registered as an investment adviser, investment company, commodity trading advisor, commodity pool operator, futures commission merchant, insurance agent, or transfer agent under any United States federal, state, local or foreign statutes, laws, rules or regulations (including any of the Securities Laws).
- (m) Specialist Securities. On the date hereof, the Partnership and certain of its Subsidiaries act as the Self-Regulatory Organization specialist unit for each security Previously Disclosed. Except as Previously Disclosed, the Company has no reason to believe and has not been advised that it may be required to cease acting as the Self-Regulatory Organization specialist unit for any of such securities, nor has it agreed to act as the Self-Regulatory Organization specialist unit for any other security, nor will it so agree without the prior written approval of Acquiror. The Company will use its best efforts to retain its status as the Self-Regulatory specialist unit for each security Previously Disclosed.
- (n) No Brokers. None of the Company or its Subsidiaries, or any of their respective Affiliates, has employed any broker or finder, or incurred any brokers or finders commissions or fees, in connection with the transactions contemplated hereby.

- (1) The Company has Previously Disclosed a list of all significant employment, consulting, benefit and compensation plans, contracts, policies or arrangements and descriptions of any unwritten plans, policies or arrangements covering current or former Members or employees or consultants of the Company or any of its Subsidiaries (the "Employees") and current or former Partners of the Partnership or as to which the Company or any of its Subsidiaries could have, directly or indirectly, any liability, including, but not limited to, "employee benefit plans" within the meaning of Section 3(3) of ERISA, and deferred compensation, stock option, stock purchase, stock appreciation rights, stock based, incentive and bonus plans and any applicable "change in control" or similar provisions in any plan, contract or arrangement (the "Compensation Plans"). True and complete copies of all Compensation Plans, including, but not limited to, any trust instruments and insurance contracts forming a part of any Compensation Plans, and all amendments thereto have been provided or made available to Acquiror. No Compensation Plan provides for any benefit based in whole or in part on the value of any Membership Interest or equity interest owned by the Company or any of its Subsidiaries. For purposes of this paragraph (1), a Compensation Plan is deemed significant if (i) it covers fifty (50) or more individuals or (ii) it requires annual contributions or payments by the Company or any of its Subsidiaries or provides annual benefits aggregating an amount equal to or in excess of \$1,000,000.
- (2) All employee benefit plans covering Employees (the "Plans"), to the extent subject to ERISA, are in substantial compliance with ERISA. Each Plan which is an "employee pension benefit plan" within the meaning of Section 3(2) of ERISA ("Pension Plan") and which is intended to be qualified under Section 401(a) of the Code, has received a favorable determination letter from the Internal Revenue Service with respect to "TRA" (as defined in Section 1 of Rev. Proc. 93-39), and the Company is not aware of any circumstances likely to result in revocation of any such favorable determination letter. There is no material pending or, to the knowledge of the Company threatened, litigation relating to the Plans, and there is no existing condition, situation or set of circumstances known to the Company or any of its Subsidiaries which could reasonably be expected to result in such litigation. Neither the Company nor any of its Subsidiaries has engaged in a transaction with respect to any Plan that, assuming the taxable period of such transaction expired as of the date hereof, could subject the Company or any Subsidiary to a tax or penalty imposed by either Section 4975 of the Code or Section 502(i) of ERISA in an amount which would be material.
- (3) No liability under Subtitle C or D of Title IV of ERISA has been or is expected to be incurred by the Company or any of its Subsidiaries with respect to any ongoing, frozen or terminated "single-employer plan", within the meaning of Section 4001(a) (15) of ERISA, currently or formerly maintained by any of them, or the

single-employer plan of any entity which is considered one employer with the Company under Section 4001 of ERISA or Section 414 of the Code (an "ERISA Affiliate"). Neither the Company, any of its Subsidiaries nor an ERISA Affiliate has contributed to a Pension Plan that is a "multiemployer plan", within the meaning of Section 3(37) of ERISA, at any time on or after September 26, 1980 or could have liability in respect of any such plan. No notice of a "reportable event", within the meaning of Section 4043 of ERISA for which the 30-day reporting requirement has not been waived, has been required to be filed for any Pension Plan or by any ERISA Affiliate within the 12-month period ending on the date hereof or will be required to be filed in connection with the transactions contemplated by this Agreement.

- (4) All contributions required to be made under the terms of any Compensation Plan have been timely made or have been reflected on the Financial Statements. Neither any Pension Plan nor any single-employer plan of an ERISA Affiliate has an "accumulated funding deficiency" (whether or not waived) within the meaning of Section 412 of the Code or Section 302 of ERISA and no ERISA Affiliate has an outstanding funding waiver. Neither the Company nor any of its Subsidiaries has provided, or is required to provide, security to any Pension Plan or to any single-employer plan of an ERISA Affiliate pursuant to Section 401(a)(29) of the Code.
- (5) Under each Pension Plan which is a single-employer plan, as of the last day of the most recent plan year ended prior to the date hereof, the actuarially determined present value of all "benefit liabilities", within the meaning of Section 4001(a)(16) of ERISA (as determined on the basis of the actuarial assumptions contained in the Plan's most recent actuarial valuation), did not exceed the then current value of the assets of such Plan, and there has been no adverse change in the financial condition of such Plan since the last day of the most recent plan year.
- (6) Neither the Company nor any of its Subsidiaries has any obligations for retiree health, life or other benefits under any Compensation Plan.
- (7) The consummation of the transactions contemplated by this Agreement, either alone or in conjunction with any other event, will not (x) entitle any Employees to severance pay, (y) accelerate the time of payment or vesting or trigger any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increase the amount payable or trigger any other material obligation pursuant to, any of the Compensation Plans or (z) result in any payments under, any of the Compensation Plans which would constitute a "parachute payment" to any "disqualified individual" as those terms are defined in Section 280G of the Code; provided, that the representations of this paragraph (7) shall not apply with respect to any awards, payments, compensation and benefits to be paid or made pursuant to this Agreement or other agreements related to the transactions to be effected hereby.

- (8) All Compensation Plans covering current or former non-U.S. employees or otherwise maintained outside the U.S. for U.S. employees are in substantial compliance with applicable local law ("Foreign Plans"). The Company and its Subsidiaries have no unfunded liabilities with respect to any Foreign Plan that are not properly reflected on the Financial Statements:
- (9) There has been no amendment to, written interpretation, communication or announcement by the Company or any of its affiliates relating to, or change in employee participation or coverage under, any Compensation Plan, which would significantly increase the expense of maintaining such Compensation Plan above the level of the expense incurred in respect thereof for the most recent fiscal year; and
- (p) Labor Relations. Each of the Company and its Subsidiaries is in compliance with all currently applicable laws respecting employment and employment practices, terms and conditions of employment and wages and hours, including, without limitation, the Immigration Reform and Control Act, the Worker Adjustment and Retraining Notification Act, any such laws respecting employment discrimination, disability rights or benefits, equal opportunity, plant closure issues, affirmative action, workers' compensation, employee benefits, severance payments, labor relations, employee leave issues, wage and hour standards, occupational safety and health requirements and unemployment insurance and related matters. None of the Company or its Subsidiaries is engaged in any unfair labor practice (within the meaning of the National Labor Relations Act). As of the date hereof, there is no unfair labor practice complaint pending or, to the knowledge of the Company, threatened against any of the Company or its Subsidiaries before the National Labor Relations Board. Neither the Company nor any of its Subsidiaries is a party to, or is bound by, any collective bargaining agreement, contract or other agreement or understanding with a labor union or labor organization, nor is the Company or any of its Subsidiaries the subject of a proceeding as of the date hereof asserting that the Company or any such Subsidiary has committed an unfair labor practice (within the meaning of the National Labor Relations Act) or seeking to compel it or such Subsidiary to bargain with any labor organization as to wages and conditions of employment, nor is there any strike or other labor dispute involving the Company or any of its Subsidiaries, pending or, to the best of the Company's knowledge, threatened, nor is it aware of any activity involving the Company's or any of its Subsidiaries' employees seeking to certify a collective bargaining unit or engaging in any other organization activity.
- (q) Insurance. The Company and its Subsidiaries are insured with reputable insurers against such risks and in such amounts as is in accordance with industry practices. Copies of all insurance policies and related documents have been made available to Acquiror.
- (r) Accounting Controls. Each Subsidiary of the Company that is registered as a broker-dealer has adopted recordkeeping systems that comply with the requirements of Section 17 of the Exchange Act and the rules and regulations promulgated thereunder and

the rules of all Self-Regulatory Organizations having jurisdiction, and maintains its records in accordance therewith. Each of the Company and its Subsidiaries has devised and maintained systems of internal accounting controls sufficient to provide reasonable assurances that (1) all transactions are executed in accordance with management's general or specific authorization; (2) all transactions are recorded as necessary to permit the preparation of financial statements in conformity with generally accepted accounting principles consistently applied with respect to broker-dealers, if applicable, or any other criteria applicable to such statements; (3) access to the property and assets of the Company and its Subsidiaries is permitted only in accordance with management's general or specific authorization; and (4) the recorded amounts for items is compared with the actual levels at reasonable intervals and appropriate action is taken with respect to any differences.

- (s) Proprietary Rights. The Company and its Subsidiaries exclusively own or have the perpetual and exclusive right to use Intellectual Property material to the conduct of their business, and all of these have been Previously Disclosed; and such ownership or right to use such Intellectual Property is free and clear of any Liens, and no other person has the right to use such Intellectual Property. Except as Previously Disclosed: (A) neither the Company nor any of its Subsidiaries is, or will be as a result of the execution and delivery of this Agreement or the performance of its obligations under this Agreement, in violation of any licenses, sublicenses and other agreements as to which the Company or any such Subsidiary is a party and pursuant to which the Company or any such Subsidiary is authorized to use any third-party patents, trademarks, service marks, copyrights, trade secrets, computer software or other proprietary rights (collectively, "Third-Party Intellectual Property Rights") nor shall such agreements be subject to termination or material modification by reason thereof; (B) no claims with respect to (I) the Intellectual Property or (II) Third-Party Intellectual Property Rights are currently pending or, to the knowledge of the Company or any of its Subsidiaries, threatened by any person; (C) neither the Company nor any of its Subsidiaries knows of any valid grounds for any bona fide claims (I) to the effect that the Company or its Subsidiaries is infringing, or has infringed, any Third-Party Intellectual Property Rights; (II) against the use of the Intellectual Property or Third-Party Intellectual
 Property Rights by the Company or any of its Subsidiaries as currently conducted; (III) challenging the ownership, validity or enforceability of any Intellectual Property; or (IV) challenging the license or legally enforceable right to the use of the Third-Party Intellectual Rights by the Company or any of its Subsidiaries; and (D) to the knowledge of the Company or any of its Subsidiaries, there is no unauthorized use, infringement or misappropriation of any of the Intellectual Property by any third party, including any employee or former employee of the Company or any of its Subsidiaries.
- (t) Affiliate Transactions. There is no indebtedness between the Company or any of its Subsidiaries, on the one hand, and any Owner, officer, manager, director or Affiliate (other than the Company or any of its Subsidiaries) of the Company or the Partnership, on the other, other than usual and customary advances made in the ordinary course of business; no such Owner, officer, manager, director or Affiliate provides or causes to be provided any assets, services (other than services as an officer, director or employee) or

facilities to the Company or any of its Subsidiaries; neither the Company nor any of its Subsidiaries provides or causes to be provided any assets, services or facilities to any such Owner, officer, manager, director or Affiliate (other than as reasonably necessary for them to perform their duties as officers, directors or employees); neither the Company nor any of its Subsidiaries beneficially owns, directly or indirectly, any investment in or issued by any such officer, manager, director or Affiliate; and no such Owner, officer, manager, director or Affiliate has any direct or indirect ownership interest in any person with which the Company or any of its Subsidiaries competes or has a business relationship other than an ownership interest that represents less than five percent of the outstanding equity interests in a publicly traded company.

- (u) Clearing Activities. Each of the Company and its Subsidiaries have taken reasonable steps to manage risk in its clearing business including performing due diligence with respect to the character and integrity of customers in the opening of its customer accounts. Since January 1, 1997, no person or Self-Regulatory Organization has made a written claim that the Company or any of its Subsidiaries is responsible or liable for the trading activities of its clearing customers which the Company reasonably believes is likely to give rise to a liability.
- (v) Disclosure. The copies of all documents furnished to Acquiror in connection with this Agreement were accurate and complete copies of the originals thereof, including all amendments and supplements thereto.
- (w) Anti-takeover Statutes. No "fair price,"
 "moratorium," "control share acquisition" or other similar
 anti-takeover statute or regulation enacted under state or federal laws
 in the United States applicable to the Company or any of its
 Subsidiaries is applicable to the Merger or the transactions
 contemplated thereby.
- (x) Derivatives. All swap, forward, future, option, or any other similar agreement or arrangement executed or arranged by the Company or any of its Subsidiaries, whether entered into for the Company's account, or for the account of one or more of the Company's Subsidiaries or their customers were, to the Company's knowledge, entered into (i) in accordance with all applicable laws, rules, regulations and regulatory policies and (ii) with counterparties believed at the time to be financially responsible; and each of them constitutes the valid and legally binding obligation of the Company or any of its Subsidiaries, enforceable in accordance with its terms, except as the enforceability thereof may be limited by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other similar laws relating to the enforcement of creditors' rights generally and by general principles of equity, and are in full force and effect. Neither the Company nor any of its Subsidiaries nor, to the Company's knowledge, any other party thereto, is in breach of any of its obligations under any such agreement or arrangement.
- (y) Offering Memorandum. The Company has delivered to Acquiror a copy of the Partnership's confidential offering memorandum, dated August 16, 2000, for the

Partnership's 8.25% Senior Notes due 2005 (the "Offering Memorandum"). As of the date thereof, the Offering Memorandum did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances in which they were made, not misleading.

- 5.04 Representations and Warranties with Respect to the Partnership. Subject to Section 5.02, except as Previously Disclosed in a paragraph of its Disclosure Schedule corresponding to the relevant paragraph below, the Company hereby represents and warrants to, and agrees with, Acquiror as follows:
 - (a) Organization, Standing and Authority. The Partnership has been duly organized and is validly existing as a limited partnership in good standing under the laws of the State of New York. The Partnership is duly qualified to do business and is in good standing in the States of the United States and foreign jurisdictions where its ownership or leasing of property or the conduct of its business requires it to be so qualified. Each of the Partnership and its Subsidiaries has in effect all federal, state, local, and foreign governmental authorizations necessary for it to own or lease its properties and assets and to carry on its business as it is now conducted. The Partnership has delivered to the Acquiror a true and complete copy of its Certificate of Limited Partnership and its Amended and Restated Articles of Partnership, which are the only Constitutive Documents of the Partnership.
 - (b) Partnership Interests. Subject to such changes in the Class C limited partners of the Partnership permitted by Section 4.01(b) of which the Company shall promptly give notice to Acquiror, Annex 2 lists all of the Partners of the Partnership and the amount of their Partnership Interests, and the Partnership has no Partnership Interests reserved for issuance and no obligation to admit any other person as a general or limited partner. All of the outstanding Partnership Interests have been duly authorized, validly issued, fully paid and, except for the general partnership interest, non-assessable, and were not issued in violation of any subscriptive or preemptive rights. There are no other Partnership Interests outstanding (or claims made against the Partnership or the Company in respect thereof) and there are no preemptive rights or any outstanding rights of any character relating to issued or unissued Partnership Interests (including those relating to the issuance, sale, purchase, redemption, conversion, exchange, redemption, voting or transfer thereof). Since September 30, 1999, no dividend or other distribution has been declared or paid on or in respect of the Partnership Interests or any other interest in the Partnership.
- 5.05 Representations and Warranties of Acquiror. Subject to Section 5.02, except as Previously Disclosed by Acquiror in a paragraph of its Disclosure Schedule corresponding to the relevant paragraph below, Acquiror hereby represents and warrants to the Company as follows:

- (a) Organization, Standing and Authority. Acquiror has been duly incorporated and is validly existing as a corporation in good standing under the laws of the State of Delaware. Each of Acquiror and NewCo is or will be duly qualified to do business and is or will be in good standing in the States of the United States and foreign jurisdictions where its ownership or leasing of property or the conduct of its business requires it to be so qualified. Each of Acquiror and its Subsidiaries has in effect all federal, state, local and foreign governmental authorizations necessary for it to own or lease its properties and assets and to carry on its business as it is now conducted.
- (b) Authority. Each of Acquiror and NewCo has the requisite corporate or other power and authority, and has taken all action necessary, in order (1) to authorize the execution and delivery of, and performance of its obligations under, this Agreement and (2) to approve and adopt the Merger and, in accordance therewith, to consummate the Merger and the other transactions contemplated by this Agreement. Without limiting the foregoing, any action of the stockholders of Acquiror and the members of NewCo required to approve or adopt this Agreement and the transactions contemplated by this Agreement has been duly taken in accordance with the requirements of the Delaware General Corporation Law or the New York Limited Liability Company Law, as the case may be. No further action of the stockholders or board of directors of Acquiror or the members of NewCo is required in order to consummate the Mergers and the other transactions contemplated by this Agreement. This Agreement constitutes the valid and binding agreement of each of Acquiror and NewCo enforceable in accordance with its terms except, in each case, as the enforceability hereof may be limited by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other similar laws relating to the enforcement of creditor's rights generally and by general principles of equity.
- (c) No Defaults. Except as Previously Disclosed, the execution, delivery and performance of this Agreement, and the consummation of the transactions contemplated by this Agreement by Acquiror and NewCo does not and will not (1) conflict with or constitute a breach or violation of, or a default under, or cause or allow the acceleration or creation of any right, obligation or Lien (with or without the giving of notice, passage of time or both) pursuant to, any law, rule or regulation or any judgment, decree, order or governmental or non-governmental permit, license, franchise or privilege or any Contract of Acquiror or of any of its Subsidiaries or to which Acquiror or any of its Subsidiaries or its or their properties is subject or bound, (2) constitute a breach or violation of, or a default under, the Constitutive Documents of Acquiror or any of its Subsidiaries, or (3) require any consent or approval under any such law, rule, regulation, judgment, decree, order or governmental or non-governmental permit, license, franchise or privilege or the consent or approval of any other party to any such Contract.
- (d) SEC Documents and Financial Statements. (1) Acquiror has made available to the Company copies of its Annual Report on Form 10-K for the fiscal year ended November 26, 1999, and its Quarterly Reports on Form 10-Q for the fiscal quarters ended February 25, 2000 and May 26, 2000, each in the form (including exhibits and any

amendments thereto) filed with the SEC (collectively, including any such reports filed subsequent to the date hereof, the "SEC Documents"). As of their respective dates, each of the SEC Documents (including the financial statements, exhibits and schedules thereto), filed, used or circulated prior to the date hereof complied, and any such SEC Documents filed, used or circulated subsequent to the date hereof will comply, as to form with the Exchange Act and did not, and will not, contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made therein, in the light of the circumstances under which they were made, not misleading. Each of Acquiror's consolidated statements of financial condition included in the SEC Documents, including the related notes and schedules thereto, fairly presents, or with respect to such statements included in SEC Documents filed after the date hereof will fairly present, in all material respects, the consolidated financial position of Acquiror and its Subsidiaries as of the date of such statement of financial condition and each of the consolidated statements of income, cash flows and owners' equity included in the SEC Documents, including any related notes and schedules thereto, fairly presents, or with respect to such statements included in SEC Documents filed after the date hereof will fairly present, in all material respects, the consolidated results of operations, cash flows and stockholders' equity of Acquiror and its Subsidiaries for the periods set forth therein (subject, in the case of unaudited statements, to normal year-end audit adjustments and footnote disclosure), in each case in accordance with generally accepted accounting principles consistently applied during the periods involved (except as may be noted therein and except that unaudited statements may not include notes) (the "Acquiror Financial" Statements").

- (2) There are no liabilities as of the date hereof of Acquiror or any of its Subsidiaries of any kind whatsoever that are required to be disclosed on its balance sheet, whether accrued, contingent, absolute, determined, determinable or otherwise, and there is no existing condition, situation or set of circumstances known to Acquiror or any of its Subsidiaries which could reasonably be expected to result in such a liability, other than:
 - (A) liabilities reflected or reserved against in the Acquiror Financial Statements; and
 - (B) liabilities arising in the ordinary course of business consistent with past practice, after the date of the most recent balance sheet included in the Acquiror Financial Statements that are not and could not reasonably be expected to have a Material Adverse Effect with respect to Acquiror.
- (3) Since November 26, 1999, there has not occurred any change, occurrence or event, and no change, occurrence or event has become reasonably likely, which has had, or is reasonably likely to have, a Material Adverse Effect with respect to Acquiror.
 - (e) Capital Stock. The Acquiror has (1) 4,000,000,000 authorized shares of Acquiror Common Stock, of which 440,901,874 were outstanding as of May 26, 2000, (2) 200,000,000 authorized shares of nonvoting common stock, of which 7,440,362 were

outstanding as of May 26, 2000 and (2) 150,000,000 authorized shares of Preferred Stock, of which none are outstanding. All of the outstanding shares of Acquiror Common Stock have been duly authorized and validly issued and are fully paid and nonassessable and were not issued in violation of any subscriptive or preemptive rights. When issued in the Merger in accordance with the terms of this Agreement, the shares of Acquiror Common Stock constituting Merger Consideration will be duly authorized, validly issued, fully paid and nonassessable, shall not have been issued in violation of any subscriptive or preemptive rights and shall have been approved for listing on the NYSE, upon official notice of issuance.

- (f) Compliance with Laws. Except as set forth in the SEC Documents filed prior to the date hereof, the businesses of each of Acquiror and its Subsidiaries have not been, and are not being, conducted in violation of any applicable laws, except as is not reasonably likely to have a Material Adverse Effect on Acquiror. Except as set forth in the SEC Documents filed prior to the date hereof, no investigation or review by any Governmental Authority with respect to Acquiror or any of its Subsidiaries is pending or, to the knowledge of Acquiror, threatened, except as is not reasonably likely to have a Material Adverse Effect on Acquiror. To the knowledge of Acquiror, no material change is required in Acquiror's or any of its Subsidiaries' processes, properties or procedures in connection with any such laws, and Acquiror has not received any notice or communication or any material noncompliance with any such laws that has not been cured as of the date hereof, except as is not reasonably likely to have a Material Adverse Effect on Acquiror.
- 5.06 Representations Regarding NewCo. No representation or warranty contained in Section 5.05 with respect or relating to NewCo will be deemed made until the date on which it executes and delivers to the Company the Supplement to this Agreement contemplated by Section 2.04.

ARTICLE VI

COVENANTS

6.01 Reasonable Best Efforts. (a) Subject to the terms and conditions of this Agreement, each of the Company and Acquiror agrees to use its reasonable best efforts in good faith to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary, proper or desirable, or advisable under applicable laws, so as to permit consummation of the transactions contemplated hereby as promptly as practicable and otherwise to enable consummation of the transactions contemplated hereby and shall cooperate fully with the other parties hereto to that end, including cooperating in seeking to obtain any contractual consents required from third parties; provided, however, that nothing in this Section 6.01 shall require, or be construed to require, Acquiror to proffer to, or agree to, sell or hold separate and agree to sell, before or after the Effective Time, any assets, businesses, or interest in any assets or businesses of Acquiror, the Company or any of their respective Affiliates (or to consent to any sale, or agreement to sell, by the Company of any

of its assets or businesses) or to agree to any material changes or restriction in the operations of any such assets or businesses.

(b) The Company agrees that it shall use its best efforts to obtain, within 20 business days of the date hereof, executed and delivered Member Agreements from Members owning not less than 90% of the total outstanding Company Membership Interests.

6.02 Access; Information.

- (a) The Company agrees that upon reasonable prior notice and subject to applicable laws relating to the exchange of information, and in a manner so as to not interfere with the normal business operations of the Company and its Subsidiaries, it shall afford Acquiror and its officers, employees, counsel, accountants and other authorized representatives, such access during normal business hours throughout the period from the date hereof to the Closing to the books, records (including Tax Returns and work papers, whether prepared by employees, consultants, or independent auditors), properties, personnel and such other information of the Company, and its Subsidiaries as Acquiror (or any such representative) may reasonably request and, during such period, the Company shall furnish to Acquiror (or such other representative) (1) a copy of each material report, schedule and other document filed by the Company or any of its Subsidiaries pursuant to the requirements of Securities Laws, promptly after the filing thereof, (2) monthly Focus Reports not later than the 25th day after the end of each month, commencing with the first Focus Report filed after the date of the Agreement, and (3) all other information concerning the business, properties and personnel of the Company and its Subsidiaries as Acquiror (or any such other representative) may reasonably request, reasonably promptly after such request.
- (b) No investigation by Acquiror of the business and affairs of the Company or its Affiliates shall affect or be deemed to modify or waive any representation, warranty, covenant or agreement in this Agreement, or any conditions to any party's obligation to consummate the transactions contemplated by this Agreement.
- 6.03 No Rights Triggered. The Company shall take all reasonable steps necessary to ensure that the entry into this Agreement and the consummation of the transactions contemplated by this Agreement and any other action or combination of actions, or any other transactions contemplated by this Agreement, do not and will not result in the grant of any Rights to any person (a) under the Constitutive Documents of the Company or any of its Affiliates or (b) under any agreement to which the Company or any of its Affiliates is a party or by which they are bound or to which any of their respective properties are subject.

6.04 Regulatory Applications.

(a) Subject to the proviso contained in Section 6.01, each of Acquiror and the Company shall use its respective reasonable best efforts to prepare all documentation, to effect all filings and to obtain all permits, consents, approvals and authorizations of all third parties, Self-Regulatory Organizations and Governmental Authorities necessary to consummate the

transactions contemplated by this Agreement. Each party will consult with the other party, subject to applicable law, and rules or regulations of any Governmental Authority, with respect to the obtaining of all material permits, consents, approvals and authorizations of all third parties, Self-Regulatory Organizations and Governmental Authorities necessary or reasonably advisable to consummate the transactions contemplated by this Agreement and each will keep the other parties apprised of the status of material matters relating to completion of the transactions contemplated hereby.

- (b) Unless precluded by applicable law, rule or regulation of any Governmental Authority, each of Acquiror and the Company agrees, upon request, to furnish the other party with all information concerning itself, its Subsidiaries, Owners, directors, officers, employees and such other matters as may be necessary or reasonably advisable in connection with any filing, notice or application made by or on behalf of such other party to any third party, Self-Regulatory Organization or Governmental Authority.
 - 6.05 Regulatory Compliance.
- (a) The Company will use its reasonable best efforts to maintain all of the Company's existing material permits, licenses, authorizations, orders and regulatory approvals and the minimum net capital necessary to conduct its businesses as currently conducted by the Company and comply in all material respects with all regulatory requirements applicable to the business of the Company.
- (b) The Company shall provide to Acquiror, promptly after the filing thereof, a copy of each report filed by the Company with any Self-Regulatory Organization.
- 6.06 Performance Ranking. The Company will use its best efforts to operate the specialist business in such a manner that the Company will maintain, at all times prior to the Closing Date, an overall tier classification of not lower than Tier 3 in the quarterly Specialist Performance Evaluation Questionnaire results prepared by the NYSE.
 - 6.07 Notification of Certain Matters.
- (a) Each of Acquiror and the Company shall give prompt notice to the other of any fact, event or circumstance known to it or any of its Subsidiaries that is reasonably likely, individually or taken together with all other facts, events and circumstances known to it, to result in a breach of any of its representations and warranties, after giving effect to Section 5.02, or covenants or agreements contained herein.
- (b) Prior to the Closing, the Company shall promptly notify Acquiror, and Acquiror shall promptly notify the Company, of:
 - (1) any notice or other communication from any person making any offer or proposal referred to in Section 4.01(f) or alleging that the consent of such person is or may be required as a condition to the Closing;

- (2) any notice or other written communications from any client (A) terminating or threatening to terminate any Contract relating to the rendering of services to such client by the Company or any of its Subsidiaries or (B) relating to any dispute with such client by the Company or any of its Subsidiaries, in each case which, individually or in the aggregate, are or are reasonably likely to be material; or
- (3) any notice or other communication from any Governmental Authority or Self-Regulatory Organization in connection with the transactions contemplated by this Agreement.
- (c) Promptly after the date of this Agreement, the Company shall provide to each Member that has not, as of the date of this Agreement, executed and delivered a Member Agreement a written notice that satisfies the requirements of Section 407(c) of the New York Limited Liability Company Law informing such Members that the Members that have executed and delivered a Member Agreement (i) hold in the aggregate voting interests having a majority in interest of the Members and (ii) have adopted and consented to the Merger.
- 6.08 Public Announcements. The parties agree that they will not, without the prior approval of the other parties, issue any press release or written statement for general circulation relating to the transactions contemplated by this Agreement, except as otherwise required by applicable law or regulation or the rules of any applicable Self-Regulatory Organization and, if so required, such party will endeavor to give the other party a reasonable opportunity to review and comment on the statement prior to its release.
- 6.09 Fee Agreements. The Company shall provide to Acquiror, promptly after executing the same, a copy of any written fee agreement or any material contract, agreement or other instrument relating specifically to the operation of the specialist business into which the Company enters after the date hereof.
- 6.10 Private Placement. The parties understand and agree that the sale of shares of Acquiror Common Stock to the Members in the Merger and to Withdrawn Members or holders of Managing Director Subordinated Notes pursuant to Article III is intended to be a "private placement" exempt from registration under the Securities Act by virtue of Regulation D thereof. Each party represents and warrants to the others that, as of the Closing Date, assuming the truth of the representations and warranties made herein and in the Supplemental Agreements by each of the parties thereto, such party has taken no action without the prior written consent of the other parties that would cause (i) the sale of Acquiror Common Stock in the Merger or (ii) the sale of Acquiror Subordinated Notes pursuant to Section 3.03 to become subject to the registration requirements of the Securities Act.

6.11 Employee Benefits. Acquiror agrees that during the period commencing at the Effective Time and ending on the first anniversary thereof, the employees of the Company and its Subsidiaries will continue to be provided with benefits under employee benefit plans (other than stock options or other plans involving the issuance of securities of the Company or Acquiror) which in the aggregate are substantially comparable to those currently provided by the Company to such employees or that are comparable to those provided to similarly situated employees of Acquiror or its Subsidiaries; provided, however, that employees covered by collective bargaining agreements need not be provided with such benefits. Acquiror will cause each employee benefit plan of Acquiror or any of its Subsidiaries in which employees of the Company or any of its Subsidiaries are eligible to participate to take into account for purposes of eligibility and vesting thereunder the service of such employees with the Company or any of its Subsidiaries as if such service were with Acquiror or any of its Subsidiaries, to the same extent that such past service was credited under a comparable plan of the Company or any of its Subsidiaries or is provided generally to employees of Acquiror (and similar treatment shall apply to past employee compensation). Within twenty days of the date hereof, the Company shall furnish a list to Acquiror of all Compensation Plans that are not significant Compensation Plans that were not Previously Disclosed pursuant to Section 5.03(o)(1).

6.12 Indemnification of Members and Employees. For the six-year period immediately following the Effective Time, Acquiror agrees that it will indemnify and hold harmless each Member (in its capacity as an officer or a Managing Director acting in an executive capacity of the Company, SLK Management Inc. or the Partnership) against any Losses incurred in connection with any third party claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of matters existing or occurring at or prior to the Effective Time, whether asserted or claimed prior to, at or after the Effective Time, to the fullest extent that the Company would have been permitted under the New York Limited Liability Company Law to indemnify such person but only if such Member acted in good faith, for a purpose which such Member reasonably believed to be in the best interests of the Company, SLK Management Inc. or the Partnership, as applicable, and, in criminal actions or proceedings, had no reasonable cause to believe that his conduct was unlawful; provided, further, that Acquiror shall have the right to assume and control the defense of such claim and any litigation resulting therefrom) it being understood that such Member shall have the right to participate in such defense and that Acquiror shall not consent to entry of any judgment or entry into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to the Member of a release from all liability in respect to such claim or litigation.

ARTICLE VII

TAX MATTERS

7.01 Tax Representations. Except as Previously Disclosed in a paragraph of its Disclosure Schedule corresponding to the relevant paragraph below, the Company, the Partnership and each of their Subsidiaries hereby represent and warrant to Acquiror as follows:

- (a) All Tax Returns required to be filed or furnished to any person (including its owners) on or before the Closing Date (taking into account applicable extensions) by or with respect to the Company, the Partnership and each of their Subsidiaries or any of their income, properties or operations have been, or will be, duly and timely filed or furnished, the information reflected on those Tax Returns was, or when filed or furnished will be, accurate and complete, and the Company, the Partnership and each of their Subsidiaries have, or will have, timely paid all Taxes other than Taxes resulting from transactions or actions occurring on the Closing Date, but after the Closing, or after the Closing Date for which they are responsible that are due on or before the Closing Date or have provided for such Taxes in a reserve which is adequate for the payment of such Taxes and is identified in the Financial Statements, and through the Closing Date will have maintained adequate provisions on their books for all Taxes other than Taxes resulting from transactions or actions occurring on the Closing Date, but after the Closing, or after the Closing Date payable by them that have accrued but are not yet due;
- (b) There are no outstanding assessments, claims or deficiencies for any Taxes of the Company, the Partnership or any of their Subsidiaries that have been proposed, asserted or assessed in writing;
- (c) No Tax audit or examination is currently being conducted or proposed in writing by any taxing authority with respect to the Company, the Partnership or any of their Subsidiaries;
- (d) There are no outstanding waivers or agreements extending the applicable statute of limitations for any period with respect to any Taxes of the Company, the Partnership or any of their Subsidiaries;
- (e) None of the Company, the Partnership or any of their Subsidiaries will be required, as a result of (A) a change in accounting method for a Tax period beginning on or before the Closing, to include any adjustment under Section 481(c) of the Code (or any similar provision of state, local or foreign law) in taxable income for any Tax period beginning on or after the Closing Date, or (B) any "closing agreement" as described in Section 7121 of the Code (or any similar provision of state, local or foreign law), to include any item of income in or exclude any item of deduction from any Tax period beginning on or after the Closing Date;
- (f) There is no outstanding written claim by a taxing authority that the Company, the Partnership or any of their Subsidiaries may be subject to taxation or required to file a Tax Return in a jurisdiction where it does not file Tax Returns and none of the Company, the Partnership or any of their Subsidiaries is aware of any jurisdiction that could properly make such a claim;
- (g) There are no Tax allocation or sharing agreements to which the Company, the Partnership or any of their Subsidiaries is a party;
- (h) None of the Company's Subsidiaries is or has been a member of an affiliated group (within the meaning of Section 1504 of the Code or similar provisions of state, local or foreign

law) with respect to which it is liable for Taxes of another person under Treasury Regulations Section 1.1502-6 or any similar provision under state, local or foreign law;

- (i) Each of the Company's Subsidiaries that is required to be is registered for the purposes of sales tax, transfer taxes, value added taxes or any similar Tax has been so registered at all times that it has been required to be so registered, and it has complied in all material respects with all statutory requirements, orders, provisions, directions or conditions relating to such Taxes;
- (j) There are no Material Liens or encumbrances for Taxes on any of the assets of the Company, the Partnership or any of their Subsidiaries;
- (k) Each of the Company, the Partnership and each of their Subsidiaries has paid all Taxes owed or which it is required to withhold from amounts owing to employees, creditors or other third parties and has complied with all requirements (including record retention) applicable to information reporting or other reporting requirements;
- (1) None of the Company, the Partnership or any of their Subsidiaries or any predecessor to the Company, the Partnership or their Subsidiaries has made any consent under Section 341 of the Code;
- $\,$ (m) $\,$ No Tax is required to be withheld pursuant to Section 1445 of the Code as a result of the Merger or the acquisition of any Partnership Interest by Acquiror;
- (n) For U.S. federal income tax purposes, each of the Company, the Partnership and each of their Subsidiaries is, and through the Closing Date will be, treated as a partnership or disregarded entity, and none of the Company, the Partnership, any of their Subsidiaries or, to the best knowledge of the Company, any Owner or other owner of any interest in the Company or any of its Subsidiaries has, or through the Closing Date will have, made any election, taken any action or filed any Tax Return on a basis that is inconsistent with the foregoing; and
- (o) All Tax Returns (including attachments thereto) of the Company, the Partnership and each of their Subsidiaries and all closing agreements, private letter rulings, technical advice memoranda or similar agreements or rulings (if any) entered into with or issued by any taxing authority, by or with respect to the Company, the Partnership or any of their Subsidiaries that were delivered by the Company pursuant to Acquiror's due diligence request are true copies of such documents.

7.02 Covenants.

(a) Elections. Until the Effective Time, except as Previously Disclosed or as expressly contemplated by this Agreement or any Supplemental Agreement, the Company will not, and the Company will cause any of its Subsidiaries and the Partnership not to, make any Tax elections, amend any Tax elections currently in effect, change or consent to any change in any method of accounting for any Tax purpose or file any Tax Return (including any amended return) on

a basis that is not consistent with past practice, in each case without the prior written consent of the Acquiror, which consent shall not be unreasonably withheld, provided, however, that such consent shall not be considered unreasonably withheld if the election, amendment or change, as the case may be, would adversely affect the Acquiror.

Tax Returns. Any Tax Returns with respect to the operations of the Company, the Partnership and their Subsidiaries for all periods ending on or before the Closing Date (the "Pre-Closing Period"), including for those jurisdictions and tax authorities that permit or require a short period Tax Return, shall be or caused to be prepared and filed on a timely basis after the Closing by the Acquiror. Whenever it is necessary to determine the liability for Taxes of the Company, the Partnership or any of their Subsidiaries for a portion of a taxable year or period that begins before and ends after the Closing Date, the determination of the Taxes of the Company, the Partnership or any of their Subsidiaries for the portion of the taxable year or period ending on, and the portion of the year or period beginning after, the Closing Date shall be determined and shall equal: (i) in the case of any real or personal property Tax or any other Tax imposed on a periodic basis, an amount equal to the Tax for the entire taxable period beginning before and ending after the Closing Date, multiplied by a fraction the numerator of which is the number of days in such portion of the period ending on the Closing Date and the denominator of which is the number of days in such entire taxable period; and (ii) in the case of any other Tax, the amount that would be payable by the Company, the Partnership or any of their Subsidiaries if its taxable year that began prior to the Closing Date ended on the Closing Date. Any Taxes due in respect of Tax Returns filed pursuant to the first sentence of this paragraph and any Taxes allocable to Pre-Closing Periods under the immediately preceding sentence (including any interest or penalties assessed with respect to such Taxes) shall be considered "Pre-Closing Taxes". At least 10 business days prior to the filing of any such Tax Return that includes Pre-Closing Taxes, Acquiror shall make available to the SLK Representative copies of such proposed Tax Return, and the filing of such Tax Return shall be subject to the approval of the SLK Representative, which approval shall not be unreasonably withheld; provided, that, in the event the SLK Representative does not provide such approval or disapproval to Acquiror within 5 business days after receipt of such proposed Tax Return copies, the SLK Representative shall be deemed to have provided an irrevocable approval with respect to such Tax Return.

(c) Cooperation. The currently designated Tax Matters Partner (as defined in Section 6231(a)(7) of the Code) of the Company, the Partnership and each of their Subsidiaries that is treated as a partnership for U.S. federal income tax purposes, shall continue as Tax Matters Partner. If such Tax Matters Partner shall desire to resign such position (or is to be liquidated or otherwise becomes ineligible) or the Acquiror shall request the Tax Matters Partner's resignation, the current Tax Matters Partner shall, subject to the Acquiror's consent which shall not be unreasonably withheld, designate his or her replacement (and in the case of a liquidation of the Tax Matters Partner, such designation shall occur prior to liquidation) (and complete and execute any forms as may be required to evidence or effect such designation). The Acquiror and the Surviving LLC shall cooperate with the Tax Matters Partner in any Tax audit, examination or other proceeding involving the Company, the Partnership or any of their Subsidiaries that is treated as a partnership for U.S. federal income tax purposes for any taxable period that relates to periods prior to the

Closing. The Tax Matters Partner shall not enter into any compromise or agree to settle any claim pursuant to any Tax audit, examination or other proceeding which would adversely affect the Acquiror for any taxable year without the written consent of the Acquiror, which consent may not be unreasonably withheld.

- and contract, the Company, the Partnership and each of their respective Subsidiaries that is treated as a partnership for U.S. federal income tax purposes, shall make an election, or continue in effect any election previously made, under Section 754 of the Code and any comparable provision of any other tax law for the taxable year in which the Closing occurs, provided, that the Company and the Partnership shall be required to exercise any right (or, to the extent the Company or the Partnership can do so, the Company and the Partnership shall cause any of their Subsidiaries to exercise any right) they have to amend any contract to permit such election to be made.
- (e) Transfer Taxes. Fifty percent (50%) of all transfer, documentary, sales, use, stamp, registration, value added and other such Taxes and fees (including any penalties and interest) incurred in connection with the transactions contemplated by the Merger (including any real property transfer tax and any similar Tax) shall be paid by each of Acquiror on the one hand, and the Members, on the other. Each party will, at its own expense, file all necessary Tax Returns and other documentation with respect to all such Taxes and fees, and to the extent required by applicable law, each party will join, and cause its Subsidiaries and Affiliates to join, in the execution of any such Tax Returns and other documentation.
- (f) Allocation of Purchase Price. Acquiror shall determine the allocation of any adjustments to the basis of the assets of the Company, the Partnership and each of their respective Subsidiaries resulting from the Merger; provided, that any such allocation to assets covered under Section 751 of the Code, if any, shall be subject to approval of the Company, or, if after the Effective Time, the SLK Representative, which consent shall not be unreasonably withheld.
- 7.03 Termination of Tax Sharing Agreements. Any Tax allocation or sharing agreement or arrangement, whether or not written, that may have been entered into by any of the Company's Subsidiaries or any member of its affiliated groups (as defined in Section 1504(a) of the Code) shall be terminated as of the Closing Date unless a continuation of such agreement is required by law or contract, or consented to by the Acquiror. After the Closing Date, none of the Surviving LLC or any of its Subsidiaries shall have any further rights or liabilities thereunder.
 - 7.04 Seller Tax Indemnification.
- (a) Each Acquiror Party shall, pursuant to this Section 7.04(a) and the Supplemental Agreements, be entitled to indemnification from the Members, severally and not jointly (in proportion to their respective Consideration Percentages), against, and shall be held harmless (subject to the limitations set forth herein) by the Members, severally and not jointly (in proportion to their respective Consideration Percentages), from:

- (i) Pre-Closing Taxes in excess of Taxes for which reserves have been established prior to the date hereof or will be established prior to the Closing Date, consistent with past practice;
- (ii) Pre-Closing Taxes imposed on any Acquiror Party as members of an "affiliated group" (within the meaning of Section 1504(a) of the Code or any comparable provision of foreign, state or local law) that arises under Treasury Regulations Section 1.1502-6(a) or any comparable provision of foreign, state or local law; and
- (iii) a breach of the representations set forth in Section
 7.01;

provided, however, that the Members shall not be liable for and shall not indemnify any Acquiror Party for any Taxes resulting from transactions or actions taken on the Closing Date (but after the Closing) or after the Closing Date; provided, further, that notwithstanding the foregoing, no Acquiror Party shall be entitled to payment of indemnification pursuant to this Section 7.04 (a) with respect to any individual Loss unless the aggregate of all such Losses (after reduction for any tax benefits realized as a result of the payment or accrual of such Losses) shall exceed \$5 million (and then only for the amount of such excess).

- (b) The amount of any indemnification payment payable in accordance with this Section 7.04 and Article IX shall be reduced by the amount of any reduction in Taxes actually realized (or to be realized in such year) as a result of the event giving rise to the indemnification payment. The Indemnified Party shall pay the Indemnifying Party the amount of any reduction in Taxes actually realized in a future year as a result of the event giving rise to such indemnification payment within ten (10) business days of receiving such payment with respect to such reduction to the extent such payment with respect to such reduction to the extent such payment with respect to such reduction is received after the Indemnified Party receives such indemnification payment, provided, that, the amount paid shall not exceed the payments previously made by such Members under this Section 7.04 or Article IX, as the case may be. Each Member shall be severally and not jointly liable for such indemnification obligations in proportion to their respective Consideration Percentages. Unless settled in a reasonable period of time in cash by a Member or by delivery of Acquiror Common Stock (valued at market price at the time of delivery), all amounts payable for indemnification of Acquiror by any Member pursuant to this Section 7.04 may, in Acquiror's sole discretion, be offset against shares of Acquiror Common Stock owned by such Member or against other amounts payable by Acquiror or any Affiliates of Acquiror to such Member. It is understood that an Member's Acquiror Common Stock may be used to satisfy amounts payable for indemnification even if such shares are not then transferable under the transfer restrictions in the Member's Member Agreement.
 - (c) Contests.
 - (i) The Acquiror shall notify the SLK Representative if the Internal Revenue Service (or other taxing authority) shall propose an adjustment, which, if upheld, would result in the Members making a payment under Section 7.04(a). If the SLK

Representative shall request, the Acquiror shall (x) consult with the SLK Representative regarding the contest of such proposed adjustment, (y) use good faith efforts to permit the SLK Representative to attend scheduled meetings with the Internal Revenue Service (or other taxing authority) regarding such contest and (z) if feasible, provide the SLK Representative with copies of written materials prior to submission. Notwithstanding the foregoing, the Acquiror shall have absolute and complete control of any contest provided, that the Acquiror shall not settle any contest without the consent of the SLK Representative, which consent shall not be unreasonably withheld.

- (ii) The SLK Representative shall control any contest involving proposed adjustments to Tax returns filed by the Company, the Partnership or any of their respective Subsidiaries for which the liability for additional Taxes, if the adjustments were to be upheld, would be borne directly by the Members. The SLK Representative shall consult in good faith with the Acquiror during the course of such contest, provided, that without the consent of the Acquiror, which consent shall not be unreasonably withheld, the SLK Representative shall not consent to any adjustment that would adversely affect Acquiror after the Closing.
- (iii) To the extent consistent with the foregoing, the Company and the SLK Representative shall cooperate with each other in pursuing any contest under this section.
- 7.05 Exclusivity. The indemnification and other provisions of this Article VII shall be the sole provisions governing tax indemnification matters
- 7.06 Survival of Obligations. The obligations of the parties set forth in this Article VII shall remain in effect until 60 days after the expiration of the applicable statute of limitations.
- 7.07 Agreed Tax Treatment. Any Payment made to an Acquiror Party pursuant to Section 7.04 and any Payment pursuant to Article IX shall constitute an adjustment of the Merger Consideration for tax purposes, and shall be treated as such by the Acquiror Parties, the Company and any Member on their respective Tax Returns to the extent permitted by law. The parties intend that the Merger Consideration shall, for U.S. federal income tax purposes, be treated as a payment for the Members' Company Membership Interest.

ARTICLE VIII

CONDITIONS TO CONSUMMATION OF THE MERGER

8.01 Conditions to Each Party's Obligation to Effect the Merger. The respective obligation of each of Acquiror, NewCo and the Company to consummate the Merger is subject to the

fulfillment or written waiver by the other parties prior to the Closing of each of the following conditions:

- (a) Governmental and Regulatory Consents. The waiting period applicable to the consummation of the Merger under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, shall have expired or been terminated. All approvals and authorizations of, filings and registrations with, and notifications to, all Governmental Authorities and Self-Regulatory Organizations required for the consummation of the Merger and the other transactions contemplated by this Agreement and for the prevention of any termination of any material right, privilege, license or franchise of either (1) Acquiror or its Subsidiaries or (2) the Company or its Subsidiaries (and, in each case, granted by any Governmental Authority or Self-Regulatory Organization) shall have been obtained or made and shall be in full force and effect and all waiting periods required by law or by rule or regulation of any Governmental Authority shall have expired. Notwithstanding the foregoing, none of the preceding in the previous two sentences shall be deemed obtained or made if it shall be subject to any condition or restriction the effect of which is that such condition or restriction could reasonably be expected to have a Material Adverse Effect on the Company or Acquiror.
- (b) No Injunction. No Governmental Authority of competent jurisdiction shall have, after the date of this Agreement, enacted, issued, promulgated, enforced or entered any statute, rule, regulation, judgment, decree, injunction or other order (whether temporary, preliminary or permanent) which is in effect and prohibits or materially restricts or materially adversely affects the consummation of the transactions contemplated by this Agreement (an "Order").
- 8.02 Conditions to Obligations of the Company. The obligations of the Company to consummate the Merger are also subject to the fulfillment or written waiver by the Company prior to the Closing of each of the following conditions:
 - (a) Representations and Warranties. The representations and warranties of Acquiror set forth in this Agreement shall be true and correct (after giving effect to Section 5.02) as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date, except that representations and warranties that by their terms speak only as of the date of this Agreement or some other date need be true and correct only as of such date; and the Company shall have received a certificate, dated the Closing Date, signed on behalf of Acquiror by a senior executive officer to such effect.
 - (b) Performance of Obligations of Acquiror. Acquiror and NewCo shall have performed and complied with in all material respects all agreements, covenants and obligations required to be performed by them under this Agreement at or prior to the Closing Date, and the Company shall have received a certificate, dated the Closing Date, signed on behalf of Acquiror by a senior executive officer to such effect.

- 8.03 Conditions to Obligations of Acquiror and NewCo. The obligations of each of Acquiror and NewCo to consummate the Merger are also subject to the fulfillment or written waiver by Acquiror prior to the Closing of each of the following conditions:
 - (a) Representations and Warranties. The representations and warranties of the Company set forth in this Agreement (after giving effect to Section 5.02 in the case of the representations and warranties of the Company) and of the Members in the Member Agreements (other than violations of the representations and warranties of the Members that will not adversely affect Acquiror's ownership of, or rights with respect to, any Partnership Interest or Membership Interest) shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date, except that representations and warranties that by their terms speak only as of the date of this Agreement or some other date need be true and correct only as of such date; and Acquiror shall have received a certificate, dated the Closing Date, signed on behalf of the Company by a senior executive officer of the Company to such effect.
 - (b) Performance of Obligations of the Company. The Company shall have performed and complied in all material respects with all agreements, covenants and obligations required to be performed by it under this Agreement at or prior to the Closing Date, and Acquiror shall have received a certificate, dated the Closing Date, signed on behalf of the Company by a senior executive officer of the Company to such effect.
 - (c) Third Party Consents. All consents or approvals of all persons, other than Governmental Authorities and those Previously Disclosed in Items 1 through 4 in Section 5.03(e) of the Disclosure Schedule, (i) required for or in connection with the execution, delivery and performance of this Agreement and the consummation of the Merger and the other transactions contemplated by this Agreement shall have been obtained and shall be in full force and effect, unless the failure to obtain any such consent or approval is not reasonably likely to have, individually or in the aggregate, a Material Adverse Effect on Acquiror or the Surviving LLC and (ii) required so that Acquiror or any of its Affiliates are not restricted from engaging in any material business or activity in, or relating to, the financial services industry.
 - (d) Member Agreement. A Member Agreement and Supplemental Member Agreement for each person who is a Member as of the date hereof or is a Member as of the Closing Date shall have been executed and delivered.
 - (e) Acquiror Shareholders Agreement. Each Member of the Company shall have executed and delivered to the Acquiror a counterpart of the Acquiror Shareholders Agreement.
 - (f) Agreements with Exchanges. Acquiror shall have established procedures that shall have been approved by the NYSE pursuant to NYSE Rule 98, and procedures that shall have been approved by the American Stock Exchange pursuant to its Rule 193 (as well as

procedures approved by the Philadelphia Stock Exchange, the Chicago Board Options Exchange and the Pacific Stock Exchange pursuant to analogous rules), each in a form, and with terms and conditions, reasonably satisfactory to Acquiror.

- (g) SLK Investing Co. Concurrently with the Closing, Acquiror shall have acquired, either directly or indirectly, all interests with respect to, and be the sole record and beneficial owner of, the outstanding Class A limited partnership interests of the Partnership for an amount not to exceed \$25,000,000.
- (h) Pledge Agreement. Each Member of the Company shall have executed and delivered to the Acquiror a counterpart of the Pledge Agreement (as defined in the Member Agreement).
- (i) Company Condition. As of the Effective Time, the Company shall have no assets, liabilities (whether accrued, contingent, absolute, determined, determinable or otherwise) or other obligations other than its Partnership Interest and shall not have transferred to the Partnership or any other Subsidiary any liability (other than actions Previously Disclosed on Section 5.03(g) of the Disclosure Schedule).
- (j) Manager. At the Effective Time, SLK Management Inc. shall have resigned as manager of the Company in accordance with Section 415 of the New York Limited Liability Company Law.

ARTICLE IX

INDEMNIFICATION

9.01 Indemnification.

(a) After the Effective Time, each Acquiror Party shall, pursuant to this Section 9.01(a) and the Supplemental Agreements, be entitled to indemnification from the Members against, and shall be held harmless (subject to the limitations set forth herein) by the Members from, any and all damage, loss, liability and expense (including without limitation reasonable expenses of investigation and attorneys' fees and expenses in connection with any action, suit or proceeding) ("Losses") incurred or suffered by such person arising out of, related to, or in connection with (i) any breaches of the Company's representations and warranties (other than breaches of those representations and warranties set forth in Article VII for which indemnification is provided in Section 7.04 and other than for which indemnification is provided in Section 9.01(b)) set forth in this Agreement (determined without regard to Section 5.02), (ii) the failure by the Company to perform in any material respect any of its covenants or agreements contained in this Agreement, (iii) any claim of any current or former Partner or Partners (other than a current or former holder of a Class C limited partnership interest) or other person alleging rights equivalent to those of a Partner against Acquiror (excluding any unrelated business or other transaction), the Company (or the Surviving LLC as successor to the Company) or any of its Affiliates for any event, occurrence, act or omission

occurring at or prior to the Effective Time, (iv) any claim of any current or former Member or Members (or other person alleging rights equivalent to those of a Member) against Acquiror (excluding any unrelated business or other transaction), the Company (or the Surviving LLC as successor to the Company) or any of its Affiliates for any event, occurrence, act or omission occurring at or prior to the Effective Time (other than any claim of a current Member contemplated by clause (v)) and (v) any claim of a current Member or Members in respect of any issue with the Company or another Member arising out of Article III hereof; provided, however, that notwithstanding the foregoing each Member shall be severally and not jointly liable for such indemnification obligations in proportion to their respective Consideration Percentages; and provided, further, that notice of any claim under this Section 9.01(a) in respect of or relating to a breach of the representations and warranties or the Company shall have been received by the Members on or prior to the date until which the relevant representation or warranty shall survive pursuant to Section 9.03; and provided, further, that notwithstanding the foregoing, no Acquiror Party shall be entitled to payment of indemnification pursuant to this Section 9.01(a)(i), (iii) or (iv) with respect to any individual item of Loss, (A) unless, with respect to this Section 9.01(a)(i), such item exceeds \$1,000,000 (after reduction for any tax benefits realized by any Acquiror Party as a result of such Loss) and, with respect to this Section 9.01(a)(iii) and (iv), such item exceeds \$100,000 (after reduction for any tax benefits realized by any Acquiror Party as a result of such Loss) and (B) unless, with respect to this Section 9.01(a)(i), the aggregate of all such Losses of all Acquiror Parties shall exceed \$10 million (after reduction for any tax benefits realized by any Acquiror Party as a result of such Loss and then only for the amount of such excess) (it being understood that this proviso shall not apply to any Loss arising out of, related to or in connection with, any breach of the last sentence of Section 5.03(g)(1)); and provided, further, that no Acquiror Party shall be entitled to payment of indemnification pursuant to the provisions of this Section 9.01(a)(i), (iii) or (iv) to the extent that the amount of all Losses against which the Acquiror Parties shall have been previously indemnified under such provisions of this Section 9.01(a)(i), (iii) or (iv) exceeds \$300 million in the aggregate (it being understood that this proviso shall not apply to any Loss arising out of, related to or in connection with, any breach of the last sentence of Section 5.03(g)(1)). Unless settled in a reasonable period of time in cash by a Member or by delivery of Acquiror Common Stock (valued at market price at the time of delivery), all amounts payable for indemnification of Acquiror by any Member pursuant to this Section 9.01(a) may, in Acquiror's sole discretion, be offset against shares of Acquiror Common Stock owned by such Member or against other amounts payable by Acquiror or any Affiliates of Acquiror to such Member and held pursuant to the terms of the Custody Agreement. It is understood that a Member's Acquiror Common Stock may be used to satisfy amounts payable for indemnification even if such shares are not then transferable under the transfer restrictions in the Member's Member Agreement.

(b) After the Effective Time, each Acquiror Party shall, pursuant to this Section 9.01(b) and the Supplemental Agreements, be entitled to indemnification from the Members against, and shall be held harmless by the Members from, any and all Losses incurred or suffered by such person arising out of, related to or in connection with the pending or threatened Litigation involving the Company or any of its Affiliates as of the date hereof and Previously Disclosed; provided, however, that (i) the first proviso contained in Section 9.01(a) and the last two sentences contained in Section 9.01(a) shall apply mutatis mutandis to the indemnification provided in this Section 9.01(b),

- (ii) no Acquiror Party shall be entitled to payment of indemnification pursuant to this Section 9.01(b) unless the aggregate of all such Losses of all Acquiror Parties (after reduction for any benefits realized by any Acquiror Party as a result of such Loss) shall exceed \$4 million (and then only for the amount of such excess) and (iii) no Acquiror Party shall be entitled to payment of indemnification pursuant to the provisions of this Section 9.01(b) to the extent that the amount of all Losses against which the Acquiror Parties shall have been previously indemnified under this Section 9.01(b) exceed \$29 million in the aggregate.
- (c) After the Effective Time, the Members shall, pursuant to this Section 9.01(c), be entitled to indemnification from Acquiror against, and shall be held harmless (subject to the limitations set forth herein) by Acquiror from, any and all Losses incurred or suffered by them arising out of, related to, or in connection with, any breaches of, (i) Acquiror's representations and warranties set forth in this Agreement (determined without regard to Section 5.02); or (ii) the failure by Acquiror to perform in any material respect any of its covenants or agreements contained in this Agreement; provided, that notice of any claim under this Section 9.01(c)(i) shall have been received by Acquiror on or prior to the date until which the relevant representation or warranty shall survive pursuant to Section 9.03; and provided, further, that notwithstanding the foregoing, no Member shall be entitled to payment of indemnification pursuant to this Section 9.01(c)(i) with respect to any individual item of Loss, (A) unless, with respect to this Section 9.01(c)(i), such item exceeds \$1,000,000 (after reduction for any tax benefits realized by any Member as a result of such Loss) and (B) unless the aggregate of all such Losses of all such parties with respect to this Section 9.01(c)(i) shall exceed \$10 million (after reduction for any tax benefits realized by any Member as a result of such Loss and then only for the amount of such excess); and provided, further, that notwithstanding the foregoing, no Member will be entitled to payment of indemnification pursuant to this Section 9.01(c)(i) to the extent that the amount of all Losses for which Acquiror has previously provided indemnification under Section 9.01(c)(i) exceeds \$300 million in the aggregate.
- (d) For the purposes of this Section 9.01, in computing such individual or aggregate amounts of claims, the amount of each claim shall be deemed to be an amount net of any insurance proceeds and any indemnity, contribution or other similar payment recoverable by the indemnified party from any third party with respect thereto.
 - 9.02 Notice and Defense of Claims.
- (a) Each party entitled to indemnification under this Article IX (the "Indemnified Party") shall give notice to the party or parties required to provide indemnification (the "Indemnifying Party") promptly, but not later than 45 days, after such Indemnified Party receives written notice of any claim, event or matter as to which indemnity may be sought; provided, that the failure of the Indemnified Party to give notice as provided in this Section 9.02 shall not relieve any Indemnifying Party of its obligations under Section 9.01 or Section 7.04, except to the extent that such failure actually and materially prejudices the rights of any such Indemnifying Party and then only to the extent of such prejudice. In the event of any claim, action, suit, proceeding or demand asserted by any person who is not a party (or a successor to a party) to this Agreement (a "Third-Party Claim") which is or gives rise to an indemnification claim, the Indemnifying Party

may elect to assume and control the defense of any such claim and any litigation resulting therefrom, provided, that counsel for the Indemnifying Party, who shall conduct the defense of such claim or any litigation resulting therefrom, shall be approved by the Indemnified Party (whose approval shall not unreasonably be withheld or delayed), and the Indemnified Party may participate in such defense at the Indemnified Party's expense, which shall include counsel of its choice, provided, further, that the Indemnified Party shall have the right to employ, at the Indemnifying Party's expense, one counsel of its choice in each applicable jurisdiction (if more than one jurisdiction is involved) to represent the Indemnified Party if, in the Indemnified Party's reasonable judgment, there exists an actual or potential conflict of interest between the Indemnified Party and the Indemnifying Party or if the Indemnifying Party elects not to defend, compromise or settle a Third-Party Claim, provided, further, that if the Indemnifying Party elects to assume and control the defense but fails to retain counsel to prosecute the action within thirty days of such election, then the Indemnified Party shall have the right to defend such Third-Party Claim on behalf of and for the account and risk of the Indemnifying Party. The Indemnifying Party, in the defense of any such claim or litigation, shall not, except with the prior written consent of the Indemnified Party, consent to entry of any judgment or entry into any settlement (i) which does not include as an unconditional term thereof the giving by the claimant or plaintiff to the Indemnified Party of a release from all liability in respect to such claim or litigation or (ii) which could reasonably be expected to restrict materially the conduct of business of Acquiror or any of its Subsidiaries. In the event the Indemnifying Party is prepared to settle an action and the Indemnified Party reasonably believes that such settlement could have such an impact on it then the Indemnified Party may choose to continue to defend such action in which case the Indemnified Party shall be responsible for the incremental costs of continuing such matter (including, the additional legal costs and additional costs or losses with respect to such matter) in excess of the amount offered to be settled by the Indemnifying Party (and the legal costs incurred to date) which costs shall be borne by the Indemnifying Party. The Indemnified Party shall not settle or compromise any such claim without the prior written consent of the Indemnifying Party, which consent shall not be unreasonably withheld. The Indemnified Party shall make its employees available and furnish such information regarding itself or the claim in question as the Indemnifying Party may reasonably request in writing and as shall be reasonably required in connection with the defense of such claim and litigation resulting therefrom.

9.03 Survival of Representations and Warranties. The representations and warranties shall survive the Closing, until February 28, 2002 and, thereafter, to the extent a claim is made prior to such date with respect to any breach of any such representation or warranty until such claim is finally determined or settled, provided, that the representations and warranties contained in Sections 5.03(o) and Article VII shall survive the Closing until the expiration of all applicable statutes of limitation. Notwithstanding any other provision of this Agreement, the last sentence of the representation and warranty set forth in Section 5.03(g)(i) and the indemnity obligation contained in Section 9.01(b) shall survive forever.

ARTICLE X

TERMINATION

- $10.01\,$ Termination. This Agreement may be terminated, and the Merger may be abandoned:
 - (a) Consent. At any time prior to the Effective Time, by the mutual consent of each of Acquiror and the Company.
 - (b) Breach. At any time prior to the Effective Time, by Acquiror on the one hand, or the Company, on the other hand, in the event of either: (1) a breach by the other of any representation or warranty contained herein, which breach cannot be or has not been cured within 30 days after the giving of written notice to the breaching party of such breach; or (2) a breach by the other of any of the covenants or agreements contained herein, which breach cannot be or has not been cured within 10 days after the giving of written notice to the breaching party of such breach and which breach in the case of either of (1) and (2) above, would cause any condition set forth in Article VIII not to be satisfied.
 - (c) Delay. At any time prior to the Effective Time, by Acquiror, on the one hand, or the Company on the other hand, in the event that the Effective Time shall have failed to occur on or before May 31, 2001 except to the extent that such failure arises out of or results from any breach of this Agreement by or knowing action or inaction of the party seeking to terminate pursuant to this Section 10.01(c).
 - (d) No Approval. By Acquiror, on the one hand, or the Company, on the other hand, if the approval of any Governmental Authority required for consummation of the Merger and the other transactions contemplated by this Agreement shall have been denied by final nonappealable action of such Governmental Authority, or such Governmental Authority shall have requested the permanent withdrawal of any application therefor, or any such approval shall be made subject to any condition or restriction described in the proviso to Section 8.01(b).
 - (e) Order. At any time prior to the Effective Time, by Acquiror, on the one hand, or the Company on the other hand, if any Order permanently restraining, enjoining or otherwise prohibiting consummation of the Merger shall become final and non-appealable or any Law is in effect or is adopted or issued, which has the effect of prohibiting the Merger.
- 10.02 Effect of Termination and Abandonment. In the event of termination of this Agreement and the abandonment of the Merger pursuant to this Article X, no party to this Agreement shall have any liability or further obligation to any other party hereunder except (a) as set forth in Section 11.03 and (b) that termination will not relieve a breaching party from liability for any willful breach of this Agreement.

ARTICLE XI

MISCELLANEOUS

This Agreement and the Supplemental Agreements represent the entire understanding of the parties hereto with reference to the transactions contemplated hereby and this Agreement and the Supplemental Agreements supersede any and all other oral or written agreements heretofore made other than the Confidentiality Agreement. Except for Section 3.05 (with respect to the Members or employees of the Company as it relates to Acquiror's obligation to award the Retention RSUs and SLK RSUs only), for Section 3.03 (with respect to Withdrawn Members and holders of the Managing Director Subordinated Notes), for Section 6.12 (with respect to Members), for Article VII (with respect to Members and the SLK Representative), and for Article IX, in each case only, on and after the Effective Time, nothing in this Agreement, expressed or implied, is intended to confer upon any person, other than the parties hereto or their respective successors and permitted assigns, any rights, remedies, obligations or liabilities under or by reason of this Agreement.

11.02 Waiver; Amendment. Prior to the Effective Time, any provision of this Agreement may be (a) waived by the party benefitted by the provision or (b) amended or modified at any time, by an agreement in writing between the parties hereto approved or authorized by their respective Boards of Directors or similar governing body and executed in the same manner as this Agreement.

11.03 Expenses. Each party hereto will bear all expenses incurred by it in connection with this Agreement and the transactions contemplated hereby.

11.04 Notices. All notices, requests and other communications hereunder to a party shall be in writing and shall be deemed given (a) on the date of delivery, if personally delivered or telecopied (with confirmation), (b) on the first business day following the date of dispatch, if delivered by a recognized next-day courier service, or (c) on the third business day following the date of mailing, if mailed by registered or certified mail (return receipt requested), in each case to such party at its address or telecopy number set forth below or such other address or numbers as such party may specify by notice to the parties hereto.

If to the Company to:

SLK LLC 120 Broadway

New York, New York 10271 Facsimile: (212) 433-7294

Attention: Carl H. Hewitt

General Counsel and Managing Director

With a copy to:

Paul, Weiss, Rifkind, Wharton & Garrison 1285 Avenue of the Americas New York, New York 10019 Facsimile: (212) 757-3990 Attention: Robert B. Schumer, Esq.

If to Acquiror or NewCo to:

The Goldman Sachs Group, Inc. One New York Plaza New York, New York 10004 Facsimile: (212) 902-3000

Attention: Gregory Palm

General Counsel and Managing Director

With a copy to:

Sullivan & Cromwell 125 Broad Street New York, New York 10004 Facsimile: (212) 558-3588 Attention: James C. Morphy, Esq.

11.05 Additional Provisions. (a) The parties recognize that certain Members whose aggregate percentage allocation of the profits and losses of the Company do not exceed 5% may wish to remain members of the Company following the Effective Time. The parties agree to negotiate in good faith such changes to this Agreement, including to Sections 2.01(b), 2.01(c) and 3.01(a), with a view to permitting such Members to remain as members and, if the parties reach Agreement thereon, in each party's absolute discretion, they will set forth in writing such terms as may be agreed including (i) that such members' interests in the Company after the Effective Time shall have no allocation of profit and loss or any voting rights, by law or otherwise, (ii) that such members' interests shall be entitled to distributions limited to a fixed interest rate applicable to their capital account for so long as such persons remain members and (iii) that none of Acquiror or any of its Affiliates (including the managing member of the Company) shall have any duty (including fiduciary duty) or obligation to such members not required by law.

- (b) Any Member who withdraws from the Company after the date hereof and prior to the Closing shall not be deemed a "Withdrawn Member" for purposes of this Agreement and shall continue to be deemed a Member for purposes of this Agreement.
- 11.06 Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed to constitute an original.

Governing Law; Enforcement; Waiver of Jury Trial. (a) 11.07 This Agreement shall be governed by, and interpreted in accordance with, the laws of the State of New York applicable to contracts made and to be performed entirely within such State. The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement in any court of the State of New York or the Federal courts of the United States of America located in the State of New York, in each case in the borough of Manhattan (the "Chosen Courts"). The parties hereby irrevocably submit to the exclusive jurisdiction of the Chosen Courts in respect of the interpretation and enforcement of the provisions of this Agreement and of the documents referred to in this Agreement, and in respect of the transactions contemplated hereby, and hereby waive, and agree not to assert, as a defense in any action, suit or proceeding for the interpretation or enforcement hereof or of any such document, that it is not subject thereto or that such action, suit or proceeding may not be brought or is not maintainable in the Chosen Courts or that the venue thereof may not be appropriate or that this Agreement or any such document may not be enforced in or by such courts, and the parties hereto irrevocably agree that all claims with respect to such action or proceeding shall be heard and determined in the Chosen Courts. The parties hereby consent to and grant the Chosen Courts exclusive jurisdiction over the person of such parties and over the subject matter of such dispute and agree that mailing of process or other papers in connection with any such action or proceeding in the manner provided in Section 11.04 or in such other manner as may be permitted by law shall be valid and sufficient service thereof.

(b) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (i) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (ii) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (iii) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (iv) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 11.07.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed in counterparts by their duly authorized officers, all as of the day and year first above written.

THE GOLDMAN SACHS GROUP, INC.

By: /s/ HENRY M. PAULSON, JR.

Name: Henry M. Paulson, Jr. Title: Chairman and Chief Executive Officer

SLK LLC By SLK Management Inc., its Managing Member

By: /s/ GARY GOLDRING

Name: Gary Goldring Title: Co-Chief Executive Officer

Form of Member Agreement

MEMBER AGREEMENT, dated as of September 10, 2000 (this "Agreement"), between The Goldman Sachs Group, Inc., a Delaware corporation ("Acquiror"), the signatory hereto, who is the owner of that membership interest ("Membership Interest") of SLK LLC, a New York limited liability company (the "Company") set forth next to such signatory's name in Exhibit A (the "Member") and, for purposes of Section 1(b) hereof only, SLK LLC.

RECITALS

- A. Subject to the terms and conditions contained in the Agreement and Plan of Merger, dated as of September 10, 2000, by and between Acquiror and the Company (the "Merger Agreement"), the Company and Acquiror intend to effect the merger (the "Merger") of NewCo with and into the Company. Capitalized terms used but not defined herein shall have the meaning set forth in the Merger Agreement.
- B. Upon the consummation of the Merger, the Merger Agreement provides for the conversion of the Member's Membership Interests into such amount of cash and shares of common stock, par value \$0.01 per share ("Acquiror Common Stock"), of Acquiror (the "Shares"), as provided in the Merger Agreement. Member will derive substantial value from Acquiror's execution, delivery and performance of the Merger Agreement.
- C. As an inducement to, and a condition of, Acquiror's willingness to enter into the Merger Agreement, and having reviewed the Merger Agreement and the terms of the proposed Merger, Member is executing this Agreement for the benefit of Acquiror, the Firm (as hereinafter defined), and each Acquiror Party.
- NOW, THEREFORE, in consideration of the premises and the covenants and agreements contained in the Merger Agreement, and intending to be legally bound hereby, Member agrees as follows:
- 1. Adoption of Merger Agreement; Irrevocable Proxy; Termination. (a) Member hereby (i) adopts and approves the resolutions attached hereto as Exhibit B and adopts and consents to the Merger Agreement and the transactions contemplated thereby including, without limitation, the Merger, in accordance with the provisions of Section 407 of the New York Limited Liability Company Law, (ii) upon request by Acquiror, agrees to vote to adopt and approve such matters described in clause (i) of this Section 1(a), (iii) agrees to vote against, and to withhold consent from, any action or proposal that would compete with or would serve to materially interfere with, delay, discourage, adversely affect or inhibit

the timely consummation of the transactions contemplated by the Merger Agreement, including, without limitation, the Merger, (iv) consents to the replacement of SLK Management Inc. by Acquiror or an Affiliate designated thereby, as the managing member of the Company, such replacement to be effective as of the Effective Time, and (v) elects to cause the payment and distribution by the Company of the Shares (as such term is defined in the Operating Agreement) of the Withdrawn Members (as such term is defined in the Operating Agreement) to such Withdrawn Members prior to the close of any applicable Additional Holding Periods (as such term is defined in the Operating Agreement) in the manner set forth in Section 3.04 of the Merger Agreement. To the extent necessary and as permitted by applicable law, Member hereby also irrevocably (i) waives any notice, or requirement thereof, with respect to any meeting of Members or other proceeding for the purpose of adopting and approving the Merger Agreement, the Merger or any related matters and (ii) acknowledges that Member will not have any dissenters' or similar rights in connection with the Merger and the consummation of the other transactions contemplated by the Merger Agreement. If the Merger Agreement is properly terminated for any reason in accordance with its terms and the Merger is not consummated, this Agreement shall terminate concurrently with such termination of the Merger Agreement.

- (b) The Company hereby acknowledges receipt and delivery at its principal place of business of an executed copy of this Agreement sufficient to comply with Section 407(b) of the New York Limited Liability Company Law.
- 2. Cooperation and Support; HSR; No Transfer of Membership Interest.
- (a) Member will timely execute and deliver all Supplemental Agreements, if any, to which it is to be a party as provided in the Merger Agreement.
- (b) Each Member who will be considered an "acquiring person" under the rules and regulations promulgated under the Hart-Scott-Rodino Antitrust Improvements Act of 1976 (the "HSR Act") in connection with the consummation of the transactions contemplated by the Merger Agreement agrees to use its reasonable best efforts to file a Notification and Report Form under the HSR Act with respect to the transactions contemplated by the Merger Agreement as soon as practicable after the date hereof.
- (c) From and after the date hereof and until the earlier of the Effective Time or the termination of this Agreement in accordance with its terms, Member shall not Transfer (as hereinafter defined), directly or indirectly, all or any portion of its Membership Interest without Acquiror's prior written consent; provided, however, that in the event of such Member's death during the term of this Agreement, Member's Membership Interest may be transferred in accordance with the Operating Agreement.
- (d) Member hereby agrees that from and after the date hereof until the earlier of the Effective Time or the termination of this Agreement in accordance with its terms, Member will not, in any manner, directly or indirectly (including through advisors, agents or other intermediaries), take any action to seek, encourage, support or discuss any offer from

any corporation, partnership, person or other entity or group (other than Acquiror) to acquire any direct or indirect Company Membership Interests, Partnership Interests or equity securities of any Subsidiary of the Company (other than the issuance by the Partnership of Class C Limited Partnership Interests and the issuance by First Options of Chicago Inc. of shares of Class A Preferred Stock in connection with joint back office arrangements in the ordinary course of business and consistent with past practices), to merge the Company or any Subsidiary or Affiliate of the Company with any such person, or to otherwise acquire any significant portion of the assets of the Company or any Subsidiary of the Company.

- (e) Member hereby agrees to execute and deliver to Acquiror at the Closing, a counterpart of the Acquiror Shareholders Agreement, which agreement shall be in full force and effect.
- (f) At any time after the date hereof, Member shall promptly execute, acknowledge and deliver any other assurances or documents reasonably requested by Acquiror and necessary for Member to satisfy its obligations hereunder.
- (g) If, at the Closing, Member shall have an interest, direct or indirect, in SLK Investing Co. ("SLK"), Member shall, acting together with all other Members holding such an interest in SLK, cause SLK to convey to Acquiror or its designee any and all of SLK's current right, title and interest, whether direct or indirect, in the Class A Limited Partnership Interest of the Partnership for an amount that shall be determined by the Company, provided, that in no event shall Acquiror pay more than \$25 million in the aggregate for 100% of all of the Class A Partnership Interest of the Partnership. After the date hereof, Member shall not Transfer its interest in SLK and, except as otherwise provided by the preceding sentence, Member shall, acting together with all other Members holding such an interest in SLK, take all actions necessary to cause SLK not to transfer, sell or otherwise reduce its right, title or interest in the Class A Limited Partnership Interest of the Partnership.
- 3. Representations, Warranties and Agreements. Member represents and warrants to, and agrees with, Acquiror as follows:
- (a) Member has all requisite power and authority to execute and deliver this Agreement and to perform all of the obligations imposed upon Member hereunder. Member is the lawful record and beneficial owner of Member's Membership Interest set forth next to Member's name in Exhibit A hereto, free and clear of all Liens other than Liens created by the Constitutive Documents; no other person (other than any wholly-owned affiliate controlled by Member) has an interest, legal, beneficial or otherwise, in Member's Membership Interest and no consent of any other person is required for the execution and delivery by Member of, and performance by Member of its obligations under, this Agreement and the Merger Agreement. Without limiting the foregoing, no person has any rights with respect to Member's Membership Interest or the cash or Shares to be issued to Member pursuant to the Merger Agreement under any community property or similar legal provision

or concept. The representations and warranties contained in the second and third sentences of this Section 3(a) shall not apply to any Membership Interest to the extent that such representations and warranties would be deemed breached by virtue of any existing pledge of such Membership Interest or by an existing contractual restriction or because spousal consent is required as of the date hereof (so long as such representations and warranties shall be true and correct, and any required spousal consent under any community property law shall have been obtained, at the Closing); provided, however, that this sentence shall not affect any representation and warranty relating to the right, power or authority of Member to vote or give a proxy or consent with respect to Member's Membership Interest. There are no actions, suits or proceedings pending or, to Member's knowledge, threatened against or affecting Member or the assets of Member in any court or before or by any federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality which is reasonably likely to materially impair, restrict or delay the ability of Member to perform Member's obligations under this Agreement and the Merger Agreement or would make this paragraph untrue in any material respect. This Agreement constitutes the valid and legally binding agreement of Member, enforceable in accordance with its terms except as the enforceability hereof may be limited by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other similar laws relating to the enforcement of creditors' rights generally and by general principles of equity.

(b) The execution, delivery and performance of this Agreement do not and will not (1) constitute a breach or violation of, or a default under, or cause or allow the acceleration or creation of a Lien (with or without the giving of notice, passage of time or both) pursuant to any law, rule or regulation or any judgment, decree, order, governmental or non-governmental permit or license, or any Contract to which Member is a party or to which Member or any of Member's assets are subject or bound or (2) require any consent or approval of any other person under any such law, rule, regulation, judgment, decree, order, governmental or non-governmental permit, license or Contract, in each case, which would materially impair, restrict or delay the ability of Member to perform Member's obligations under this Agreement.

(c) Member will acquire the Shares for Member's own account and not with a view to, or for resale in connection with, the distribution thereof and Member has no present intention of selling, Transferring, granting any participation in, or otherwise distributing the Shares except in conformity with the Securities Act and other applicable federal and state securities laws (the "Securities Laws").

(d) Member has read and fully understands this Agreement and the Merger Agreement and the terms of the proposed Merger. The SEC Documents related to Acquiror have been made available to Member, and Member understands and has evaluated the risks of an investment in the Shares. Member has been given the opportunity to ask questions of, and receive answers from, Acquiror and its representatives concerning the matters pertaining to Member's investment in the Shares and has been given the opportunity to review such

additional information as was necessary to evaluate the merits and risks of an investment in the Shares. Member can bear the economic risk of an investment in the Shares.

- (e) Member is an "accredited investor" as defined in Regulation D, which has been adopted by the Securities and Exchange Commission (the "SEC") under the Securities Act .
- (f) Member understands that all the Shares will be characterized as "restricted securities" under the Securities Laws inasmuch as they are being acquired from Acquiror in a transaction not involving a public offering and that, consequently, the Shares may not be resold without first being registered under the Securities Laws except in certain limited circumstances. Specifically, Member is familiar with SEC Rules 144 and 145 and understands, and agrees to comply with, the resale limitations imposed thereby, by the legends described in paragraph (g) below and by the Securities Laws generally or cause any other person who has an interest in such Shares to so comply.
- (g) Member understands and agrees that the certificates issued to Member representing the Shares will bear the following legends and such other legends as Acquiror may reasonably deem necessary or desirable:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO A MEMBER'S AGREEMENT WITH THE GOLDMAN SACHS GROUP, INC. (THE "MEMBER'S AGREEMENT"), AND A SHAREHOLDERS' AGREEMENT AMONG THE GOLDMAN SACHS GROUP, INC. AND THE PERSONS NAMED THEREIN, COPIES OF WHICH ARE ON FILE AT THE PRINCIPAL EXECUTIVE OFFICE OF THE GOLDMAN SACHS GROUP, INC., AND WHICH, AMONG OTHER MATTERS, PLACE RESTRICTIONS ON THE VOTING AND DISPOSITION OF SUCH SECURITIES. THE SECURITIES REPRESENTED BY THIS CERTIFICATE MAY BE INDIRECTLY OR DIRECTLY SOLD, TRANSFERRED, PLEDGED OR OTHERWISE DISPOSED OF, INCLUDING ANY DISPOSITION OF THE ECONOMIC OR OTHER RISKS OF OWNERSHIP THROUGH HEDGING TRANSACTIONS OR DERIVATIVES INVOLVING SUCH SECURITIES, ONLY IN ACCORDANCE WITH THE PROVISIONS OF THE MEMBER'S AGREEMENT."

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE U.S. SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR OTHER SECURITIES LAWS AND MAY NOT BE OFFERED, SOLD, EXCHANGED, TRANSFERRED, ASSIGNED, PLEDGED, PARTICIPATED, HYPOTHECATED OR OTHERWISE DISPOSED OF (EACH A "TRANSFER") EXCEPT PURSUANT TO AN EXEMPTION

FROM REGISTRATION UNDER THE SECURITIES ACT AND, IF APPLICABLE, SUCH OTHER SECURITIES LAWS AND FOLLOWING RECEIPT BY THE GOLDMAN SACHS GROUP, INC. OF A LEGAL OPINION IN FORM AND SUBSTANCE SATISFACTORY TO IT THAT SUCH TRANSFER IS PERMITTED."

The foregoing legends will be removed from a Share certificate at the request of Member or another holder thereof in connection with the proposed transfer thereof only upon satisfaction of Acquiror that such legend is no longer required or appropriate, including, in the case of the Securities Laws legend, receipt by Acquiror of an opinion of counsel, in form and substance satisfactory to Acquiror, to the effect that registration under the Securities Act is unnecessary in respect of such proposed transfer, in reliance upon SEC Rule 144 or 145 under the Securities Act, and that such legend is not required by law to appear on such certificate.

Member agrees and consents to the entry of stop transfer orders against the transfer of Shares subject to transfer restrictions.

- (h) Member meets any suitability standards imposed by the state of Member's residence or imposed by any other applicable laws.
- (i) From and after the date hereof and until the earlier of the Effective Time or the termination of this Agreement in accordance with its terms, Member agrees to comply as soon as reasonably practicable with the restrictions on hedging contained in Exhibit D hereto; it being understood that Member shall in no event on or after the date hereof commence any activities that involve the disposition of the economic or other risks of ownership through hedging transactions or derivatives involving Acquiror securities or Member's Membership Interest.
- (j) Member has not, and through the Closing Date will not have, made any election, taken any action or filed any Tax Return or any other document filed or provided to any taxing authority that is inconsistent with the Tax Returns filed and furnished by the Company and the Company's Subsidiaries or the treatment of the Company as a partnership and of each of the Company's Subsidiaries as a partnership or disregarded entity (except as set forth in Section 7.01(n) of the Disclosure Schedule to the Merger Agreement) for U.S. federal, state and local Tax purposes and Member agrees that the foregoing representations shall be subject to the same indemnification provisions as contemplated by Section 13 herein.
- (k) Member will not make any election, take any action or take any position on any Tax Return or any other document filed or provided to any Taxing Authority that is inconsistent with any Tax Return filed with any taxing authority, or furnished to such Member, by or on behalf of the Company, the Company's Subsidiaries or the Surviving LLC.

- (1) Member agrees to provide, and to use its reasonable best efforts to cause its respective relatives and affiliates to provide, any document or take any other action reasonably requested by Acquiror in connection with any Tax matters relating to the Company, the Company's Subsidiaries and the Surviving LLC so long as such action does not adversely affect the Member (or any of its respective relatives or affiliates).
- (m) Member has received, separately considered and executed the Supplemental Members Agreement attached as Exhibit ${\tt E.}$
- $4.\ \mbox{Restrictions}$ on Transfer. (a) Member agrees that the Shares may be Transferred only as follows:
 - (i) 33 1/3% of the Shares may be Transferred at any time after May 8, 2002;
 - (ii) An additional 33 1/3% of the Shares may be Transferred at any time after May 8, 2003; and
 - (iii) All of the Shares may be Transferred at any time after May $8,\ 2004.$

Notwithstanding the foregoing, if Acquiror generally waives the restrictions on transfer imposed upon the former Schedule II limited partners (the "LPs") of The Goldman Sachs Group, L.P. ("Group") in connection with Group's 1999 Plan of Incorporation in order to permit the LPs generally to engage in a registered secondary offering or estate planning transactions, Acquiror will waive the transfer restrictions in this Section 4 with respect to the Shares on (i) the same percentage basis as the waiver granted to LPs in connection with such registered secondary offering and to permit Member to sell such shares in the registered secondary offering and (ii) the same basis as the waiver granted to LPs in connection with such estate planning transactions. (For example, if each LP is permitted to sell 5% of his Acquiror Common Stock in a registered secondary offering, Member shall be permitted to sell 5% of Member's Shares; and if each LP is permitted to engage in estate planning transactions, Member shall be permitted to do so to a similar extent.)

For purposes of this Agreement, the term "Transfer" means any direct or indirect sale, transfer, pledge or other disposition of securities of Acquiror or Member's Membership Interests, as the case may be, including any disposition of the economic or other risks of ownership through hedging transactions or derivatives involving Acquiror securities or Member's Membership Interests; provided, however, that in the event of such Member's death during the term of this Agreement, Member's Shares may be transferred in the same manner and to the same extent as the LPs may transfer their shares of Acquiror Common Stock pursuant to the Acquiror Shareholders Agreement.

 $\,$ (c) Member agrees that, at any time Member is employed by the Firm, it will:

- (i) comply with respect to all Shares with Transfer restrictions related to future primary or secondary offerings of Shares if requested to do so by Acquiror to the extent that such restrictions are generally applicable to similarly titled employees of Acquiror;
- (ii) comply with restrictions that may be imposed by Acquiror from time to time to enable Acquiror or another party to account for a business combination using the pooling-of-interests method of accounting to the extent that such restrictions are generally applicable to similarly titled employees of Acquiror;
- (iii) be subject to the same internal compliance and trading policies as are in effect from time to time for similarly titled employees of Acquiror; and
- (iv) comply with the hedging restrictions of Acquiror relating to securities of Acquiror and financial services companies as are in effect from time to time for managing directors of Acquiror.

References in this Section 4 to "Shares" shall be deemed to also refer to securities received in exchange for Shares on the same basis as provided in Section 6.4 of the Acquiror Shareholders Agreement.

5. Confidential Information. (a) In the course of involvement in the Firm's activities or otherwise, Member has obtained or may obtain confidential information concerning the Firm's businesses, strategies, operations, financial affairs, organizational and personnel matters (including information regarding any aspect of Member's tenure as a member in, or officer or employee of, the Firm or of the termination of such membership, officership or employment), policies, procedures and other non-public matters, or concerning those of third parties. Such information ("Confidential Information") may have been or be provided in written or electronic form or orally. In consideration of, and as a condition to, continued access to Confidential Information, and without prejudice to or limitation on any other confidential ty obligations imposed by agreement or by law, Member hereby undertakes to use and protect Confidential Information in accordance with any reasonable restrictions placed on its use or disclosure. Without limiting the foregoing, except as authorized by the Firm or as required by law, Member may not disclose, directly or indirectly, any Confidential Information, or any information derived therefrom, in whatever form, to any person unless such person is a director, officer, partner, employee, attorney or agent of the Firm and, in Member's reasonable good faith judgment, has a need to know the Confidential Information or information derived therefrom in furtherance of the business of the Firm. The foregoing obligations will survive, and remain binding and enforceable notwithstanding any termination of Member's employment with the Firm and any settlement of the financial rights and obligations arising from Member's employment with the Firm. Without limiting the foregoing, the existence of, and any information concerning, any dispute

between Member and the Firm shall constitute Confidential Information except that Member may disclose information concerning such dispute to the arbitrator or court that is considering such dispute, or to Member's legal counsel (provided that such counsel agrees not to disclose any such information other than as necessary to the prosecution or defense of the dispute).

- (b) For purposes of this Agreement, "Firm" means (i) prior to the consummation of the Merger, the Company and its Subsidiaries and Affiliates, and (ii) from and after the consummation of the Merger, Acquiror and its Subsidiaries and Affiliates (including the Company and its Subsidiaries).
- 6. Noncompetition. (a) In view of Member's importance to the Firm, Member hereby agrees that the Firm would likely suffer significant harm from Member's competing with the Firm for some period of time following the consummation of the Merger, and at any time prior to the date of termination specified in the notice of termination pursuant to Section 10 hereof (the "Employment Period") and for some time thereafter, and Member understands that Acquiror would not have agreed to acquire the Company and its business unless Member entered into this Agreement. Moreover, Member recognizes and agrees that the business activities of the Firm are worldwide and that the restrictions on competition included herein are commensurate in geographic scope with those activities. Accordingly, Member hereby agrees that commencing at the time of consummation of the Merger, Member will not, without the written consent of Acquiror, until the later of (x) three years following the Effective Time and (y) two years following the date of termination of the Employment Period (such later date is referred to as the "Expiration Date"):
 - (1) form, or acquire a 5% or greater equity ownership, voting or profit participation interest in, any Competitive Enterprise (as defined below); or
 - (2) associate (including, but not limited to, association as an officer, employee, partner, director, consultant, agent or advisor) with any Competitive Enterprise and in connection with such association engage in, or directly or indirectly manage or supervise personnel engaged in, any activity
 - (i) which is similar or substantially related to any activity in which Member was engaged, in whole or in part, at the Firm,
 - (ii) for which Member had direct or indirect managerial or supervisory responsibility at the Firm, or
 - (iii) which calls for the application of the same or similar specialized knowledge or skills as those utilized by Member in Member's activities with the Firm,

at any time during the one-year period immediately prior to the Expiration Date (or, in the case of an action taken during the Employment Period, during the one-year period immediately prior to such action), and, in any such case, irrespective of the purpose of the activity or whether the activity is or was in furtherance of advisory, agency, proprietary or fiduciary business of either the Firm or the Competitive Enterprise.

(By way of example only, this provision precludes an information systems professional from joining a management or other consulting firm and providing information technology consulting services or advice to any Competitive Enterprise.)

(b) For purposes of this Agreement, a "Competitive Enterprise" is a business enterprise that (1) engages in any activity, or (2) owns or controls a significant interest in any entity that engages in any activity, that, in either case, competes anywhere with any activity in which the Firm is engaged. The activities covered by the previous sentence include, without limitation, financial services such as investment banking, public or private finance, lending, financial advisory services, private investing (for anyone other than Member and members of Member's family), merchant banking, asset or hedge fund management, insurance or reinsurance underwriting or brokerage, property management, or securities, futures, commodities, energy, derivatives or currency brokerage, sales, market making, lending, custody, clearance, settlement or trading. It is the intent that the provisions of this Section 6 shall include, for the entire period through the Expiration Date, the activities described in subparagraphs 1 through 4 of Article 14.E of the Operating Agreement as in effect as of the date hereof.

If Member is employed by the Firm immediately following the Effective Time and the Firm terminates Member's employment with the Firm without Cause, the Firm may only enforce the provisions of this Section 6 for so long as the Firm continues to pay Member the same base salary (i.e., excluding any incentive, bonus or similar compensation) Member was receiving immediately prior to such termination. In determining whether the Firm has paid Member's base salary for any period, the Firm shall receive credit for any payments under any severance, salary continuation or similar plan or arrangement.

For the purposes of this Section 6 and Section 10 only, "Cause" means (i) Member's conviction, whether following trial or by plea of guilty or nolo contendere (or similar plea), in a criminal proceeding (A) on a misdemeanor charge involving fraud, false statements or misleading omissions, wrongful taking, embezzlement, bribery, forgery, counterfeiting or extortion, or (B) on a felony charge or (C) on an equivalent charge to those in clauses (A) and (B) in jurisdictions which do not use those designations; (ii) Member's engaging in any conduct which constitutes an employment disqualification under applicable law (including statutory disqualification as defined under the Exchange Act); (iii) Member's willful or grossly negligent failure to perform Member's duties to the Firm; (iv) Member's violation of any securities or commodities laws, any rules or regulations issued pursuant to such laws, or the rules and regulations of any securities or commodities exchange or

association of which Acquiror or any of its subsidiaries or affiliates is a member; (v) Member's violation of any Firm policy concerning hedging or confidential or proprietary information, or Member's material violation of any other Firm policy as in effect from time to time; (vi) Member's engaging in any act or making any statement which impairs, impugns, denigrates, disparages or negatively reflects upon the name, reputation or business interests of the Firm; or (vii) Member's engaging in any willful or grossly negligent conduct detrimental to the Firm. The determination as to whether "Cause" has occurred shall be made by the Board of Directors of Acquiror in its good faith judgment.

- 7. Nonsolicitation of Clients. (a) Member hereby agrees that during the Employment Period and thereafter until the Expiration Date, Member will not, in any manner, directly or indirectly, (1) Solicit a Client to transact business with a Competitive Enterprise or to reduce or refrain from doing any business with the Firm, or (2) interfere with or damage (or attempt to interfere with or damage) any relationship between the Firm and a Client.
- (b) For purposes of this Agreement, the term "Solicit" means any direct or indirect communication of any kind whatsoever, regardless of by whom initiated, inviting, advising, encouraging or requesting any person or entity, in any manner, to take or refrain from taking any action. It is the intent that the provisions of this Section 7 shall include, for the entire period through the Expiration Date, the activities described in subparagraph 5 of Article 14.E of the Operating Agreement as in effect as of the date hereof.
- (c) For purposes of this Agreement, the term "Client" means any client or prospective client of the Firm to whom Member provided services, or for whom Member transacted business, or whose identity became known to Member in connection with Member's relationship with or employment by the Firm.
- 8. Nonsolicitation of Employees; Expiration. Member hereby agrees that during the Employment Period and thereafter until the Expiration Date, Member will not, in any manner, directly or indirectly, Solicit any person who is a Member or former Member of the Company or a key employee of the Firm to resign from the Firm or to apply for or accept employment with any Competitive Enterprise.
- 9. Damages. (a) Member acknowledges that Acquiror would not have entered into this Agreement or the Merger Agreement in the absence of the Member's agreement to the provisions of this Section 9 and the Covenants and Member further acknowledges that such compliance with such Covenants is an important factor to the continued success of the Firm's operations and its future prospects. Member and Acquiror agree that upon the occurrence of any of the following events, the damages to the Firm would be material, but that the amount of such damages would be uncertain and not readily ascertainable. Accordingly, Member and Acquiror agree that, if, prior to the fifth anniversary of the date of this Agreement, Member breaches any of the Covenants set forth in Section 6, 7 or 8, as determined by the Board of Directors of Acquiror (the "Board") in its

good faith judgment, Acquiror will be entitled to receive immediately following such determination and written demand therefor, and Member will make, within ten business days after written demand has been received, a payment in cash or Acquiror Common Stock (valued at the average closing per share price of Acquiror Common Stock for the five trading days immediately preceding the date of payment under this Section 9(a)) in such combination of cash and/or Acquiror Common Stock as the Member shall determine as and for liquidated damages (the "Liquidated Damages") in the amount set forth next to such Member's name in Exhibit C (under the heading "Liquidated Damages") attached hereto, which amount shall be the aggregate amount of liquidated damages due for all such breaches prior to the fifth anniversary of this Agreement.

The payment of any amount as liquidated damages will not be construed as a release or waiver by Acquiror of the right to prevent the continuation of any such violation of such Covenants in equity or otherwise. In addition, Member and Acquiror agree that it would be too speculative to attempt to determine any amount of liquidated damages that would be applicable following the fifth anniversary of the date of this Agreement, and that any damages payable as a result of any breach following such date shall be determined without regard to this Section 9.

- (b) Member and Acquiror agree that the Liquidated Damages are reasonable in proportion to the probable damages likely to be sustained by the Firm if Member breaches at any time prior to the fifth anniversary of this Agreement any of the Covenants set forth in Sections 6, 7 and 8 hereof, that the amount of actual damages to be sustained by the Firm in the event of such breach is incapable of precise estimation and that such cash payments are not intended to constitute a penalty or punitive damages for any purposes.
- (c) Member acknowledges and agrees that Member's payment obligations under this Section 9 will be full recourse obligations and will be secured pursuant to a Pledge Agreement, in substantially the form set forth as Exhibit F hereto (the "Pledge Agreement"). Member agrees to execute and deliver to Acquiror a Pledge Agreement prior to the Closing.
- (d) Member acknowledges and agrees that any cash payment of Liquidated Damages pursuant to this Section 9 shall be in addition to, and not in lieu of, any forfeitures of awards (required pursuant to the terms of any such awards) that may be granted to Member in the future under one or more of the Firm's compensation and benefit plans.
- 10. Employment with the Firm. (a) Member acknowledges that Acquiror would not have entered into this Agreement or the Merger Agreement in the absence of Member's agreement to the provisions of this Section 10, and Member further acknowledges that such Member's continued employment with the Firm through the second anniversary of the Effective Time is essential to assure the proper integration of the business operations of the Company into the Firm and is an important factor to the continued success of the Firm's operations and its future prospects. Accordingly, Member hereby agrees to remain employed with the Firm for the period commencing on the Effective Time through the second

anniversary of the Effective Time (the "Initial Employment Period"). After the Initial Employment Period (unless otherwise agreed by Member and the Firm in writing), there will be no set term of employment. The Firm may terminate Member's employment at any time during or after the Initial Employment Period for any reason, or for no reason, and Member may terminate employment for any reason or no reason after the Initial Employment Period. Such termination shall be effected only by giving not less than ninety (90) days' prior written notice of termination; provided, however, that (i) the Firm may elect to place Member on paid leave for all or any part of such 90-day period; (ii) no advance notice need be given by the Firm to Member in connection with a termination of Member's employment for Cause or on account of Disability; and (iii) provided that the 90 days prior written notice is given, Member may terminate his employment during the Initial Employment Period on account of Good Reason, Disability or with Acquiror's written consent. For purposes of this Section 10, "Good Reason" means, without the consent of Member, a materially adverse alteration in Member's position or in the nature or status of Member's responsibilities from those in effect immediately after the Effective Time, or (ii) the Firm's requiring Member's principal place of employment to be located more than seventy-five (75) miles from the location at which Member is principally employed immediately after the Effective Time (except for required travel on the Firm's business to an extent substantially consistent with Member's customary business travel obligations in the ordinary course of business prior to the Effective Time). For purposes of this Section 10, "Disability" means Member's absence from employment for at least 180 days in any 12-month period as a result of Member's incapacity due to mental or physical illness or incapacity, as reasonably determined by the Firm.

(b) At the outset of the Employment Period, Member's duties and responsibilities will not be, without Member's written consent, materially diminished from Member's duties and responsibilities immediately prior to the Effective Time (it being understood that any diminishment in duties or responsibilities as a result of the Company and its Affiliates being Subsidiaries of Acquiror shall be disregarded). During the Employment Period: (i) Member will have such duties and responsibilities as the Firm may from time to time determine; (ii) Member will devote his entire working time, labor, skill and energies to the business and affairs of the Firm, provided, however, that Member shall not be prohibited from making passive personal investments or conducting private affairs if those activities do not interfere with the services required under this Agreement; and (iii) Member will be paid such base salary and other compensation as shall be separately communicated to him.

(c) It is understood and agreed that the provisions of this Section 10 shall not apply to: Pumpkin Trust; Michael Appleby; Patrick C. Boyle; Michael H. Davis; Robert W. Luckow; Lowell J. Millar; William G. Peskoff; Norman R. Schlanger; Alfred Thomas; and Frank Weinberg III (collectively, the "Retiring Members").

11. Transfer of Client Relationships. (a) During the Coverage Period, Member hereby agrees to take all actions and do all such things as may be reasonably requested by the Firm from time to time to maintain for the Firm the business, goodwill, and

business relationships with any of the Firm's Clients with whom Member worked during the term of Member's employment with the Firm.

- (b) For purposes of this Agreement, the term "Coverage Period" means, the 90-day period beginning on the date on which notice of Member's termination of employment is delivered to or by the Firm pursuant to Section 10, or in the case of termination of Member's employment by the Firm for Cause or on account of Extended Absence, the 90-day period beginning on the date of termination.
- 12. Prior Notice Required. Member hereby agrees that prior to accepting employment with any other person or entity prior to the date of termination, Member will provide such prospective employer with written notice of the provisions of this Agreement, with a copy of such notice delivered simultaneously to the General Counsel of Acquiror.
- 13. Indemnification. Member hereby agrees that it shall indemnify each Acquiror Party as an Indemnifying Party as set forth in Articles 7 and 9 of the Merger Agreement.

14. Taxes.

- (a) Tax Returns. Member hereby agrees to prepare and file (or cause to be prepared and filed) when due (taking into account any applicable extensions) all of his, her or its Tax Returns related to periods during which the Merger occurs, and Member will timely pay all Taxes reflected on such Tax Returns (or which are due with respect to such Tax Returns after adjustment by any taxing authority). Member further agrees to cooperate with Acquiror and to provide Acquiror with any documentation as reasonably requested in establishing the timely filing of such Tax Returns and timely payment of such Taxes.
- (b) Tax Treatment of the Common Stock. Member agrees that the Acquiror Common Stock subject to the restrictions on Transfer (as described in Section 4 herein) and received by the Members in the Merger will be valued for all U.S. federal income tax purposes at 100% of the mean of the high and low of the trading price of the Acquiror Common Stock on the Effective Date.
- (c) Cooperation. Member agrees to cooperate with the Tax Matters Partner (as defined in Section 6231(a)(7) of the Code) (or its designee) to the extent reasonably requested in any Tax audit, examination or other proceeding involving the Company, the Partnership or any Subsidiary that is treated as a partnership for U.S. federal income tax purposes for any taxable period that relates to periods prior to the Closing. Member further agrees that the existing Tax Matters Partner (or if the existing Tax Matters Partners does not so act, the Acquiror (or its designee)) shall have the right to designate or appoint any individual or entity in the name and on behalf of such Member, as the Tax Matters Partner, with respect to any taxable year of the Company, the Partnership or any of their Subsidiaries.

(d) Overpayments and Underpayments. The parties hereby agree that in the event the Company makes distributions to Member pursuant to the terms of 4.01(c) (i) or (iv) of the Merger Agreement, the parties shall as promptly as practicable after the Closing determine whether the actual amount of Taxes owed by Member in respect of the net earnings of the Company during the period from July 1, 2000 to the Closing Date (after taking into account Tax credits and other available Tax assets and not including any Taxes payable as the result of sales or other dispositions requiring the recognition of unrealized gains reflected on the June 30, 2000 Financial Statements (including the "short against the box" positions)) is greater than (an "Underpayment") or less than (an "Overpayment") the amounts distributed to such Member with respect to the period from July 1, 2000 to the Closing Date in respect to such Taxes, and, in the event of an Overpayment, Member shall promptly pay to Acquiror and, in the event of an Underpayment, Acquiror shall promptly pay to Member, the amount of such difference.

15. Intentionally Omitted.

- 16. Covenants Generally. (a) Member's covenants as set forth in Sections 5 through 14 of this Agreement are from time to time referred to herein as the "Covenants." If any of the Covenants is finally held to be invalid, illegal or unenforceable (whether in whole or in part), such Covenant shall be deemed modified to the extent, but only to the extent, of such invalidity, illegality or unenforceability and the remaining such Covenants (or part of such Covenants, as the case may be) shall not be affected thereby; provided, however, that if any of such Covenants is finally held to be invalid, illegal or unenforceable because it exceeds the maximum scope determined to be acceptable to permit such provision to be enforceable, such Covenant will be deemed to be modified to the minimum extent necessary to modify such scope in order to make such provision enforceable hereunder.
- (b) Member understands that the provisions of the Covenants may limit Member's ability to earn a livelihood in a business similar to the business of the Firm.
- (c) Member acknowledges that a violation on Member's part of any of the Covenants would cause irreparable damage to the Firm. Accordingly, Member agrees that the Firm will be entitled to injunctive relief for any actual or threatened violation of any of the Covenants in addition to any other remedies it may have.
- (d) Acquiror will at all times maintain an effective S-8 or other registration statement covering shares of Acquiror Common Stock to be delivered pursuant to the restricted stock unit awards specified in Section 3.05 of the Merger Agreement and in connection with the administration of such awards shall treat the Member in the same manner as similarly titled employees of Acquiror and its subsidiaries.
- 17. Waiver and Release. (a) Member hereby irrevocably waives any right to contest the terms of the Merger Agreement and the transactions contemplated thereby,

whether on the grounds of unequal or disparate treatment, inconsistency or conflict with the terms and provisions of the Operating Agreement.

- (b) Member hereby irrevocably releases Acquiror, each and every affiliate, shareholder, subsidiary, partner, officer, member, director and employee of Acquiror and its affiliates in their capacities as such ("Releasees") from any claims, liabilities, costs, expenses, actions, suits or demands however arising, whether at law or in equity, contingent, known or unknown, which Member or his heirs, successors or assigns may have or assert, in respect of any interest in the Company and its affiliates or arising out of any Membership Interest, Partnership Interest, or Member, partnership or employment relationship with the Company or its affiliates which Member or Member's heirs, successors or assigns may have or have had; provided that this release shall not extend to (i) agreements entered into hereunder or in connection with the transactions contemplated by the Merger Agreement and (ii) any conduct that resulted from a Releasee's bad faith, fraud or criminal act or omission.
- 18. Arbitration. Subject to the provisions of Sections 19 and 20 hereof, any dispute, controversy or claim between Member and the Firm arising at or after the Effective Time out of or relating to or concerning the provisions of this Agreement, relating to or arising out of Member's employment with the Firm or otherwise concerning any rights, obligations or other aspects of Member's employment relationship in respect of the Firm ("Employment Related Matters"), shall be finally settled by arbitration in New York City before, and in accordance with the rules then obtaining of, the New York Stock Exchange, Inc. (the "NYSE") or, if the NYSE declines to arbitrate the matter, the American Arbitration Association (the "AAA") in accordance with the commercial arbitration rules of the AAA.
- 19. Injunctive Relief; Submission to Jurisdiction; Specific Performance. (a) Notwithstanding the provisions of Section 18, and in addition to its right to submit any dispute or controversy to arbitration, the Firm may bring an action or special proceeding in a state or federal court of competent jurisdiction sitting in the City of New York, whether or not an arbitration proceeding has theretofore been or is ever initiated, for the purpose of temporarily, preliminarily, or permanently enforcing the provisions of the Covenants, or to enforce an arbitration award, and, for the purposes of this Section 19, Member (i) expressly consents to the application of Section 20 to any such action or proceeding, (ii) agrees that proof will not be required that monetary damages for breach of the provisions of the Covenants would be difficult to calculate and that remedies at law would be inadequate and (iii) irrevocably appoints the General Counsel of Acquiror as Member's agent for service of process in connection with any such action or proceeding, who shall promptly advise Member of any such service of process.
- (b) Each party hereto severally acknowledges that it will be impossible to measure in money the damage to the other party if the party hereto fails to comply with any of the obligations imposed by this Agreement and that every such obligation is material. Accordingly, each party hereto severally agrees that injunctive relief or other equitable remedy, in addition to remedies at law or damages, is the appropriate remedy for any such

failure and will not oppose the granting of such relief on the basis that the other party has an adequate remedy at law.

- 20. Choice of Forum. (a) MEMBER AND THE FIRM HEREBY IRREVOCABLY SUBMIT TO THE EXCLUSIVE JURISDICTION OF ANY STATE OR FEDERAL COURT LOCATED IN THE CITY OF NEW YORK OVER ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO OR CONCERNING THIS AGREEMENT OR ANY EMPLOYMENT RELATED MATTER THAT IS NOT OTHERWISE ARBITRATED ACCORDING TO THE PROVISIONS OF SECTION 18 HEREOF. This includes any suit, action or proceeding to compel arbitration or to enforce an arbitration award. This also includes any suit, action, or proceeding arising out of or relating to any post-employment Employment Related Matters. Member and the Firm acknowledge that the forum designated by this Section 20 has a reasonable relation to this Agreement, and to Member's relationship to the Firm. Notwithstanding the foregoing, nothing herein shall preclude the Firm from bringing any action or proceeding in any other court for the purpose of enforcing the provisions of Sections 18, 19 or 20.
- (b) The agreement of Member and the Firm as to forum is independent of the law that may be applied in the action, and Member and the Firm agree to such forum even if the forum may under applicable law choose to apply non-forum law. Member and the Firm hereby waive, to the fullest extent permitted by applicable law, any objection which Member or the Firm now or hereafter may have to personal jurisdiction or to the laying of venue of any such suit, action or proceeding in any court referred to in Section 20(a). Member and the Firm undertake not to commence any action arising out of or relating to or concerning this Agreement in any forum other than a forum described in this Section 20. Member and the Firm agree that, to the fullest extent permitted by applicable law, a final and non-appealable judgment in any such suit, action or proceeding in any such court shall be conclusive and binding upon Member and the Firm.
- 21. Choice of Law. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO PRINCIPLES OF CONFLICT OF LAWS.
- 22. No Right to Employment. Nothing in this Agreement or the Merger Agreement shall confer upon Member the right to employment or continued employment by the Firm or affect the Firm's right to terminate such employment at will.
- $23.\ {\tt Miscellaneous.}$ (a) This Agreement and the Merger Agreement shall supersede any other agreement, written or oral, pertaining to the matters covered herein.
- (b) Notices hereunder shall be delivered to Acquiror at its principal executive office directed to the attention of its General Counsel, and to Member at Member's last address appearing in the Firm's employment records.

- (c) This Agreement may not be amended or modified, other than by a written agreement executed by Member and Acquiror or its successors, nor may any provision hereof be waived other than by a writing executed by Member or Acquiror or its successors; provided, that any waiver, consent, amendment or modification of any of the provisions of this Agreement will not be effective against the Firm without the written consent of Acquiror or its successors. Member may not, directly or indirectly (including by operation of law), assign Member's rights or obligations hereunder without the prior written consent of Acquiror or its successors, and any such assignment by Member in violation of this Agreement shall be void. This Agreement shall be binding upon Member's permitted successors and assigns. Without impairing Member's obligations hereunder, Acquiror may at any time and from time to time assign its rights and obligations hereunder to any of its subsidiaries or affiliates (and have such rights and obligations reassigned to it or to any other subsidiary or affiliate). This Agreement shall be binding upon and inure to the benefit of the Firm and its assigns.
- (d) Without limiting the provisions of Section 16(a) hereof, if any provision of this Agreement is finally held to be invalid, illegal or unenforceable (whether in whole or in part), such provision shall be deemed modified to the extent, but only to the extent, of such invalidity, illegality or unenforceability and the remaining provisions shall not be affected thereby.
- (e) Except as expressly provided herein, this Agreement shall not confer on any person other than Acquiror, the Firm and each Member any rights or remedies hereunder.
- (f) The captions in this Agreement are for convenience of reference only and shall not define or limit the provisions hereof.

IN WITNESS WHEREOF, each signatory hereto has caused this Agreement, including the written consent evidenced hereby, to be executed and delivered this 10th day of September, 2000.

THE GOLDMAN SACHS GROUP, INC. (on its behalf, and on behalf of its subsidiaries and affiliates)
By:
Name:
SLK LLC (with respect to Section 1(b) only) By SLK Management Inc., its Managing Member
By:
Name:
(As received by SLK LLC at its principal office)
MEMBER
Name:

INDEMNIFICATION AGREEMENT

THIS INDEMNIFICATION AGREEMENT (this "Agreement") is made and entered into as of the 5th day of July 2000, by and between The Goldman Sachs Group, Inc., a Delaware corporation ("GS Inc."), and each of the Indemnitees listed on the signature pages to this Agreement (each, an "Indemnitee", and collectively, the "Indemnitees") as such signature pages may be amended and supplemented from time to time pursuant to the terms of this Agreement.

WITNESSETH

WHEREAS, GS Inc. has filed and proposes to file registration statements with the Securities and Exchange Commission under the Securities Act of 1933 (the "Securities Act") for the public offering and sale (the "Public Offerings") of securities of GS Inc., which Securities may include shares of its common stock (including shares to be sold by stockholders of GS Inc. or issuable in connection with employee benefit plans), debt securities (including medium-term notes), guarantees, back-up undertakings, rights, options, warrants, preferred stock and/or any other securities of GS Inc. (or interests in any employee benefit plan sponsored by or established by GS Inc.) approved by, or pursuant to action of, the Board of Directors of GS Inc. (the "Securities");

WHEREAS, GS Inc. and/or stockholders of GS Inc. has sold and may in the future sell Securities in transactions not requiring registration under the Securities Act, including, without limitation, in private placements, pursuant to Rule 144 or Rule 144A under the Securities Act and pursuant to Regulation S under the Securities Act ("Unregistered Offerings" and, together with "Public Offerings," the "Sales");

WHEREAS, GS Inc. has in the past requested and will in the future request certain of the Indemnitees to execute registration statements in the capacity or capacities listed or to be listed in registration statements, to act as an attorney-in-fact in connection with Sales and to take other actions in connection with such registration statements and Sales; and

WHEREAS, each Indemnitee is one of the following: (i) an officer or director of GS Inc. or (ii) a person requested or authorized by the Board of Directors of GS Inc. or any committee thereof to take actions on behalf of GS Inc. in connection with a Sale.

NOW, therefore, in consideration of each Indemnitee's acting and agreeing to act in the capacities referred to above, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereby agree as follows:

1. General. GS Inc. will indemnify and hold harmless each Indemnitee against any Losses (as hereinafter defined), joint or several, to which such Indemnitee may become subject, under the Securities Act or otherwise, insofar as such Losses (or actions in respect thereof) arise out of or are based upon an untrue statement or alleged untrue statement of a material fact contained in any Registration Statement or Offering Circular (in each case, as defined below) or any preliminary prospectus or preliminary Offering Circular, prospectus, or prospectus supplement or Offering Circular supplement comprising a part thereof or relating thereto, or any amendment or supplement to any of the foregoing documents (collectively, the "Offering Documents") or any untrue or alleged untrue oral statement relating to any offering contemplated by any Offering Document, arise out of or are based upon an omission or alleged omission to state in any Offering Document or such oral statement a material fact required to be stated therein or necessary to make the statements in any Offering Document or such oral statement not misleading, or arise out of or are based upon or asserted against any Indemnitee in connection with such Indemnitee's acting in the capacity of attorney-in-fact or any other authorized capacity in connection with any Sale; provided, however, that GS Inc. shall not be liable in any such case to the extent that any such Losses arise out of or are based upon an untrue statement or alleged untrue statement or omission or alleged omission relating to such Indemnitee made in any Offering Document or such oral statement in reliance upon and in conformity with written information relating to such Indemnitee furnished to GS Inc. by such Indemnitee expressly for use therein. "Registration Statement" means any registration statement previously filed or hereafter filed by GS Inc. under the Securities Act on any applicable form (including Forms S-8 and S-4) for the registration of any Securities under the Securities Act, and shall include any amendment, post-effective or otherwise, thereto and any related registration statement filed pursuant to Rule 462 under the Securities Act. "Offering Circular" means any offering document or instrument previously used or hereafter used by GS Inc. to offer or sell any Securities in a transaction or series of transactions not requiring registration under the Securities Act, including, without limitation, any private placement memorandum or offering circular relating to an offering of Securities made pursuant to Regulation S.

Notwithstanding the foregoing provisions of this Section 1, GS Inc. and each Indemnitee agree that insofar as indemnification for liabilities arising under the Securities Act may be permitted under this Agreement to an Indemnitee who is a director, officer or controlling person of GS Inc., in the event that a claim for indemnification against such liabilities is made by such an Indemnitee (other than the payment by GS Inc. of expenses incurred or paid by such Indemnitee in the successful defense of any action, suit or proceeding) in connection with a Registration Statement, GS Inc. will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is

against public policy as expressed in the Securities Act, and GS Inc. and such Indemnitee will be governed by the final adjudication of such question.

- 2. Losses. As used in this Agreement, the term "Losses" shall include, without limitation, damages, losses, claims, judgments, liabilities, fines, penalties, excise taxes, settlements, and costs, attorneys' fees, accountants' fees, and disbursements and costs of attachment or similar bonds, investigation costs, defense preparation costs, costs of preparing for and presenting evidence or testimony, and any expenses of establishing a right to indemnification under this Agreement. The term "Losses" shall not include taxes except to the extent taxes are imposed in respect of payments otherwise made pursuant to this Agreement, in which case such Indemnitee's Losses shall include an amount not greater than the net taxes payable (taking into account any deductions, credits or other tax benefits available to such Indemnitee as a result of the Losses in respect of which such payment is made and the payment of the taxes imposed in respect of such payment).
- 3. Enforcement. Subject to the provisions of the second paragraph of Section 1 hereof, if a claim or request by an Indemnitee under this Agreement is not paid by GS Inc. or on its behalf, within thirty (30) days after a written claim or request has been received by GS Inc. and, if applicable, the written undertaking in Section 5 hereof has been received by GS Inc., such Indemnitee may at any time thereafter commence an arbitration proceeding in accordance with Section 9 hereof against GS Inc. to recover the unpaid amount of the claim or request and, if successful in whole or in part, such Indemnitee shall be entitled to be paid also the expenses of prosecuting such proceeding.
- 4. Partial Indemnification; Contribution. If an Indemnitee is entitled under any provision of this Agreement to indemnification by GS Inc. for some or a portion of any Losses, but not for the total amount thereof, GS Inc. shall nevertheless indemnify such Indemnitee for the portion of such Losses to which such Indemnitee is entitled. If the indemnification provided for in this Agreement is insufficient or unavailable for any reason, GS Inc. shall contribute to relevant Losses to the maximum extent permitted by law.
- 5. Expenses. Expenses incurred by an Indemnitee in connection with any proceeding shall be paid by GS Inc. upon request of such Indemnitee that GS Inc. pay such expenses, but only upon receipt by GS Inc. of (i) in the case of a Public Offering, a written undertaking by or on behalf of such Indemnitee to reimburse GS Inc. for expenses if and to the extent that it is ultimately determined that such indemnification is not permitted by the Securities Act (and that contribution is unavailable with respect to such payments) and (ii) satisfactory evidence of the amount of such expenses.

- 6. Notice of Claim. Each Indemnitee shall promptly notify GS Inc. in writing of any claim against such Indemnitee for which indemnification will or could be sought under this Agreement. In addition, each Indemnitee shall give GS Inc. such information and cooperation as it may reasonably require and as shall be within such Indemnitee's power and at such times and places as are not unduly burdensome for such Indemnitee.
- 7. Defense of Claim. With respect to any proceeding as to which an Indemnitee notifies GS Inc. of the commencement thereof:
 - (a) GS Inc. will be entitled to participate at its own expense;
 - (b) subject to Section 7(c) hereof, GS Inc. shall not, in connection with any proceeding or related proceedings in the same jurisdiction against any Indemnitee and any other Indemnitees, be liable to such Indemnitee and such other Indemnitees for the fees and expenses of more than one separate law firm (in addition to a single firm of local counsel);
 - (c) except as otherwise provided below, to the extent that it may wish, GS Inc. will be entitled to assume the defense thereof, with counsel reasonably satisfactory to such Indemnitee, which in GS Inc.'s sole discretion may be regular counsel to GS Inc. and may be counsel to other Indemnitees. The Indemnitees also shall have the right to employ one separate counsel for such Indemnitees in such action, suit or proceeding if such Indemnitees reasonably conclude that if they did not there would be a conflict of interest between GS Inc. and such Indemnitees, and under such circumstances the fees and expenses of such counsel shall be paid by GS Inc.; and
 - (d) GS Inc. shall not be liable to indemnify an Indemnitee under this Agreement for any amounts paid in settlement of any action, suit or proceeding effected without GS Inc.'s written consent. GS Inc. shall not settle any action, suit or proceeding in any manner which would impose any cost or limitation on an Indemnitee or would admit fault by an Indemnitee without such Indemnitee's written consent. No Indemnitee shall settle any action, suit, or proceeding without the prior written consent of GS Inc. Neither GS Inc. nor an Indemnitee will unreasonably withhold or delay its consent to any proposed settlement.
- 8. Non-exclusivity. The right to indemnification and the payment of expenses incurred in defending a proceeding in advance of its final disposition conferred

in this Agreement shall not be exclusive of or affected in any way by any other right which an Indemnitee may have or hereafter may acquire under any statute, certificate of incorporation, by-laws, agreement, arrangement, resolution or instrument providing indemnification or expense payment, except that any payments otherwise required to be made by GS Inc. hereunder shall be offset by any and all amounts received by an Indemnitee from any other indemnitor or under one or more liability insurance policies maintained by an indemnitor or otherwise and shall not be duplicative of any other payments received by an Indemnitee from GS Inc. in respect of the matter giving rise to the indemnity hereunder. When an Indemnitee is entitled to indemnification, expense advancement or reimbursement under this Instrument and any other agreement, arrangement, resolution or instrument of GS Inc., the Indemnitee may choose to pursue its rights under one or more, but less than all, of such applicable agreements, arrangements, resolutions or instruments, in which case such Indemnitee need only comply with the standards and procedures of the agreements, arrangements, resolutions or instruments under which it chooses to pursue its rights.

9. Arbitration.

- (a) Subject to the provisions of the second paragraph of Section 1 and Section 9(b) hereof, any dispute, controversy or claim between an Indemnitee and GS Inc. arising out of or relating to or concerning the provisions of this Agreement shall be finally settled by arbitration in New York City before, and in accordance with the rules then applying of, the New York Stock Exchange, Inc. ("NYSE") or, if the NYSE declines to arbitrate the matter or the matter is not otherwise arbitrable before it, the American Arbitration Association (the "AAA") in accordance with the commercial arbitration rules of the AAA.
- (b) Notwithstanding the provision of Section 9(a) and in addition to its right to submit any dispute or controversy to arbitration, GS Inc. may bring an action or special proceeding in a state or federal court of competent jurisdiction sitting in the State of Delaware, whether or not an arbitration proceeding has theretofore been or is ever initiated, for the purpose of temporarily, preliminarily or permanently enforcing the provisions of this Agreement or to enforce an arbitration award, and, for the purposes of this Section 9(b), each Indemnitee (i) expressly consents to the application of Section 9(c) hereof to any such action or proceeding, (ii) agrees that proof shall not be required that monetary damages for breach of the provisions of this Agreement would be difficult to calculate and that remedies at law would be inadequate and (iii) irrevocably appoints each General Counsel of GS Inc., c/o The Corporation Trust Company, Corporation Trust Center, 1209 Orange Street, Wilmington,

Delaware 19801 as such Indemnitee's agent for service of process in connection with any such action or proceeding, who shall promptly advise such Indemnitee of any such service of process.

- (c) (i) EACH INDEMNITEE HEREBY IRREVOCABLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF ANY STATE OR FEDERAL COURT LOCATED IN THE STATE OF DELAWARE OVER ANY SUIT, ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO OR CONCERNING THIS AGREEMENT THAT IS NOT OTHERWISE ARBITRATED ACCORDING TO THE PROVISIONS OF SECTION 9(a) HEREOF. This includes any suit, action or proceeding to compel arbitration or to enforce an arbitration award. The parties acknowledge that the forum designated by this Section 9(c) has a reasonable relation to this Agreement, and to the parties' relationship with one another. Notwithstanding the foregoing, nothing herein shall preclude GS Inc. from bringing any action or proceeding in any other court for the purpose of enforcing the provisions of this Section 9.
- (ii) The agreement of an Indemnitee as to forum is independent of the law that may be applied in the action, and each Indemnitee agrees to this forum even if the forum may under applicable law choose to apply non-forum law. Each Indemnitee hereby waives, to the fullest extent permitted by applicable law, any objection which such Indemnitee now or hereafter may have to personal jurisdiction or to the laying of venue of any such suit, action or proceeding in any court referred to in Section 9(c)(i). The parties undertake not to commence any action arising out of or relating to this Agreement in any forum other than the forum described in this Section 9(c). The parties agree that, to the fullest extent permitted by applicable law, a final and non-appealable judgment in any such suit, action or proceeding in any such court shall be conclusive and binding upon the parties.
- 10. Binding Effect. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective successors, assigns (including any direct or indirect successor by merger or consolidation), heirs, executors and administrators.
- $11.\ \,$ GOVERNING LAW. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF DELAWARE, WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OF LAWS.
- 12. Amendment. Each party understands that from time to time certain other persons may become Indemnitees and certain Indemnitees will cease to be Indemnitees to the extent provided in this Section 12. Accordingly, this Agreement may

be amended by action of GS Inc. from time to time to add additional Indemnitees, without the approval of any other person other than such proposed additional Indemnitees, each of whom shall execute a counterpart of the signature page of this Agreement. This Agreement may also be amended by action of GS Inc. and without the approval of any other person to remove an Indemnitee; provided that such amendment shall not be effective unless GS Inc. has provided 30 days prior written notice to the Indemnitee and, in any event, such amendment shall not affect any rights of such Indemnitee to be indemnified in respect of Losses associated with the acts, omissions or status of such Indemnitee through the effective date of such termination (including the right to subsequent indemnification and expense advancement and reimbursement relating to such acts, omissions or status).

- 13. Waiver of Breach. The failure or delay of a party at any time to require performance by any other party of any provision of this Agreement, even if known, shall not affect the right of such party to require performance of that provision or to exercise any right, power, or remedy hereunder, and any waiver by any party of any breach of any provision of this Agreement shall not be construed as a waiver of any continuing or succeeding breach of such provision, a waiver of the provision itself, or a waiver of any right, power, or remedy under this Agreement. No notice to or demand on any party in any case shall, of itself, entitle such party to other or further notice or demand in similar or other circumstances.
- 14. Severability. GS Inc. and each Indemnitee agree that the agreements and provisions contained in this Agreement are severable and divisible, that each such agreement and provision does not depend upon any other provision or agreement for its enforceability, and that each such agreement and provision set forth herein constitutes an enforceable obligation between GS Inc. and such Indemnitee. Consequently, GS Inc. and each Indemnitee hereto agree that neither the invalidity nor the unenforceability of any provision of this Agreement shall affect the other provisions hereof, and this Agreement shall remain in full force and effect and be construed in all respects as if such invalid or unenforceable provision were omitted.
- 15. Notices. Any communication, demand or notice to be given hereunder will be duly given when delivered in writing by hand or first class mail to GS Inc. at its principal executive office or to an Indemnitee at its last address appearing in the business records of GS Inc. (or to such other addresses as a party may designate by written notice to GS Inc.).
- 16. No Assignments. No Indemnitee may assign its rights or delegate obligations under this Agreement without the prior written consent of GS Inc. Any assignment or delegation in violation of this Section 16 shall be null and void.

- 17. No Third Party Rights. Nothing expressed or referred to in this Agreement will be construed to give any person other than the parties to this Agreement any legal or equitable right, remedy or claim under or with respect to this Agreement or any provision of this Agreement. This Agreement and all of its provisions are for the sole and exclusive benefit of the parties to this Agreement and their successors and permitted assigns.
- 18. Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all such counterparts shall together constitute but one and the same instrument.

 $\,$ IN WITNESS WHEREOF, the parties hereto have entered into this Agreement as of the date first written above.

THE GOLDMAN SACHS GROUP, INC.

By: /s/ Gregory K. Palm
Name: Gregory K. Palm
Title: Executive Vice President and
General Counsel

[Signatures Continued on Next Page]

AMENDMENT NUMBER 1 TO THE TAX INDEMNIFICATION AGREEMENT DATED AS OF MAY 7, 1999

AMENDMENT NUMBER 1, dated as of September 5, 2000 (the "First Amendment") to the Tax Indemnification Agreement between The Goldman Sachs Group, Inc., a Delaware corporation (the "Company") and the Indemnitees listed therein (the "Indemnitees"), dated as of May 7, 1999 (the "Tax Indemnification Agreement");

WHEREAS, Section 10 of the Tax Indemnification Agreement authorizes the Company to amend the Tax Indemnification Agreement in any respect so long as such amendment does not materially adversely affect the amount an Indemnitee is otherwise entitled to receive from the Company pursuant to the Tax Indemnification Agreement; and

WHEREAS, the Company desires to amend the Tax Indemnification Agreement as set forth herein in order to clarify certain provisions contained therein:

NOW THEREFORE, the Tax Indemnification Agreement is amended by the addition of the following as a new Section 3(g):

(g) Without limiting any other rights of the Company contained herein, (i) the Company (or its designee) shall have the right to designate or appoint (or consent or agree to the designation or appointment of) any individual or entity, including making any such designation or appointment (or consenting or agreeing to any such designation or appointment) in the name and on behalf of the Indemnitee (and in the case of an Indemnitee who or which is the transferee of substantially all of the assets of a former partner of the Partnership, on behalf of such former partner), as the Tax Matters Partner (as defined in Section 6231(a)(7) of the Internal Revenue Code), with respect to any taxable year, of the Partnership or any Affiliate; (ii) the Company, any designee of the Company or any relevant Tax Matters Partner shall have the right (including, without limitation, in its own name or in the name of, and on behalf of, any Indemnitee (and in the case of an Indemnitee who or which is the transferee of substantially all of the assets of a former partner of the Partnership, on behalf of such former partner)) to extend or to consent to the extension of any statute of limitations applicable to the Partnership, any Affiliate, or, to the extent applicable to a Partnership Item, the Indemnitee; and (iii) the Company, any designee of the Company or any relevant Tax Matters Partner shall have the right to take any action, including in the name and on behalf of the Indemnitee (and in the case of an Indemnitee who or which is the transferee of substantially all of the assets of a former partner of the Partnership, on behalf of such former partner), with respect to any actual or proposed assessment, audit, examination or other similar proceeding or matter, including agreeing to any audit adjustment or other assessment. Each Indemnitee (i) hereby approves of such designation, appointment, agreement, consent, extension or action and consents and agrees to execute, certify, acknowledge, deliver, swear to, file and record at the appropriate public offices evidence of such approval, designation, appointment, agreement, consent, extension or action and to take any other action reasonably requested by the Company (or its designee) as may be deemed necessary or appropriate and (ii) to the extent permitted by law, agrees not to challenge or claim or cooperate or assist any other party in challenging or claiming, as a complaint or a defense, or

otherwise asserting in any respect whatsoever, that such approval, designation, appointment, agreement, consent, extension or action is not valid or enforceable."

Unless otherwise defined herein, all capitalized and undefined terms shall have the meanings assigned to them in the Tax Indemnification Agreement.

Except as amended hereby, all of the terms of the Tax Indemnification Agreement shall remain and continue in full force and effect and are hereby confirmed in all respects.

IN WITNESS WHEREOF, the Company has executed and delivered this Amendment Number 1 as of the date above written.

THE GOLDMAN SACHS GROUP, INC.

/s/ John A. Thain By:

Name: John A. Thain
Title: Director, President and
Co-Chief Operating Officer

THE GOLDMAN SACHS GROUP, INC. and SUBSIDIARIES

COMPUTATION OF PER SHARE EARNINGS

(in millions, except share and per share amounts)

	Three Months Ended August		Nine Months Ended August	
	2000	1999	2000	1999
Numerator for basic and diluted EPS — earnings available to common stockholders	<u>\$ 824</u>	<u>\$ 638</u>	\$ 2,466	<u>\$ 1,985</u>
Denominator for basic EPS — weighted average number of common shares (1)	481,252,647	474,694,245	483,403,066	474,698,130
Restricted stock units Stock options	17,101,019 10,540,979	4,987,721 4,210,711	14,775,882 10,002,524	4,508,530 3,939,451
Dilutive potential common shares	27,641,998	9,198,432	24,778,406	8,447,981
Denominator for diluted EPS — weighted average number of common shares and dilutive potential common shares	508,894,645	483,892,677	508,181,472	483,146,111
Basic EPS	\$ 1.71 1.62	\$ 1.34 1.32	\$ 5.10 4.85	\$ 4.18 4.11

⁽¹⁾ Includes common stock and nonvoting common stock as well as restricted stock units awarded to employees for which no future service is required as a condition to the delivery of the underlying shares of common stock.

THE GOLDMAN SACHS GROUP, INC. and SUBSIDIARIES COMPUTATION OF RATIOS OF EARNINGS TO FIXED CHARGES (\$ in millions)

	Three Months Ended August 2000	Nine Months Ended August 2000
Net earnings	\$ 824	\$ 2,466
Add:		
Provision for taxes	549	1,644
Portion of rents representative of an interest factor	22	55
Interest expense on all indebtedness	4,324	11,836
Earnings, as adjusted	\$5,719	\$16,001
Fixed charges:		
Portion of rents representative of an interest factor	\$ 22	\$ 55
Interest expense on all indebtedness	4,324	11,836
Fixed charges	\$4,346	<u>\$11,891</u>
Ratio of earnings to fixed charges	1.32x	1.35x

October 10, 2000

Securities and Exchange Commission 450 Fifth Street, N.W. Washington, DC 20549

Re: The Goldman Sachs Group, Inc.

Registration Statements on Form S-8

(No. 333-80839) (No. 333-42068)

Registration Statements on Form S-3

(No. 333-34042) (No. 333-90677) (No. 333-75213) (No. 333-36178)

Commissioners:

We are aware that our report dated October 9, 2000 on our review of the condensed consolidated statement of financial condition of The Goldman Sachs Group, Inc. and Subsidiaries (the "Company") as of August 25, 2000, the related condensed consolidated statements of earnings for the three and nine months ended August 25, 2000 and August 27, 1999, the condensed consolidated statement of changes in stockholders' equity and partners' capital for the nine months ended August 25, 2000, the condensed consolidated statements of cash flows for the nine months ended August 25, 2000 and August 27, 1999, and the condensed consolidated statements of comprehensive income for the three and nine months ended August 25, 2000 and August 27, 1999, included in the Company's quarterly report on Form 10-Q for the quarter ended August 25, 2000 is incorporated by reference in the registration statements referred to above. Pursuant to Rule 436(c) under the Securities Act of 1933, such report should not be considered a part of such registration statements, and is not a report within the meaning of Sections 7 and 11 of that Act.

Very truly yours,

/s/ PricewaterhouseCoopers LLP

PERIOD-TYPE	9-MOS	
FISCAL-YEAR-END	9-MOS	NOV-24-2000
PERIOD-START		NOV-29-1999 F2
PERIOD-SIARI PERIOD-END		AUG-25-2000
CASH		17,406 F3
RECEIVABLES		29,162
SECURITIES-RESALE		41,016
SECURITIES RESALE SECURITIES-BORROWED		87,631
INSTRUMENTS-OWNED		92,502
PP&E		1,822 F4
TOTAL-ASSETS		275,004
SHORT-TERM		37,917
PAYABLES		61,299
REPOS-SOLD		42,697
SECURITIES-LOANED		6,535
INSTRUMENTS-SOLD		74,078
LONG-TERM		28,528
PREFERRED-MANDATORY		0
PREFERRED		0
COMMON		5
OTHER-SE		12,688
TOTAL-LIABILITY-AND-EQUITY		275,004
TRADING-REVENUE		5,543 F5
INTEREST-DIVIDENDS		12,579
COMMISSIONS		1,711 F6
INVESTMENT-BANKING-REVENUES		4,131
FEE-REVENUE		1,047 F6
INTEREST-EXPENSE		11,836
COMPENSATION		6,901 F7
INCOME-PRETAX		4,110
INCOME-PRE-EXTRAORDINARY		4,110
EXTRAORDINARY		0
CHANGES		0
NET-INCOME		2,466
EPS-BASIC		5.10
EPS-DILUTED		4.85

- F1 The amounts disclosed in the financial data summary should be read in conjunction with the condensed consolidated financial statements and the notes thereto.
- F2 Represents the first Monday of the period. F3 Includes cash and cash equivalents and cash and securities segregated in compliance with U.S. federal and other regulations as disclosed on the
- condensed consolidated statement of financial condition.
 F4 Included in other assets on the condensed consolidated statement of financial condition.
- F5 Includes revenues from principal investments, which mainly represents revenues from the firm's merchant banking investments.
- F6 Included in revenues from asset management and securities services on the condensed consolidated statement of earnings.
- F7 Includes amortization of employee initial public offering awards.